Registration No. 333-198013

# **UNITED STATES** SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

# AMENDMENT No. 4 TO FORM F-1

REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933

## FORWARD PHARMA A/S

(Exact Name of Registrant as Specified in its Charter)

Denmark

(State or Other Jurisdiction of Incorporation or Organization) 2834

(Primary Standard Industrial Classification Code Number)

Not Applicable (I.R.S. Employer Identification No.)

Forward Pharma A/S Østergade 24A, 1 1100 Copenhagen K, Denmark +45 33 44 42 42

(Address, including zip code, and telephone number, including area code, of Registrant's principal executive offices)

CT Corporation System 1015 15<sup>th</sup> Street, NW **Suite 1000** Washington, DC 20005 Tel: (202) 572-3100

(Name, address, including zip code, and telephone number, including area code, of agent for service)

Copies of all correspondence to:

Kristopher D. Brown Wayne J. Rapozo Dechert LLP 1095 Avenue of the Americas New York, NY 10036 Tel: (212) 698-3500

David B. Allen David C. Lee K&L Gates LLP 1 Park Plaza Twelfth Floor Irvine, CA 92614 Tel: (949) 253-0900

Approximate date of commencement of proposed sale to the public: As soon as practicable after effectiveness of this registration statement.

If any of the securities being registered on this Form are to be offered on a delayed or continuous basis pursuant to Rule 415 under the Securities Act of 1933, check the following box. o

If this Form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. o

If this Form is a post-effective amendment filed pursuant to Rule 462(c) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. o

If this Form is a post-effective amendment filed pursuant to Rule 462(d) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. o

#### CALCULATION OF REGISTRATION FEE

	Proposed maximum	
	aggregate offering	Amount of
Title of each class of securities to be registered	price(1)(2)(3)	registration fee(4)
Ordinary shares, par value DKK 0.10 per share(5)	\$240,952,382	\$30,518.67(6)

- Includes ordinary shares represented by American Depositary Shares, or ADSs, that may be purchased by the underwriters to cover over-allotments, if any. Estimated solely for the purpose of calculating the registration fee pursuant to Rule 457(o) of the Securities Act of 1933, as amended (the "Securities Act"). Includes ordinary shares loaned by Nordic Biotech Opportunity Fund K/S to the Company (the "Borrowed Shares") as part of the initial deposit of ordinary shares into the

American Depositary Receipt Program and the issuance of ADSs immediately following the consummation of this offering and sales of the ADSs for purposes of facilitating the orderly closing of this offering. The Company will issue ordinary shares in an amount equal to the Borrowed Shares to the underwriters and the underwriters will deposit such shares into the American Depositary Receipt Program, following which the Borrowed Shares underlying the ADSs shall be returned to Nordic Biotech Opportunity Fund K/S.

- Calculated pursuant to Rule 457(o) of the Securities Act based on an estimate of the proposed maximum aggregate offering price.
  Each ADS represents one ordinary share. ADSs issuable upon deposit of the ordinary shares registered hereby are being registered pursuant to a separate registration statement on Form F-6.
  A filing fee of \$25,760 was previously paid in connection with the initial filing of the Form F-1 on August 11, 2014. The remaining filing fee amount of \$4,758.67 is being paid with the filing of this Amendment No. 4 to Form F-1. (4) (5)
- (6)

The Registrant hereby amends this Registration Statement on such date or dates as may be necessary to delay its effective date until the Registrant shall file a further amendment which specifically states that this Registration Statement shall thereafter become effective in accordance with Section 8(a) of the Securities Act or until the Registration Statement shall become effective on such date as the Securities and Exchange Commission, acting pursuant to said Section 8(a), may determine.

# EXPLANATORY NOTE

This Amendment No. 4 to Form F-1 is being filed solely for the purpose of filing certain exhibits to the Registration Statement on Form F-1. No change is
being made to the prospectus constituting Part I of the Registration Statement or Items 6, 7 or 9 of Part II of the Registration Statement. Accordingly, Part I and
Items 6, 7 and 9 of Part II have not been included herein.

#### PART II—Information not required in the prospectus

#### Item 8. Exhibits

- (a) The following documents are filed as part of this Registration Statement:
  - 1.1 Form of Underwriting Agreement.
  - 3.1 English translation of form of Articles of Association of Forward Pharma A/S.
  - 3.2\* English translation of Articles of Association of Forward Pharma A/S dated July 24, 2014.
  - 3.3\*\* English translation of Articles of Association of Forward Pharma A/S dated September 9, 2014.
  - 3.4\*\*\*English translation of Articles of Association of Forward Pharma A/S dated September 30, 2014.
  - 4.1\*\* Registration Rights Agreement, dated September 11, 2014, between Forward Pharma A/S and each of the investors listed on Schedule A thereto.
  - 4.2 Form of Deposit Agreement.
  - 4.3 Form of American Depositary Receipt (included in Exhibit 4.2).
  - 4.4\*\* New Shareholders' Agreement, dated September 8, 2014, between Nordic Biotech K/S, Nordic Biotech Opportunity Fund K/S, NB FP Investment K/S and NB FP Investment II K/S.
  - 4.5\* Convertible Loan Agreement dated May 30, 2014 between Forward Pharma A/S and NB FP Investment II K/S.
  - 4.6\* Convertible Loan Agreement dated August 6, 2014 between Forward Pharma A/S and BVF Forward Pharma L.P.
  - 4.7 Form of Stock Lending Agreement among Nordic Biotech Opportunity Fund K/S, Leerink Partners and Forward Pharma A/S.
  - 5.1 Opinion of Nielsen Nørager, counsel of Forward Pharma A/S, as to the validity of the ordinary shares.
  - 8.1 Opinion of Deloitte, as to certain tax matters.
  - 8.2 Opinion of Dechert LLP, counsel of Forward Pharma A/S, as to U.S. tax matters.
  - 10.1\* Patent Transfer Agreement dated May 4, 2010 between Forward Pharma A/S and Aditech Pharma AG.
  - 10.2\* Framework Agreement dated July 11, 2014, between Nordic Biotech K/S, Nordic Biotech Opportunity Fund K/S, BML Healthcare I, L.P., NB FP Investment K/S, and NB FP Investment II K/S.
  - 10.3\* Form of Director and Officer Indemnification Agreement.
  - 10.4\* Indemnification Agreement with Joel Sendek.
  - 21.1\* List of subsidiaries.
  - 23.1\*\*\*Consent of Ernst & Young Denmark P/S.
  - 23.2 Consent of Nielsen Nørager, counsel of Forward Pharma A/S (included in Exhibit 5.1).
  - 23.3 Consent of Deloitte (included in Exhibit 8.1).
  - 23.4 Consent of Dechert LLP, counsel of Forward Pharma A/S (included in Exhibit 8.2).

<sup>\*</sup> Filed as part of this registration statement on Form F-1 (Registration No. 333-198013) on August 11, 2014.

<sup>\*\*</sup> Filed as part of this registration statement on Form F-1 (Registration No. 333-198013) on September 12, 2014.

<sup>\*\*\*</sup> Filed as part of this registration statement on Form F-1 (Registration No. 333-198013) on October 2, 2014.

# **Signatures**

Pursuant to the requirements of the Securities Act of 1933, the registrant certifies that it has reasonable grounds to believe that it meets all of the requirements for filing on Form F-1 and has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in Copenhagen, Denmark on October 9, 2014.

# FORWARD PHARMA A/S

By: /s/ PEDER MØLLER ANDERSEN

Name: Peder Møller Andersen Title: Chief Executive Officer

Pursuant to the requirements of the Securities Act of 1933, as amended, this registration statement has been signed by the following persons on October 9, 2014, in the capacities indicated:

	<u>Name</u>	<u>Title</u>	
	/s/ PEDER MØLLER ANDERSEN	Chief Executive Officer	
	Peder Møller Andersen	- (principal executive officer)	
	/s/ JOEL SENDEK	Chief Financial Officer (principal financial officer and principal	
	Joel Sendek	- accounting officer)	
	/s/ FLORIAN SCHÖNHARTING		
	Florian Schönharting	Director (Chairman)	
	*		
	J. Kevin Buchi	Director	
	*		
	Torsten Goesch	Director	
	*		
	Jan G. J. van de Winkel	Director	
*By:	/s/ FLORIAN SCHÖNHARTING	_	
	Florian Schönharting Attorney-in-fact		
		II-2	

# Signature of Authorized U.S. Representative of Registrant

Pursuant to the requirements of the Securities Act of 1933, the undersigned, the duly authorized representative in the United States of Forward Pharma A/S has signed this registration statement on October 9, 2014.

FORWARD PHARMA USA, LLC

By: /s/ JOEL SENDEK

Name: Joel Sendek

Title: Chief Financial Officer

II-3

#### **Exhibit index**

The following documents are filed as part of this registration statement:

- 1.1 Form of Underwriting Agreement.
- 3.1 English translation of form of Articles of Association of Forward Pharma A/S.
- 3.2\* English translation of Articles of Association of Forward Pharma A/S dated July 24, 2014.
- 3.3\*\* English translation of Articles of Association of Forward Pharma A/S dated September 9, 2014.
- 3.4\*\*\*English translation of Articles of Association of Forward Pharma A/S dated September 30, 2014.
- 4.1\*\* Registration Rights Agreement, dated September 11, 2014, between Forward Pharma A/S and each of the investors listed on Schedule A thereto.
- 4.2 Form of Deposit Agreement.
- 4.3 Form of American Depositary Receipt (included in Exhibit 4.2).
- 4.4\*\* New Shareholders' Agreement, dated September 8, 2014, between Nordic Biotech K/S, Nordic Biotech Opportunity Fund K/S, NB FP Investment K/S and NB FP Investment II K/S.
- 4.5\* Convertible Loan Agreement dated May 30, 2014 between Forward Pharma A/S and NB FP Investment II K/S.
- 4.6\* Convertible Loan Agreement dated August 6, 2014 between Forward Pharma A/S and BVF Forward Pharma L.P.
- 4.7 Form of Stock Lending Agreement among Nordic Biotech Opportunity Fund K/S, Leerink Partners and Forward Pharma A/S.
- 5.1 Opinion of Nielsen Nørager, counsel of Forward Pharma A/S, as to the validity of the ordinary shares.
- 8.1 Opinion of Deloitte, as to certain tax matters.
- 8.2 Opinion of Dechert LLP, counsel of Forward Pharma A/S, as to U.S. tax matters.
- 10.1\* Patent Transfer Agreement dated May 4, 2010 between Forward Pharma A/S and Aditech Pharma AG.
- 10.2\* Framework Agreement dated July 11, 2014, between Nordic Biotech K/S, Nordic Biotech Opportunity Fund K/S, BML Healthcare I, L.P., NB FP Investment K/S, and NB FP Investment II K/S.
- 10.3\* Form of Director and Officer Indemnification Agreement.
- 10.4\* Indemnification Agreement with Joel Sendek.
- 21.1\* List of subsidiaries.
- 23.1\*\*\*Consent of Ernst & Young Denmark P/S.
- 23.2 Consent of Nielsen Nørager, counsel of Forward Pharma A/S (included in Exhibit 5.1).
- 23.3 Consent of Deloitte (included in Exhibit 8.1).
- 23.4 Consent of Dechert LLP, counsel of Forward Pharma A/S (included in Exhibit 8.2).

\* Filed as part of this registration statement on Form F-1 (Registration No. 333-198013) on August 11, 2014.

\*\* Filed as part of this registration statement on Form F-1 (Registration No. 333-198013) on September 12, 2014.

\*\*\* Filed as part of this registration statement on Form F-1 (Registration No. 333-198013) on October 2, 2014.

# QuickLinks

EXPLANATORY NOTE
PART II—Information not required in the prospectus

Item 8. Exhibits

<u>Signatures</u> <u>Signature of Authorized U.S. Representative of Registrant Exhibit index</u>

#### Forward Pharma A/S

(a Danish public limited liability company)

[·] American Depositary Shares

Representing an Aggregate of [·] Ordinary Shares

#### **UNDERWRITING AGREEMENT**

Dated: [·], 2014

 $[\cdot]$  American Depositary Shares

Representing an Aggregate of  $[\cdot]$  Ordinary Shares

#### UNDERWRITING AGREEMENT

 $[\cdot]$ , 2014

Leerink Partners LLC Jefferies LLC as Representatives of the several Underwriters

c/o Leerink Partners LLC One Federal Street, 37<sup>th</sup> Floor Boston, Massachusetts 02110

> Jefferies LLC 520 Madison Avenue, 12th Floor New York, New York 10022

## Ladies and Gentlemen:

Forward Pharma A/S, a Danish public limited liability company (the "Company"), confirms its agreement with the several Underwriters named in Schedule A hereto (the "Underwriters," which term shall also include any underwriter substituted as hereinafter provided in Section 10 hereof), for whom Leerink Partners LLC ("Leerink") and Jefferies LLC ("Jefferies") are acting as representatives (in such capacity, the "Representatives"), with respect to (i) the sale by the Company and the purchase by the Underwriters, acting severally and not jointly, of the respective numbers of the American Depositary Shares (the "ADSs"), each ADS representing one of the Company's ordinary shares, nominal value DKK 0.10 per share (the "Ordinary Shares"), set forth in Schedule A hereto and (ii) the grant by the Company to the Underwriters, acting severally and not jointly, of the option described in Section 2(b) hereof to purchase all or any part of [·] additional ADSs solely for the purpose of covering over-allotments. Each ADS will initially represent one (1) Ordinary Share deposited pursuant to the Deposit Agreement (as defined below), in each case in the manner contemplated in the Prospectus (as defined below). The aforesaid [·] ADSs (the "Initial ADSs") to be purchased by the Underwriters and all or any part of the [·] ADSs subject to the option described in Section 2(b) hereof (the "Option ADSs") are herein called, collectively, the "Offered ADSs". Each reference to the Initial ADSs, the Optional ADSs or the Offered ADSs herein, unless the context otherwise requires, also includes the Ordinary Shares underlying such ADSs. The Ordinary Shares represented by the Initial ADSs are hereinafter called the "Option Shares" and the Underwritten Shares are hereinafter collectively referred to as the "Shares".

The Offered ADSs are to be issued pursuant to a deposit agreement (the "<u>Deposit Agreement</u>"), dated as of [·], 2014, among the Company, The Bank of New York Mellon, a New York banking corporation, as depositary (the "<u>Depositary</u>"), and owners and holders from time to time of the Offered ADSs.

Prior to the delivery of the Offered ADSs under this Agreement, the following actions, among others, will be effected by the holding of an extraordinary general meeting of the Company: the issuance of class A bonus shares to the class B shareholders of the Company, the conversion of all shares of the Company to one share class, the conversion of the entire principal amount of certain bridge financings

into shares of the Company, the issuance of bonus shares pro rata to all shareholders of the Company, and the adoption of a share split; all to the effect that following such extraordinary general meeting, the outstanding share capital of the Company will consist of [·] Ordinary Shares.

The Company is advised by you that the Underwriters propose to (i) make a public offering of the Offered ADSs as soon as the Representatives deem advisable after this Agreement has been executed and delivered and the Registration Statement has been declared effective and (ii) initially offer the Initial ADSs upon the terms set forth in the Prospectus (as defined below). The Company is further advised that the Representatives may from time to time increase or decrease the public offering price after the initial public offering to such extent as you may determine.

The Company has prepared and filed with the Securities and Exchange Commission (the "Commission") a registration statement on Form F-1 (No. 333-198013), including the related preliminary prospectus or prospectuses, covering the registration of the Offered ADSs under the Securities Act of 1933, as amended, and the rules and regulations thereunder (collectively the "1933 Act") and including a prospectus relating to the Ordinary Shares and a registration statement on Form F-6 relating to the Offered ADSs. Promptly after execution and delivery of this Agreement, the Company will prepare and file a prospectus in accordance with the provisions of Rule 430A ("Rule 430A") of the rules and regulations of the Commission under the 1933 Act (the "1933") Act Regulations") and Rule 424(b) ("Rule 424(b)") of the 1933 Act Regulations. The information included in such prospectus that was omitted from such registration statement at the time it became effective but that is deemed to be part of such registration statement at the time it became effective pursuant to Rule 430A(b) is herein called the "Rule 430A Information." Such registration statement, including the amendments thereto, the exhibits thereto and any schedules thereto at the time it became effective, and including the Rule 430A Information, is herein called the "Registration Statement." Any registration statement filed pursuant to Rule 462(b) of the 1933 Act Regulations is herein called the "Rule 462(b) Registration Statement" and, after such filing, the term "Registration Statement" shall include the Rule 462(b) Registration Statement. Each prospectus used prior to the effectiveness of the Registration Statement, and each prospectus that omitted the Rule 430A Information that was used after such effectiveness and prior to the execution and delivery of this Agreement is herein called a "preliminary prospectus." The final prospectus, in the form first furnished to the Underwriters for use in connection with the offering of the Offered ADSs, is herein called the "Prospectus." For purposes of this Agreement, all references to the Registration Statement, any preliminary prospectus, the Prospectus, the ADS Registration Statement (as defined below), the 1934 Act Registration Statement (as defined below) or any amendment or supplement to any of the foregoing shall be deemed to include the copy filed with the Commission pursuant to its Electronic Data Gathering, Analysis and Retrieval system or any successor system ("EDGAR").

The parties hereto agree that to facilitate the transactions contemplated by this Agreement, one or more of the Representatives shall execute and deliver to the Company one or more subscription lists on behalf of the Underwriters and, upon the several Underwriters becoming the owners of their respective Underwritten Shares or Option Shares, as applicable, the Representative(s) shall deposit such Shares with the Depositary against issuance of ADSs and/or ADRs (as defined below) evidencing ADSs, in each case in accordance with the terms of the Deposit Agreement. To facilitate the orderly closing of the offering of the Offered ADSs, the shares underlying the Offered ADSs immediately prior to and concurrent with the consummation of the offering of the Offered ADSs and the time of delivery of the ADSs may be shares loaned by a certain stockholder of the Company to the Company pursuant to the terms of a stock lending agreement (the "Stock Lending Agreement").

As used in this Agreement:

2

"Applicable Time" means [ : 00 P./A.M.], New York City time, on [:], 2014 or such other time as agreed by the Company and the Representatives.

"General Disclosure Package" means any Issuer General Use Free Writing Prospectuses issued at or prior to the Applicable Time, the most recent preliminary prospectus relating to the Shares and the Offered ADSs that was included in the Registration Statement and distributed to investors immediately prior to the Applicable Time, and the information included on <u>Schedule B-1</u> hereto, all considered together.

"Issuer Free Writing Prospectus" means any "issuer free writing prospectus," as defined in Rule 433 of the 1933 Act Regulations ("Rule 433"), including without limitation any "free writing prospectus" (as defined in Rule 405 of the 1933 Act Regulations ("Rule 405")) relating to the Offered ADSs that is (i) required to be filed with the Commission by the Company, (ii) a "road show that is a written communication" within the meaning of Rule 433(d)(8)(i), whether or not required to be filed with the Commission, or (iii) exempt from filing with the Commission pursuant to Rule 433(d)(5)(i) because it contains a description of the Offered ADSs or of the offering that does not reflect the final terms, in each case in the form filed or required to be filed with the Commission or, if not required to be filed, in the form retained in the Company's records pursuant to Rule 433(g).

"Issuer General Use Free Writing Prospectus" means any Issuer Free Writing Prospectus that is intended for general distribution to prospective investors (other than a "bona fide electronic road show," as defined in Rule 433 (the "Bona Fide Electronic Road Show")), as evidenced by its being specified in Schedule B-2 hereto.

"Issuer Limited Use Free Writing Prospectus" means any Issuer Free Writing Prospectus that is not an Issuer General Use Free Writing Prospectus.

"Testing-the-Waters Communication" means any oral or written communication with potential investors undertaken in reliance on Section 5(d) of the 1933 Act.

"Written Testing-the-Waters Communication" means any Testing-the-Waters Communication that is a written communication within the meaning of Rule 405 under the 1933 Act.

# SECTION 1. Representations and Warranties.

- (a) *Representations and Warranties by the Company.* The Company represents and warrants to each Underwriter as of the date hereof, the Applicable Time, the Closing Time (as defined below) and any Date of Delivery (as defined below), and agrees with each Underwriter, as follows:
  - (i) Registration Statement and Prospectuses. The Registration Statement has become effective under the 1933 Act. No stop order suspending the effectiveness of the Registration Statement or any post-effective amendment thereto has been issued under the 1933 Act, no order preventing or suspending the use of any preliminary prospectus, Issuer Free Writing Prospectus or the Prospectus has been issued and no proceedings for any of those purposes have been instituted or are pending or, to the Company's knowledge, contemplated. The Company has complied with each request (if any) from the Commission or its staff for additional information. The 1934 Act Registration Statement has become effective as provided in Section 12 of the 1934 Act (as defined below).

Each of the Registration Statement and any post-effective amendment thereto, at the time it became effective, complied in all material respects with the requirements of the 1933 Act and the 1933 Act Regulations. Each preliminary prospectus, the Prospectus and any amendment or supplement thereto, at the time each was filed with the Commission, complied in all material respects with the requirements of the 1933 Act and the 1933 Act Regulations. Each preliminary prospectus delivered to the Underwriters for use in connection with this offering and the Prospectus was or will be identical to the electronically transmitted copies thereof filed with the Commission pursuant to EDGAR, except to the extent permitted by Regulation S-T.

(ii) Accurate Disclosure. Neither the Registration Statement nor any amendment thereto, at its effective time, at the Closing Time or at any Date of Delivery, contained, contains or will contain an untrue statement of a material fact or omitted, omits or will omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading. As of the Applicable Time, none of (A) the General Disclosure Package, (B) any individual Issuer Limited Use Free Writing Prospectus, when considered together with the General Disclosure Package, nor (C) any individual Written Testing-the-Waters Communication, when considered together with the General Disclosure Package, included, includes or will include an untrue statement of a material fact or omitted, omits or will omit to state a material fact necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading. Neither the Prospectus nor any amendment or supplement thereto (including any prospectus wrapper), as of its issue date, at the time of any filing with the Commission pursuant to Rule 424(b), at the Closing Time or at any Date of Delivery, included, includes or will include an untrue statement of a material fact or omitted, omits or will omit to state a material fact necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading.

The representations and warranties in this subsection shall not apply to statements in or omissions from the Registration Statement (or any amendment thereto), the General Disclosure Package or the Prospectus (or any amendment or supplement thereto, including any prospectus wrapper) made in reliance upon and in conformity with written information furnished to the Company by any Underwriter through the Representatives expressly for use therein. For purposes of this Agreement, the only information so furnished shall be the information in the first paragraph under the heading "Underwriting—Discounts and Commissions," and the information in the first and third paragraphs under the heading "Underwriting—Price Stabilization, Short Positions and Penalty Bids" in each case contained in the Prospectus (collectively, the "<u>Underwriter Information</u>").

- (iii) <u>Issuer Free Writing Prospectuses</u>. No Issuer Free Writing Prospectus conflicts or will conflict with the information contained in the Registration Statement or the Prospectus, and any preliminary or other prospectus deemed to be a part thereof that has not been superseded or modified. The Company has made available a Bona Fide Electronic Road Show in compliance with Rule 433(d)(8)(ii) such that no filing of any "road show" (as defined in Rule 433(h)) is required in connection with the offering of the ADSs.
- (iv) <u>Testing-the-Waters Materials</u>. The Company (A) has not engaged in any Testing-the-Waters Communication other than Testing-the-Waters Communications with the consent of the Representatives with entities that are qualified institutional buyers within the meaning of Rule 144A under the 1933 Act or institutions that are accredited investors within the meaning of Rule 501 under the 1933 Act and (B) has not authorized anyone other than the Representatives to engage in Testing-the-Waters Communications. The Company confirms that the Representatives have been authorized to act on its behalf in undertaking Testing-the-Waters

4

Communications. The Company has not distributed any Written Testing-the-Waters Communications other than those listed on Schedule B-3 hereto.

- (v) Form 8-A. A registration statement on Form 8-A (No. 1-[·]) relating to the registration of the Ordinary Shares and the Offered ADSs has been filed with the Commission, has been declared effective under the Securities Exchange Act of 1934, as amended, and the rules and regulations thereunder (collectively, the "1934 Act") and the Ordinary Shares and the Offered ADSs have been duly registered under the 1934 Act pursuant to such registration statement. The various parts of such registration statement on Form 8-A for the registration of the Ordinary Shares and the Offered ADSs, including all exhibits thereto, each as amended at the time such part of the registration statement became effective, are hereinafter called the "1934 Act Registration Statement"; no stop order of the Commission preventing or suspending the effectiveness of the 1934 Act Registration Statement has been issued and no proceedings for such purpose have been instituted or, to the Company's knowledge, are contemplated by the Commission; the 1934 Act Registration Statement, when it became effective and on the date of this Agreement, complied and complies, in all material respects, with the applicable requirements of the 1934 Act, and did not, when it became effective, does not and, as of the time of purchase and any additional time of purchase, will not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading.
- (vi) Form F-6. A registration statement on Form F-6 (No. 333-[·]), and any amendments thereto, in respect of the Offered ADSs has been filed with the Commission; such registration statement in the form heretofore delivered to the Representatives and, excluding exhibits, to the Representatives for each of the other Underwriters, has been declared effective by the Commission; no other document with respect to such registration statement has heretofore been filed with the Commission; no stop order suspending the effectiveness of such registration statement has been issued and, to the Company's knowledge, no proceeding for that purpose has been initiated or threatened by the Commission (the various parts of such registration statement, including all exhibits thereto, each as amended at the time such part of the registration statement became effective, being hereinafter called the "ADS Registration Statement"); and the ADS Registration Statement when it became effective conformed, and any further amendments thereto will conform, in all material respects to the requirements of the 1933 Act, and did not, as of the applicable effective date, contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading.
- (vii) <u>Company Not Ineligible Issuer</u>. At the time of filing the Registration Statement and any post-effective amendment thereto and at the date hereof, the Company was not and is not an "ineligible issuer," as defined in Rule 405, without taking account of any determination by the Commission pursuant to Rule 405 that it is not necessary that the Company be considered an ineligible issuer.
- (viii) <u>Emerging Growth Company Status.</u> From the time of the initial confidential submission of the Registration Statement to the Commission (or, if earlier, the first date on which the Company engaged directly or through any Person authorized to act on its behalf in any Testing-the-Waters Communication) through the date hereof, the Company has been and is an "emerging growth company," as defined in Section 2(a) of the 1933 Act (an "<u>Emerging Growth Company</u>").
- (ix) <u>Independent Accountants</u>. Ernst & Young Denmark P/S, the independent registered public accounting firm that has certified the financial statements and supporting

schedules included in the Registration Statement, the General Disclosure Package and the Prospectus (including the related notes thereto), was, during the period of its audits, and is independent with respect to the Company as required by the 1933 Act, the 1933 Act Regulations, the Public Company Accounting Oversight Board and applicable Danish law.

- Financial Statements; Non-GAAP Financial Measures. The financial statements included in the Registration Statement, the General Disclosure Package and the Prospectus, together with the related schedules and notes, comply in all material respects with the applicable requirements of the 1933 Act and present fairly the consolidated financial position of the Company and its consolidated subsidiaries at the dates indicated and the results of operations, stockholders' equity and cash flows of the Company and its consolidated subsidiaries for the periods specified; said financial statements have been prepared in conformity with International Financial Reporting Standards ("IFRS") as issued by the International Accounting Standards Board and in compliance with the financial reporting requirements of Danish law, in each case applied on a consistent basis throughout the periods involved except unaudited financial statements, which are subject to normal year-end adjustment and do not contain certain footnotes as permitted by IAS 34 Interim Financial Reporting and the applicable rules of the Commission. The supporting schedules, if any, present fairly, in all material respects, in accordance with IFRS the information required to be stated therein. The selected financial data and the summary financial information included in the Registration Statement, the General Disclosure Package and the Prospectus have been derived from the accounting records of the Company and its consolidated subsidiaries, present fairly the information shown therein and have been compiled on a basis consistent with that of the audited financial statements included therein. No pro forma financial statements or supporting schedules are required to be included or incorporated by reference in the Registration Statement, the General Disclosure Package or the Prospectus under the 1933 Act or the 1933 Act Regulations. All disclosures contained in the Registration Statement, the General Disclosure Package or the Prospectus regarding "non-GAAP financial measures" (as such term is defined by the rules and regulations of the Commission) comply with Regulation G of the 1934 Act, and Item 10 of Regulation S-K, to the extent applicable.
- (xi) <u>Off-Balance Sheet Financing</u>. Except as described in the Registration Statement, the General Disclosure Package and the Prospectus, neither the Company nor its Subsidiaries have any off-balance sheet financing as defined under IFRS.
- (xii) No Material Adverse Change. Except as otherwise stated therein, since the respective dates as of which information is given in the Registration Statement, the General Disclosure Package or the Prospectus, (A) there has been no material adverse change, or any development that could reasonably be expected to result in a material adverse change, in the condition, financial or otherwise, or in the earnings, business, prospects, operations, assets, liabilities or prospects of the Company and its Subsidiaries considered as one enterprise, whether or not arising in the ordinary course of business (any such change referred to herein as a "Material Adverse Effect"), (B) there have been no transactions entered into by the Company or its Subsidiaries, other than those in the ordinary course of business, which are material with respect to the Company and its Subsidiaries considered as one enterprise, and (C) there has been no dividend or distribution of any kind declared, paid or made by the Company on any class of its capital stock.
- (xiii) <u>Good Standing of the Company</u>. The Company has been duly organized and is validly existing and in good standing (to the extent such concept exists) under the laws of Denmark, and has corporate power or similar power and authority to own, lease and operate its properties and to conduct its business as described in the Registration Statement, the General

6

Disclosure Package and the Prospectus and to enter into and perform its obligations under this Agreement; and the Company is duly qualified to transact business and is in good standing (to the extent such concept exists) in each other jurisdiction in which such qualification is required, whether by reason of the ownership or leasing of property or the conduct of business, except where the failure so to qualify or to be in good standing would not result in a Material Adverse Effect.

- (xiv) Good Standing of Subsidiaries. The only subsidiaries of the Company are the subsidiaries listed on Exhibit 21.1 to the Registration Statement (the "Subsidiaries"). Each Subsidiary has been duly organized and is validly existing in good standing (to the extent such concept exists) under the laws of the jurisdiction of its incorporation or organization, has corporate or similar power and authority to own, lease and operate its properties and to conduct its business as described in the Registration Statement, the General Disclosure Package and the Prospectus and is duly qualified to transact business and is in good standing (to the extent such concept exists) in each jurisdiction in which such qualification is required, whether by reason of the ownership or leasing of property or the conduct of business, except where the failure to so qualify or to be in good standing would not result in a Material Adverse Effect. Except as otherwise disclosed in the Registration Statement, the General Disclosure Package and the Prospectus, all of the issued and outstanding capital stock of each Subsidiary has been duly authorized and validly issued, is fully paid and non-assessable and is owned by the Company free and clear of any security interest, mortgage, pledge, lien, encumbrance, claim or equity. None of the outstanding shares of capital stock of the Subsidiaries were issued in violation of the preemptive or similar rights of any securityholder of such Subsidiary.
- (xv) <u>Capitalization</u>. The issued and outstanding shares of capital stock of the Company are as set forth in the Registration Statement, the General Disclosure Package and the Prospectus in the column entitled "Actual" under the caption "Capitalization" (except for subsequent issuances, if any, pursuant to this Agreement, pursuant to reservations, agreements or employee benefit plans referred to in the Registration Statement, the General Disclosure Package and the Prospectus or pursuant to the exercise of convertible securities or options referred to in the Registration Statement, the General Disclosure Package and the Prospectus). The outstanding shares of capital stock of the Company have been validly issued, are fully paid and non-assessable and registered with the Danish Business Authority. None of the outstanding shares of capital stock of the Company were issued in violation of the preemptive or other similar rights of any securityholder of the Company.
- (xvi) <u>Authorization of Underwriting Agreement</u>. This Agreement has been duly authorized, executed and delivered by the Company and, assuming the due authorization, execution and delivery by the other parties party hereto, constitutes a legal, valid and binding agreement of the Company enforceable in accordance with its terms, and the Company has taken or will take all corporate actions required by its articles of association, or other constitutional documents, and Danish law in connection with the performance of its obligations under this Agreement.

7

Upon (i) issuance by the Depositary of the Offered ADSs against the deposit of Shares in respect thereof, (ii) payment by the Underwriters for the Offered ADSs and for the Shares evidenced thereby in accordance with this Agreement, and (iii) due execution and delivery by the Depositary of the ADRs (as defined below) evidencing the Offered ADSs against the deposit of Shares in respect thereof, in accordance with the provisions of the Deposit Agreement and the Stock Lending Agreement, such Offered ADSs and/or ADRs will be duly and validly issued and the persons in whose names the Offered ADSs and/or the ADRs are registered will be entitled to the rights specified therein and in the Deposit Agreement; and the ADRs evidencing the Offered ADSs are in due and proper form. The Deposit Agreement conforms in all material respects to all statements relating thereto contained in the Registration Statement, the General Disclosure Package and the Prospectus.

- (xviii) Description of Shares/ADSs. The Shares to be subscribed for by the Underwriters when issued, registered with the Danish Business Authority and delivered by the Company pursuant to this Agreement against payment of the consideration set forth herein, will be validly issued, fully paid and non-assessable and free and clear of any security interest, mortgage, pledge, lien encumbrances or claim; and the issuance of the Shares is not subject to the preemptive or other similar rights of any security holder of the Company that have not been irrevocably waived. The Shares and the ADSs conform in all material respects to all statements relating thereto contained in the Registration Statement, the General Disclosure Package and the Prospectus and such description conforms to the rights set forth in the instruments defining the same. The certificates for the Offered ADSs are in due and proper form. No holder of Shares or the ADSs will be subject to personal liability by reason of being such a holder.
- Absence of Registration and Other Rights. Except as described in the Registration Statement, the General Disclosure Package and the Prospectus, (i) neither the Company nor any of its Subsidiaries has outstanding, and at the Closing Time will have outstanding, any options to purchase, or any warrants to subscribe for, or any securities or obligations convertible into, or any contracts or commitments to issue or sell, any Ordinary Shares, ADSs or any warrants or other convertible securities or obligations, (ii) no person has the right, contractual or otherwise, to cause the Company to issue or sell to it any Ordinary Shares, ADSs or any other equity interests of the Company, (iii) no person has any preemptive rights, resale rights, rights of first refusal or other rights to purchase any Ordinary Shares, ADSs or any other equity interests in the Company, (iv) no person has the right to act as an underwriter or as a financial advisor to the Company in connection with the offer and sale of the Offered ADSs and (v) no person has the right, contractual or otherwise, to cause the Company to register under the 1933 Act any Ordinary Shares, ADSs or any other equity interests in the Registration Statement or the offering contemplated hereby.
- (xx) Absence of Violations, Defaults and Conflicts. Neither the Company nor any of its Subsidiaries is (A) in violation of its respective charter, by-laws or similar organizational document, (B) in default in the performance or observance of any obligation, agreement, covenant or condition contained in any contract, indenture, mortgage, deed of trust, loan or credit agreement, note, lease or other agreement or instrument to which the Company or its Subsidiaries is a party or by which any of them may be bound or to which any of the properties or assets of the Company or its Subsidiaries is subject (collectively, "Agreements and Instruments") or (C) in violation of any law, statute, rule, regulation, judgment, order, writ or decree of any arbitrator, court, governmental body, regulatory body, administrative agency or other authority, body or agency having jurisdiction over the Company or its Subsidiaries or any of their respective properties, assets or operations (each, a "Governmental Entity"), except, in the case of (B) and

8

(C) above, for any such default that would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect. The execution, delivery and performance of this Agreement and the Deposit Agreement and the consummation of the transactions contemplated herein and therein and in the Registration Statement, the General Disclosure Package and the Prospectus (including the deposit of the Shares being deposited with the Depositary against the issuance of Offered ADSs to be delivered and/or the ADRs evidencing Offered ADSs to be delivered, the issuance and sale of the Offered ADSs and the use of the proceeds from the sale of the Offered ADSs as described therein under the caption "Use of Proceeds") and compliance by the Company with its obligations hereunder and under the Deposit Agreement have been duly authorized by all necessary corporate action and do not and will not (A) whether with or without the giving of notice or passage of time or both, conflict with or constitute a breach of, or default or Repayment Event (as defined below) under, or result in the creation or imposition of any lien, charge or encumbrance upon any properties or assets of the Company or its Subsidiaries pursuant to, the Agreements and Instruments, (B) result in any violation of the provisions of the charter, by-laws or similar organizational document of the Company or its Subsidiaries or (C) result in a violation of any law, statute, rule, regulation, judgment, order, writ or decree of any Governmental Entity, except, in the case of (A) and (C) above, for any such default that would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect. As used herein, a "Repayment Event" means any event or condition which gives the holder of any note, debenture or other evidence of indebtedness (or any person acting on such holder's behalf) the right to require the repurchase, redemption or repayment of all or a portion of such indebtedness by the Company or its Subsidiaries.

- (xxi) <u>Absence of Labor Dispute</u>. No labor dispute with the employees of the Company or its Subsidiaries exists or, to the knowledge of the Company, is imminent, and the Company is not aware of any existing or imminent labor disturbance by the employees of any of its or its Subsidiaries' principal suppliers, manufacturers, customers or contractors, which, in either case, would result in a Material Adverse Effect.
- (xxii) <u>Absence of Proceedings</u>. Except as disclosed in the Registration Statement, the General Disclosure Package and the Prospectus, there is no action, suit, proceeding, inquiry or investigation before or brought by any Governmental Entity now pending or, to the knowledge of the Company, threatened, against or affecting the Company or its Subsidiaries, which would reasonably be expected to result in a Material Adverse Effect, or which would reasonably be expected to materially and adversely affect their respective properties or assets or the consummation of the transactions contemplated in this Agreement or the performance by the Company of its obligations hereunder; and the aggregate of all pending legal

or governmental proceedings to which the Company or its Subsidiaries is a party or of which any of their respective properties or assets is the subject which are not described in the Registration Statement, the General Disclosure Package and the Prospectus, including ordinary routine litigation incidental to the business, would not reasonably be expected to result in a Material Adverse Effect.

(xxiii) <u>Accuracy of Exhibits</u>. There are no contracts or documents which are required to be described in the Registration Statement, the General Disclosure Package or the Prospectus or to be filed as exhibits to the Registration Statement which have not been so described and filed as required.

(xxiv) <u>Absence of Further Requirements</u>. No filing with, or authorization, approval, consent, license, order, registration, qualification or decree of, any Governmental Entity, or of or with any self-regulatory organization or other non-governmental regulatory authority (including,

9

without limitation, the NASDAQ Global Market (the "NASDAQ")) is necessary or required for the performance by the Company of its obligations hereunder, in connection with the offering, issuance or sale of the Shares and Offered ADSs hereunder, the deposit of the Shares being deposited with the Depositary against the issuance of Offered ADSs to be delivered and/or the ADRs evidencing Offered ADSs to be delivered or the consummation of the transactions contemplated by this Agreement or the Deposit Agreement, except such as have been already obtained or as may be required under the 1933 Act, the 1933 Act Regulations, the rules of the NASDAQ, state securities laws or the rules of the Financial Industry Regulatory Authority ("FINRA").

Possession of Licenses and Permits. Except as disclosed in the Registration Statement, the General Disclosure Package and the Prospectus, and except as would not reasonably be expected, individually or in the aggregate, to result in a Material Adverse Effect: (i) the Company has not received any FDA Form 483 notice of adverse finding, warning letter or other written correspondence or notice from the U.S. Food and Drug Administration ("FDA") or any other Governmental Entity alleging or asserting noncompliance with any Applicable Laws (as defined in clause (ii) below) or Authorizations (as defined in clause (iii) below); (ii) the Company is and has been in compliance with statutes, laws, ordinances, rules and regulations applicable to the Company for the ownership, testing, development, manufacture, packaging, processing, use, labeling, promotion, storage, import, export or disposal of any product manufactured or distributed by the Company, including without limitation, the Federal Food, Drug, and Cosmetic Act, 21 U.S.C. § 301, et seq., similar laws of other Governmental Entities and the regulations promulgated pursuant to such laws (collectively, "Applicable Laws"); (iii) the Company possesses all licenses, certificates, approvals, authorizations, permits and supplements or amendments thereto required by any such Applicable Laws and/or for the ownership of its properties or the conduct of its business as described in the Registration Statement, the General Disclosure Package and the Prospectus (collectively, "Authorizations") and such Authorizations are valid and in full force and effect and the Company is not in violation of any term of any such Authorizations; (iv) the Company has not received written notice of any ongoing claim, action, suit, proceeding, hearing, enforcement, investigation, arbitration or other action from any Governmental Entity or third party alleging that any product, operation or activity is in violation of any Applicable Laws or Authorizations or has any knowledge that any such Governmental Entity or third party is considering any such claim, litigation, arbitration, action, suit, investigation or proceeding, nor, to the best of the Company's knowledge, has there been any noncompliance with or violation of any Applicable Laws by the Company that could reasonably be expected to require the issuance of any such written notice or result in an investigation, corrective action, or enforcement action by FDA or similar Governmental Entity; (v) the Company has not received written notice that any Governmental Entity has taken, is taking or intends to take action to limit, suspend, modify or revoke any Authorizations or has any knowledge that any such Governmental Entity has threatened or is considering such action; and (vi) the Company has filed, obtained, maintained or submitted all reports, documents, forms, notices, applications, records, claims, submissions and supplements or amendments as required by any Applicable Laws or Authorizations and that all such reports, documents, forms, notices, applications, records, claims, submissions and supplements or amendments were complete, correct and not misleading on the date filed (or were corrected or supplemented by a subsequent submission). Neither the Company nor, to the Company's knowledge, any of its directors, officers, employees or agents has been convicted of any crime under the federal Food, Drug and Cosmetic Act or comparable laws of other Governmental Entities or has been the subject of an FDA debarment proceeding. The Company has not been nor is it now subject to FDA's Application Integrity Policy. To the Company's knowledge, neither the Company, nor any of its directors, officers, employees or agents, has made, or caused the making of, any false statements

10

on, or material omissions from, any other records or documentation prepared or maintained to comply with the requirements of the FDA or any other Governmental Entity.

(xxvi) <u>Title to Property</u>. The Company and its Subsidiaries have good and marketable title to all real property owned by them and good title to all other properties owned by them, in each case, free and clear of all mortgages, pledges, liens, security interests, claims, restrictions or encumbrances of any kind except such as (A) are described in the Registration Statement, the General Disclosure Package and the Prospectus or (B) do not, individually or in the aggregate, materially affect the value of such property and do not interfere with the use made and proposed to be made of such property by the Company or any of its Subsidiaries; and all of the leases and subleases material to the business of the Company and its Subsidiaries, considered as one enterprise, and under which the Company or any of its Subsidiaries holds properties described in the Registration Statement, the General Disclosure Package or the Prospectus, are in full force and effect, and neither the Company nor any such Subsidiary has any notice of any material claim of any sort that has been asserted by anyone adverse to the rights of the Company or its Subsidiaries under any of the leases or subleases mentioned above, or affecting or questioning the rights of the Company or such Subsidiary to the continued possession of the leased or subleased premises under any such lease or sublease.

(xxvii) Intellectual Property. Except as disclosed in the Registration Statement, the General Disclosure Package and the Prospectus, the Company owns or has valid, binding and enforceable licenses or other rights under the patents, patent applications, licenses, inventions, copyrights, know how (including trade secrets and other unpatented and/or unpatentable proprietary or confidential information, systems or procedures), trademarks, service marks, trade names or other intellectual property necessary for, or used in the conduct, or the proposed conduct, of the business of the Company in the manner described in the Registration Statement, the General Disclosure Package and the Prospectus (collectively, the "Intellectual Property"); the patents, trademarks, and copyrights, if any, included within the Intellectual Property are valid, enforceable, and subsisting; other than as disclosed in the Registration Statement, the General Disclosure Package and the Prospectus, (A) the Company is not obligated to pay a material royalty, grant a license to, or provide other material consideration to any third party in connection with the Intellectual Property, (B) the Company has not received any notice of any claim of infringement, misappropriation or conflict with any asserted rights of others

with respect to any of the Company's drug candidates, processes or Intellectual Property, (C) to the knowledge of the Company, neither the sale nor use of any of the discoveries, inventions, drug candidates or processes of the Company referred to in the Registration Statement, the General Disclosure Package or the Prospectus do or will, to the knowledge of the Company, infringe, misappropriate or violate any right or valid patent claim of any third party, and (D) to the knowledge of the Company, no third party has any ownership right in or to any Intellectual Property that is owned by the Company, other than any co-owner of any patent constituting Intellectual Property who is listed on the records of the U.S. Patent and Trademark Office (the "USPTO") and/or the European Patent Office (the "EPO") and any co-owner of any patent application constituting Intellectual Property who is named in such patent application, and, to the knowledge of the Company, no third party has any ownership right in or to any Intellectual Property in any field of use that is exclusively licensed to the Company, other than any licensor to the Company of such Intellectual Property.

(xxviii) <u>Patents and Patent Applications</u>. Except as disclosed in the Registration Statement, the General Disclosure Package and the Prospectus, all patents and patent applications owned by or licensed to the Company or under which the Company has rights have, to the knowledge of the Company, been duly and properly filed and maintained; to the knowledge of the

11

Company, the parties prosecuting such applications have complied with their duty of candor and disclosure to the USPTO and/or the EPO (as applicable) in connection with such applications; and the Company is not aware of any facts required to be disclosed to the USPTO and/or the EPO (as applicable) that were not disclosed to the USPTO and which would preclude the grant of a patent in connection with any such application or would reasonably be expected to form the basis of a finding of invalidity with respect to any patents that have issued with respect to such applications.

(xxix) Environmental Laws. Except as described in the Registration Statement, the General Disclosure Package and the Prospectus or would not, individually or in the aggregate, result in a Material Adverse Effect, (A) neither the Company nor any of its Subsidiaries is in violation of any federal, state, local or foreign statute, law, rule, regulation, ordinance, code, policy or rule of common law or any judicial or administrative interpretation thereof, including any judicial or administrative order, consent, decree or judgment, relating to pollution or protection of human health, the environment (including, without limitation, ambient air, surface water, groundwater, land surface or subsurface strata) or wildlife, including, without limitation, laws and regulations relating to the release or threatened release of chemicals, pollutants, contaminants, wastes, toxic substances, hazardous substances, petroleum or petroleum products, asbestos-containing materials or mold (collectively, "Hazardous Materials") or to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of Hazardous Materials (collectively, "Environmental Laws"), (B) the Company and its Subsidiaries have all permits, authorizations and approvals required under any applicable Environmental Laws and are each in compliance with their requirements, (C) there are no pending or, to the knowledge of the Company threatened, administrative, regulatory or judicial actions, suits, demands, demand letters, claims, liens, notices of noncompliance or violation, investigation or proceedings relating to any Environmental Law against the Company or its Subsidiaries and (D) there are no events or circumstances that would reasonably be expected to form the basis of an order for clean-up or remediation, or an action, suit or proceeding by any private party or Governmental Entity, against or affecting the Company or its Subsidiaries relating to Hazardous Materials or any Environmental Laws.

(xxx) Accounting Controls and Disclosure Controls. The Company and its Subsidiaries maintain effective internal control over financial reporting (as defined under Rule 13a-15 and 15d-15 under the rules and regulations of the Commission under the 1934 Act (the "1934 Act Regulations")) and a system of internal accounting controls sufficient to provide reasonable assurances that (A) transactions are executed in accordance with management's general or specific authorization; (B) transactions are recorded as necessary to permit preparation of financial statements in conformity with IFRS and to maintain accountability for assets; (C) access to assets is permitted only in accordance with management's general or specific authorization; and (D) the recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any differences. Except as described in the Registration Statement, the General Disclosure Package and the Prospectus, since the end of the Company's most recent audited fiscal year, there has been (1) no significant deficiencies or material weaknesses in the Company's internal control over financial reporting (whether or not remediated) and (2) no change in the Company's internal control over financial reporting that has materially adversely affected, or is reasonably likely to materially adversely affect, the Company's internal control over financial reporting.

(xxxi) <u>Tests and Preclinical and Clinical Trials</u>. The studies, tests and preclinical and clinical trials that are described in the Registration Statement, the General Disclosure Package and the Prospectus were and, if still pending, are being, conducted in all material respects in

12

accordance with the protocols submitted to the FDA or any foreign governmental body exercising comparable authority, procedures and controls pursuant to, where applicable, accepted professional and scientific standards, and all applicable laws and regulations; the descriptions of the studies, tests and preclinical and clinical trials conducted by or, to the Company's knowledge, on behalf of the Company, and the results thereof, contained in the Registration Statement, the General Disclosure Package and the Prospectus are accurate and complete in all material respects; the Company is not aware of any other studies, tests or preclinical and clinical trials, the results of which call into question the results described in the Registration Statement, the General Disclosure Package and the Prospectus; and the Company has not received any notices or correspondence from the FDA, any foreign, state or local governmental body exercising comparable authority or any Institutional Review Board requiring the termination, suspension, material modification or clinical hold of any studies, tests or preclinical or clinical trials conducted by or on behalf of the Company, other than ordinary course communications with respect to modifications in connection with the design and implementation of such trials, copies of which communications have been made available to you.

(xxxii) <u>Payment of Taxes</u>. All United States federal income tax returns of the Company and its Subsidiaries required by law to be filed have been filed and all taxes shown by such returns or otherwise assessed, which are due and payable, have been paid, except assessments against which appeals have been or will be promptly taken and as to which adequate reserves have been provided. The Company and its Subsidiaries have duly filed all other filings and tax returns that are required to have been filed by them pursuant to applicable foreign, state, local or other law, have paid all taxes due pursuant to such returns or pursuant to any assessment received by the Company and its Subsidiaries and made all withholdings, except as otherwise disclosed in the Registration Statement and for such taxes and withholdings, if any, as are being contested in good faith and as to which adequate reserves have been established by the Company, and, duly given all notices and supplied all other information, and kept all records

and information. Except as otherwise disclosed in the Registration Statement, the General Disclosure Package and the Prospectus, there is no tax deficiency that has been, or could reasonably be expected to be, asserted against the Company or its Subsidiaries or any of their respective properties or assets.

(xxxiii) <u>Insurance</u>. The Company and its Subsidiaries carry or are entitled to the benefits of insurance, with financially sound and reputable insurers, in such amounts and covering such risks as is generally maintained by companies of a similar nature and which are engaged in the same or similar business, and all such insurance is in full force and effect. The Company has no reason to believe that it or its Subsidiaries will not be able (A) to renew its existing insurance coverage as and when such policies expire or (B) to obtain comparable coverage from similar institutions as may be necessary or appropriate to conduct its business as now conducted and at a cost that would not result in a Material Adverse Effect. Neither of the Company nor any of its Subsidiaries has been denied any insurance coverage which it has sought or for which it has applied.

(xxxiv) <u>Investment Company Act</u>. The Company is not required, and upon the issuance and sale of the Offered ADSs as herein contemplated and the application of the net proceeds therefrom as described in the Registration Statement, the General Disclosure Package and the Prospectus will not be required, to register as an "investment company" under the Investment Company Act of 1940, as amended (the "1940 Act").

(xxxv) <u>Absence of Manipulation</u>. Neither the Company nor any affiliate of the Company has taken, nor will the Company or any affiliate take, directly or indirectly, any action which is designed, or would reasonably be expected, to cause or result in, or which constitutes,

13

the stabilization or manipulation of the price of any security of the Company to facilitate the sale or resale of the Offered ADSs or to result in a violation of Regulation M.

(xxxvi) Foreign Corrupt Practices Act. None of the Company, its Subsidiaries or, to the knowledge of the Company, any director, officer, agent, employee, affiliate or other person acting on behalf of the Company or its Subsidiaries has taken any action, directly or indirectly, that would result in a violation by such persons of the Foreign Corrupt Practices Act of 1977, as amended, and the rules and regulations thereunder (the "FCPA"), any law, rule, or regulation promulgated to implement the OECD Convention on Combating Bribery of Foreign Public Officials in International Business Transactions, signed December 17, 1997 or any other law, rule or regulation of similar purpose and scope (collectively, the "Anti-Bribery Law"), including, without limitation, making use of the mails or any means or instrumentality of interstate commerce corruptly in furtherance of an offer, payment, promise to pay or authorization of the payment of any money, or other property, gift, promise to give, or authorization of the giving of anything of value to any "foreign official" (as such term is defined in the FCPA) or any foreign political party or official thereof or any candidate for foreign political office, in contravention of the FCPA and the Company and, to the knowledge of the Company, its affiliates have conducted their businesses in compliance with the FCPA and have instituted and maintain policies and procedures designed to ensure, and which are reasonably expected to continue to ensure, continued compliance therewith.

(xxxvii) Compliance with Money Laundering Laws. The operations of the Company and its Subsidiaries are and have been conducted at all times in compliance with applicable financial recordkeeping and reporting requirements of the Currency and Foreign Transactions Reporting Act of 1970, as amended, the money laundering statutes of all jurisdictions, the rules and regulations thereunder and any related or similar rules, regulations or guidelines, issued, administered or enforced by any Governmental Entity, including Title 18 U.S. Code section 1956 and 1957, the Patriot Act, the Bank Secrecy Act, and international anti-money laundering principals or procedures by an intergovernmental group or organization, such as the Financial Action Task Force on Money Laundering, of which the United States is a member and with which designation the United States representative to the group or organization continues to concur, all as amended, and any executive order, directive, or regulation pursuant to the authority of any of the foregoing, or any orders or licenses issued thereunder (collectively, the "Money Laundering Laws"); and no action, suit or proceeding by or before any Governmental Entity involving the Company or any of its Subsidiaries with respect to the Money Laundering Laws is pending or, to the knowledge of the Company, threatened.

(xxxviii) <u>Compliance with OFAC</u>. None of the Company, its Subsidiaries or, to the knowledge of the Company, any director, officer, agent, employee, affiliate or representative of the Company or its Subsidiaries, is an individual or entity ("<u>Person</u>") that (i) has violated or is in violation of any provision laws and regulations imposing U.S. economic sanctions measures, including, but not limited to, the International Emergency Economic Powers Act, the Trading with the Enemy Act, the United Nations Participation Act, and the Syria Accountability and Lebanese Sovereignty Act, all as amended, and any executive order, directive or regulation pursuant to the authority of any of the foregoing, including the regulations of the United States Treasury Department set forth under 31 CFR, Subtitle B, Chapter V, as amended, or any orders or licenses issued thereunder or (ii) is currently the subject or target of any sanctions administered or enforced by the United States Government, including, without limitation, the U.S. Department of the Treasury's Office of Foreign Assets Control, the United Nations Security Council, the European Union, Her Majesty's Treasury, or other relevant sanctions authority (collectively, "<u>Sanctions</u>"), nor is the Company located, organized or resident in a country or territory that is

14

the subject of Sanctions; and the Company will not directly or indirectly use the proceeds of the sale of the Offered ADSs, or lend, contribute or otherwise make available such proceeds to any subsidiaries, joint venture partners or other Person, to fund any activities of or business with any Person, or in any country or territory, that, at the time of such funding, is the subject of Sanctions or in any other manner that will result in a violation by any Person (including any Person participating in the transaction, whether as underwriter, advisor, investor or otherwise) of Sanctions.

(xxxix) <u>Lending Relationship</u>. Except as disclosed in the Registration Statement, the General Disclosure Package and the Prospectus, the Company (i) does not have any material lending or other relationship with any banking or lending affiliate of any Underwriter and (ii) does not intend to use any of the proceeds from the sale of the Offered ADSs to repay any outstanding debt owed to any affiliate of any Underwriter.

(xl) <u>Statistical and Market-Related Data</u>. Any statistical and market-related data included in the Registration Statement, the General Disclosure Package or the Prospectus are based on or derived from sources that the Company believes, after reasonable inquiry, to be reliable and accurate and, to the extent required, the Company has obtained the written consent to the use of such data from such sources.

- (xli) <u>Rating of Debt Securities</u>. The Company has no debt securities or preferred stock that is rated by any "nationally recognized statistical rating organization" (as that term is defined by the Commission for purposes of Rule 436(g)(2) under the 1933 Act).
  - (xlii) <u>Foreign Issuer</u>. The Company is a "foreign private issuer" as defined in Rule 405 under the Securities Act.
- (xliii) <u>Lock-Up Agreements</u>. The Company has procured Lock-Up Agreements, substantially in the form of Exhibit B attached hereto, from each of the individuals and entities listed on Schedule C hereto.
- (xliv) <u>Submission to U.S. Jurisdiction</u>. The Company has the power to submit, and has legally, validly, effectively and irrevocably submitted, to Sections 15 and 16 of this Agreement, and Sections 15 and 16 of this Agreement are valid and binding on the Company in accordance with their terms.
- (xlv) No Stamp Duty. No stamp duties or other issuance or transfer taxes or duties and no capital gains, income, value added, withholding or other taxes are payable by or on behalf of the Underwriters in Denmark solely in connection with (A) the issuance and delivery of the Shares and the Offered ADSs in the manner contemplated by this Agreement and the Prospectus, (B) the sale and delivery by the Underwriters of the Offered ADSs as contemplated herein and the Prospectus or (C) the deposit of the Shares being deposited with the Depositary against the issuance of Offered ADSs to be delivered and/or the ADRs evidencing Offered ADSs.
- (xlvi) <u>Legality</u>. The legality, validity, enforceability or admissibility into evidence of any of the Registration Statement, the General Disclosure Package and the Prospectus, this Agreement or the Offered ADSs in any jurisdiction in which the Company is organized or does business is not dependent upon such document being submitted into, filed or recorded with any court or other authority in any such jurisdiction on or before the date hereof or that any tax, imposition or charge be paid in any such jurisdiction on or in respect of any such document,

15

except for the registration of the capital increases contemplated by this Agreement, which will be registered with the Danish Business Authority.

- (xlvii) <u>Legal Action</u>. Subject to Section 16 of this Agreement, a holder of the Offered ADSs and each Underwriter are each entitled to sue as plaintiff in the courts of the jurisdiction of formation and domicile of the Company for the enforcement of their respective rights under this Agreement and the Offered ADSs and such access to such courts will not be subject to any conditions which are not applicable to residents of such jurisdiction or a company incorporated in such jurisdiction except that plaintiffs not residing in Denmark may be required to guarantee payment of a possible order for payment of costs or damages at the request of the defendant.
- (xlviii) <u>Corporate Transactions</u>. Each of the corporate transactions set out in the framework agreement entered into on July 11, 2014 between the Company and its shareholders, have been or will be duly authorized and approved by the Board of Directors and all shareholders of the Company (either directly or pursuant to power of attorney), is or will be valid, binding and enforceable, and have or will be, in each case completed in accordance with applicable Danish law and, where required, registered with the Danish Business Authority.
- (b) *Officers' Certificates.* Any certificate signed by any officer of the Company or its Subsidiaries delivered to the Representatives or to counsel for the Underwriters shall be deemed a representation and warranty by the Company to each Underwriter as to the matters covered thereby.

## SECTION 2. Sale and Delivery to Underwriters; Closing.

- (a) *Initial ADSs.* On the basis of the representations and warranties herein contained and subject to the terms and conditions herein set forth, the Company agrees to sell to each Underwriter, severally and not jointly, and each Underwriter, severally and not jointly, agrees to purchase from the Company, at the price per Underwritten Share set forth in <u>Schedule A</u>, that number of Underwritten Shares (to be deposited with the Depositary to allow for the delivery of such Underwritten Shares in the form of ADSs) equal to the number of Initial ADSs set forth in <u>Schedule A</u> opposite the name of such Underwritter, plus any additional number of Underwritten Shares (to be deposited with the Depositary to allow for the delivery of such Underwritten Shares in the form of ADSs) which such Underwriter may become obligated to purchase pursuant to the provisions of Section 10 hereof, subject, in each case, to such adjustments among the Underwriters as the Representatives in their sole discretion shall make to eliminate any sales or purchases of fractional ADSs. The Company agrees to take such steps to cause the delivery of the Initial ADSs to the Underwriters upon the deposit of Shares with the Depositary, in the manner contemplated by the Prospectus.
- (b) Option ADSs. In addition, on the basis of the representations and warranties herein contained and subject to the terms and conditions herein set forth, the Company hereby grants an option to the Underwriters, severally and not jointly, to purchase up to an additional [•] Option Shares (to be deposited with the Depositary to allow for the delivery of such Option Shares in the form of ADSs) equal to the number of Option ADSs, at the price per share set forth in Schedule A, less an amount per share equal to any dividends or distributions declared by the Company and payable on the Underwritten Shares but not payable on the Option Shares. The option hereby granted may be exercised for 30 days after the date hereof and may be exercised in whole or in part at any time from time to time upon notice by the Representatives to the Company setting forth the number of Option Shares (to be deposited with the Depositary to allow for the delivery of such Option Shares in the form of ADSs) equal to the number of Option ADSs as to which the several Underwriters are then exercising the option and the time and date of payment and delivery for such Option Shares (to be deposited with the Depositary to allow for the delivery of such Option Shares in the form of ADSs). Any such time and date of payment and delivery (a

16

take such steps to cause the delivery of the Option ADSs to the Underwriters upon the deposit of Shares with the Depositary, as contemplated by the Prospectus.

(c) Payment. Payment of the purchase price (the "<u>Initial Purchase Price</u>") for the Shares (to be deposited with the Depositary to allow for the delivery of such Shares in the form of ADSs and evidenced by American Depositary Receipts ("<u>ADRs</u>")) shall be made in USD denominated funds by wire transfer to a blocked subscription bank account designated by the Company at 9:00 A.M. (New York City time) on the third (fourth, if the pricing occurs after 4:30 P.M. (New York City time) on any given day) business day after the date hereof (unless postponed in accordance with the provisions of Section 10), or such other time not later than ten business days after such date as shall be agreed upon by the Representatives and the Company (such time and date of payment and delivery being herein called "<u>Closing Time</u>"). Delivery of the Underwritten Shares (to be deposited with the Depositary to allow for the delivery of such Underwritten Shares in the form of ADSs and evidenced by ADRs) at the Closing Time shall be made by registration of one or more of the Representatives, on behalf of the Underwriters, of the Underwritten Shares in the register of shareholders of the Company, subject to prior receipt by the Company of the full Initial Purchase Price and registration of the associated capital increase with the Danish Business Authority.

In addition, in the event that any or all of the Option Shares (to be deposited with the Depositary to allow for the delivery of such Option Shares in the form of ADSs and evidenced by ADRs) are purchased by the Underwriters, payment of the purchase price (the "Option Purchase Price" and, collectively with the Initial Purchase Price, the "Purchase Price") for such Option Shares (to be deposited with the Depositary to allow for the delivery of such Option Shares in the form of ADSs and evidenced by ADRs) shall be made in USD denominated funds by wire transfer to a blocked subscription bank account designated by the Company at 9:00 A.M. (New York City time) on each Date of Delivery as specified in the notice from the Representatives to the Company. Delivery of the Option Shares (to be deposited with the Depositary to allow for the delivery of such Option Shares in the form of ADSs) on each such Date of Delivery shall be made by registration of the Underwriters and their respective shares of the Option Shares in the register of shareholders of the Company, subject to prior receipt by the Company of the full Option Purchase Price and registration of the associated capital increase with the Danish Business Authority.

Payment of the Purchase Price shall be made to the Company by wire transfer of immediately available funds to a blocked subscription bank account designated by the Company against subsequent delivery to the Representatives of a copy of the Company's register of shareholders duly updated to reflect the purchase of Shares (to be deposited with the Depositary to allow for the delivery of such Shares in the form of ADSs and evidenced by ADRs) by the Underwriters. The Company agrees (i) that the Company shall only be entitled to receive payment of the net amount (excluding the underwriting discount, but including any expenses payable by the Company pursuant to Section 4 of this Agreement, and (ii) that such net amount can only be released from said blocked subscription account upon the prior written approval from the Representatives (acting on behalf of the Underwriters) (such consent not to be unreasonably withheld and with an obligation to consent immediately upon delivery to the

17

Representatives of a copy of the Company's register of shareholders duly updated to reflect the purchase of the Underwritten Shares and the Option Shares, respectively. Upon release of the net amount from said blocked subscription account to the Company, the Representatives shall be entitled to receive a payment from the Company for all expenses payable pursuant to Section 4 of this Agreement. It is understood that each Underwritter has authorized the Representatives, for its account, to accept delivery of, receipt for, and make payment of the Purchase Price for, the Underwritten Shares (to be deposited with the Depositary to allow for the delivery of such Underwritten Shares in the form of ADSs and evidenced by ADRs), and the Option Shares (to be deposited with the Depositary to allow for the delivery of such Option Shares in the form of ADSs and evidenced by ADRs), if any, which it has agreed to purchase. Leerink and Jefferies, individually and not as a Representative, may (but shall not be obligated to) make payment of the Purchase Price for the Underwritten Shares (to be deposited with the Depositary to allow for the delivery of such Underwritten Shares in the form of ADSs and evidenced by ADRs) or the Option Shares (to be deposited with the Depositary to allow for the delivery of such Option Shares in the form of ADSs and evidenced by ADRs), if any, to be purchased by any Underwriter whose funds have not been received by the Closing Time or the relevant Date of Delivery, as the case may be, but such payment shall not relieve such Underwriter from its obligations hereunder.

# SECTION 3. <u>Covenants of the Company</u>. The Company covenants with each Underwriter as follows:

- (a) Compliance with Securities Regulations and Commission Requests. The Company, subject to Section 3(b), will comply with the requirements of Rule 430A, and will notify the Representatives promptly, and confirm the notice in writing, (i) when any post-effective amendment to the Registration Statement, the ADS Registration Statement, or the 1934 Act Registration Statement shall become effective or any amendment or supplement to the Prospectus shall have been filed, (ii) of the receipt of any comments from the Commission or its staff, (iii) of any request by the Commission or its staff for any amendment to the Registration Statement, the ADS Registration Statement, the 1934 Act Registration Statement or any amendment or supplement to the Prospectus or for additional information, (iv) of the issuance by the Commission of any stop order suspending the effectiveness of the Registration Statement, the ADS Registration Statement, the 1934 Act Registration Statement or any post-effective amendment or of any order preventing or suspending the use of any preliminary prospectus, or of the suspension of the qualification of the Offered ADSs for offering or sale in any jurisdiction, or of the initiation or threatening of any proceedings for any of such purposes or of any examination pursuant to Section 8(d) or 8(e) of the 1933 Act concerning the Registration Statement and (v) if the Company becomes the subject of a proceeding under Section 8A of the 1933 Act in connection with the offering of the Offered ADSs. The Company will effect all filings required under Rule 424(b), in the manner and within the time period required by Rule 424(b) (without reliance on Rule 424(b)(8)), and will take such steps as it deems necessary to ascertain promptly whether the form of prospectus transmitted for filing under Rule 424(b) was received for filing by the Commission and, in the event that it was not, it will promptly file such prospectus. The Company will make every reasonable effort to prevent the issuance of any stop order, preventio
- (b) Continued Compliance with Securities Laws. The Company will comply with the 1933 Act and the 1933 Act Regulations so as to permit the completion of the distribution of the Offered ADSs as contemplated in this Agreement and in the Registration Statement, the General Disclosure Package and the Prospectus. If at any time when a prospectus relating to the Offered ADSs is (or, but for the exception afforded by Rule 172 of the 1933 Act Regulations ("Rule 172"), would be) required by the 1933 Act to be delivered in connection with sales of the Offered ADSs, any event shall occur or condition shall exist as a result of which it is necessary, in the opinion of counsel for the Underwriters or for the Company, to (i) amend the Registration Statement, the ADS Registration Statement or the 1934 Act Registration

Statement in order that the Registration Statement, the ADS Registration Statement or the 1934 Act Registration Statement will not include an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading, (ii) amend or supplement the General Disclosure Package or the Prospectus in order that the General Disclosure Package or the Prospectus, as the case may be, will not include any untrue statement of a material fact or omit to state a material fact necessary in order to make the statements therein not misleading in the light of the circumstances existing at the time it is delivered to a purchaser or (iii) amend the Registration Statement, the ADS Registration Statement, the 1934 Act Registration Statement or amend or supplement the General Disclosure Package or the Prospectus, as the case may be, in order to comply with the requirements of the 1933 Act or the 1933 Act Regulations or the 1934 Act or the 1934 Act Regulations, as the case may be, the Company will promptly (A) give the Representatives notice of such event, (B) prepare any amendment or supplement as may be necessary to correct such statement or omission or to make the Registration Statement, the ADS Registration Statement, the 1934 Act Registration Statement, the General Disclosure Package or the Prospectus comply with such requirements and, a reasonable amount of time prior to any proposed filing or use, furnish the Representatives with copies of any such amendment or supplement and (C) file with the Commission any such amendment or supplement; provided that the Company shall not file or use any such amendment or supplement to which the Representatives or counsel for the Underwriters shall reasonably object. The Company will furnish to the Underwriters such number of copies of such amendment or supplement as the Underwriters may reasonably request. The Company has given the Representatives notice of any filings made pursuant to the 1934 Act or the 1934 Act Regulations within 48 hours prior to the Applicable Time; the Company will give the Representatives notice of its intention to make any such filing from the Applicable Time to the Closing Time and will furnish the Representatives with copies of any such documents a reasonable amount of time prior to such proposed filing, as the case may be, and will not file or use any such document to which the Representatives or counsel for the Underwriters shall reasonably object.

- (c) Delivery of Registration Statements. The Company has furnished or will deliver to the Representatives and counsel for the Underwriters, without charge, copies of the Registration Statement as originally filed and each amendment thereto (including exhibits filed therewith) and copies of all consents and certificates of experts, and will also deliver to the Representatives, without charge, a conformed copy of the Registration Statement as originally filed and each amendment thereto (without exhibits) for each of the Underwriters. The copies of the Registration Statement and each amendment thereto furnished to the Underwriters will be identical to the electronically transmitted copies thereof filed with the Commission pursuant to EDGAR, except to the extent permitted by Regulation S-T.
- (d) *Delivery of Prospectuses*. The Company has delivered to each Underwriter, without charge, as many copies of each preliminary prospectus as such Underwriter reasonably requested, and the Company hereby consents to the use of such copies for purposes permitted by the 1933 Act. The Company will furnish to each Underwriter, without charge, during the period when a prospectus relating to the Offered ADSs is (or, but for the exception afforded by Rule 172, would be) required to be delivered under the 1933 Act, such number of copies of the Prospectus (as amended or supplemented) as such Underwriter may reasonably request. The Prospectus and any amendments or supplements thereto furnished to the Underwriters will be identical to the electronically transmitted copies thereof filed with the Commission pursuant to EDGAR, except to the extent permitted by Regulation S-T.
- (e) Blue Sky Qualifications. The Company will use its best efforts, in cooperation with the Underwriters, to qualify the Offered ADSs for offering and sale under the applicable securities laws of such states and other jurisdictions (domestic or foreign) as the Representatives may reasonably designate and to maintain such qualifications in effect so long as required to complete the distribution of the Offered ADSs; provided, however, that the Company shall not be obligated to file any general consent to service of process or to qualify as a foreign corporation or as a dealer in securities in any jurisdiction in

19

which it is not so qualified or to subject itself to taxation in respect of doing business in any jurisdiction in which it is not otherwise so subject.

- (f) *Rule 158*. The Company will timely file such reports pursuant to the 1934 Act as are necessary in order to make generally available to its securityholders as soon as practicable an earnings statement (which need not be audited) for the purposes of, and to provide to the Underwriters the benefits contemplated by, the last paragraph of Section 11(a) of the 1933 Act and Rule 158 under the 1933 Act.
- (g) *Use of Proceeds*. The Company will use the net proceeds received by it from the issue of the ADSs in all material respects in the manner specified in the Registration Statement, the General Disclosure Package and the Prospectus under "Use of Proceeds."
  - (h) *Listing*. The Company will use its best efforts to list for quotation, and maintain the listing of, the Offered ADSs on the NASDAQ.
  - (i) Transfer Agent. The Company will maintain a transfer agent and a registrar for the Ordinary Shares.
- (j) Deposit of Shares. The Shares will be deposited with the Depositary in accordance with the provisions of the Deposit Agreement and the Stock Lending Agreement and otherwise comply with the Deposit Agreement and the Stock Lending Agreement so that Offered ADSs will be issued by the Depositary against receipt of such Shares and Offered ADSs and/or ADRs evidencing Offered ADSs delivered to the Underwriters at the Closing Time or on a Date of Delivery which is after the Closing Time.
- Restriction on Sale of Securities. During a period of 180 days from the date of the Prospectus, the Company will not, without the prior written consent of the Representatives, directly or indirectly, (i) offer, pledge, sell, contract to sell, sell any option or contract to purchase, purchase any option or contract to sell, grant any option, right or warrant for the purchase or sale of, or otherwise dispose of or transfer, directly or indirectly, any equity securities of the Company or any ADSs ("Company Securities"), or any securities convertible into or exchangeable or exercisable for Company Securities (collectively, the "Lock-Up Securities"), or publicly disclose the intention to do any of the foregoing, or file or cause to be filed any registration statement in connection therewith, under the 1933 Act, or (ii) enter into any swap or any other agreement or any transaction that transfers, in whole or in part, directly or indirectly, any of the economic consequences of ownership of the Lock-Up Securities, whether any such swap or transaction described in clause (i) or (ii) above is to be settled by delivery of Company Securities or other securities, in cash or otherwise. The foregoing sentence shall not apply to (A) (i) the conversion of currently outstanding Company Securities into another class or series of Company Securities as referred to in the Registration Statement, the General Disclosure Package and the Prospectus, (ii) the conversion of currently outstanding warrants exercisable for Company Securities into warrants exercisable for another class or series of Company Securities as referred to in the Registration Statement, the General Disclosure Package and the Prospectus, (iii) the conversion of currently outstanding notes or other debt convertible into Company Securities into Company Securities as referred to in the Registration Statement, the General Disclosure Package and the Prospectus, or (iv) the exercise of currently outstanding warrants to purchase Company Securities into Company Securities, in each case whether prior to, immediately upon or following consummation of the transactions contemplated by this Agreement, (B) the Offered ADSs to be issued and sold hereunder, (C) any Ordinary Shares issued by the Company upon the exercise of an option or warrant or the conversion of a security outstanding on the date hereof and referred to in the Registration Statement, the General Disclosure Package and the Prospectus, (D) any

incentive compensation plan) referred to in the Registration Statement, the General Disclosure Package and the Prospectus; (E) any Ordinary Shares issued pursuant to any non-employee director stock plan or dividend reinvestment plan referred to in the Registration Statement, the General Disclosure Package and the Prospectus; or (F) the filing by the Company of any registration statement on Form S-8 or a successor form thereto.

- (l) If the Representatives, in their sole discretion, agree to release or waive the restrictions set forth in a lock-up agreement described in Section 5(k) hereof for an officer or director of the Company and provide the Company with notice of the impending release or waiver at least three business days before the effective date of the release or waiver, the Company agrees to announce the impending release or waiver by a press release substantially in the form of Exhibit C hereto through a major news service at least two business days before the effective date of the release or waiver.
- (m) Reporting Requirements. The Company, during the period when a Prospectus relating to the Offered ADSs is (or, but for the exception afforded by Rule 172, would be) required to be delivered under the 1933 Act, will file all documents required to be filed with the Commission pursuant to the 1934 Act within the time periods required by the 1934 Act and 1934 Act Regulations. Additionally, the Company shall report the use of proceeds from the issuance of the Offered ADSs as may be required under Rule 463 under the 1933 Act.
- (n) Issuer Free Writing Prospectuses. The Company agrees that, unless it obtains the prior written consent of the Representatives, it will not make any offer relating to the Offered ADSs that would constitute an Issuer Free Writing Prospectus or that would otherwise constitute a "free writing prospectus," or a portion thereof, required to be filed by the Company with the Commission or retained by the Company under Rule 433; provided that the Representatives will be deemed to have consented to the Issuer Free Writing Prospectuses listed on Schedule B-2 hereto and any "road show that is a written communication" within the meaning of Rule 433(d)(8)(i) that has been reviewed by the Representatives. The Company represents that it has treated or agrees that it will treat each such free writing prospectus consented to, or deemed consented to, by the Representatives as an "issuer free writing prospectus," as defined in Rule 433, and that it has complied and will comply with the applicable requirements of Rule 433 with respect thereto, including timely filing with the Commission where required, legending and record keeping. If at any time following issuance of an Issuer Free Writing Prospectus there occurred or occurs an event or development as a result of which such Issuer Free Writing Prospectus conflicted or would conflict with the information contained in the Registration Statement, any preliminary prospectus or the Prospectus or included or would include an untrue statement of a material fact or omitted or would omit to state a material fact necessary in order to make the statements therein, in the light of the circumstances existing at that subsequent time, not misleading, the Company will promptly notify the Representatives and will promptly amend or supplement, at its own expense, such Issuer Free Writing Prospectus to eliminate or correct such conflict, untrue statement or omission.
- (o) <u>Testing-the-Waters Materials</u>. If at any time following the distribution of any Written Testing-the-Waters Communication there occurred or occurs an event or development as a result of which such Written Testing-the-Waters Communication included or would include an untrue statement of a material fact or omitted or would omit to state a material fact necessary in order to make the statements therein, in the light of the circumstances existing at that subsequent time, not misleading, the Company will promptly notify the Representatives and will promptly amend or supplement, at its own expense, such Written Testing-the-Waters Communication to eliminate or correct such untrue statement or omission.
- (p) <u>Emerging Growth Company Status</u>. The Company will promptly notify the Representatives if the Company ceases to be an Emerging Growth Company at any time prior to the later

21

of (i) completion of the distribution of the Offered ADSs within the meaning of the 1933 Act and (ii) completion of the 180-day restricted period referred to in Section 3(i).

- (q) Registration of Capital. The Company will cause the increase of the share capital represented by the issue of Underwritten Shares and Option Shares, if any, to be registered with the Danish Business Authority as soon as practically possible, provided however, that the Company undertakes not to file the application for registration of the Underwritten Shares and Option Shares, if any, with the Danish Business Authority without having obtained prior approval from the Representatives (acting on behalf of the Underwriters).
- (r) <u>Manipulation</u>. The Company will not at any time, directly or indirectly, take any action designed, or which might reasonably be expected to cause or result in, or which will constitute, stabilization or manipulation of the price of the ADSs or the Ordinary Shares to facilitate the sale or resale of any of the Underwritten Shares.

## SECTION 4. <u>Payment of Expenses</u>.

(a) Expenses. The Company will pay or cause to be paid all expenses incident to the performance of its obligations under this Agreement, including (i) the preparation, printing and filing of the Registration Statement (including financial statements and exhibits), the ADS Registration Statement and the 1934 Act Registration Statement, each as originally filed and each amendment thereto, (ii) the preparation, printing and delivery to the Underwriters of copies of each preliminary prospectus, each Issuer Free Writing Prospectus and the Prospectus and any amendments or supplements thereto and any costs associated with electronic delivery of any of the foregoing by the Underwriters to investors, (iii) the preparation, issuance and delivery of the certificates, if any, for the Shares and Offered ADSs, including any stock or other transfer taxes and any stamp or other duties payable upon the sale, issuance or delivery of the Shares and Offered ADSs, (iv) the fees and disbursements of the Company's counsel, accountants and other advisors, (v) the qualification of the Shares and Offered ADSs under securities laws in accordance with the provisions of Section 3(e) hereof, including filing fees and the reasonable fees and disbursements of counsel for the Underwriters in connection therewith and in connection with the preparation of a "Blue Sky Survey" and any supplement thereto up to a maximum of \$10,000, (vi) the fees and expenses of any transfer agent or registrar for the Offered ADSs, (vii) the costs and expenses incurred by the Company relating to investor presentations on any "road show" undertaken in connection with the marketing of the Offered ADSs, including without limitation, expenses associated with the production of road show slides and graphics, fees and expenses of any consultants engaged by the Company in connection with the road show presentations, and travel and lodging expenses of the representatives (other than the Underwriters) and officers of the Company and any such consultants, (viii) the filing fee

and expenses incurred in connection with the listing of the Offered ADSs on the NASDAQ, and (x) the costs and expenses (including, without limitation, any damages or other amounts payable in connection with legal or contractual liability) associated with the reforming of any contracts for sale of the Offered ADSs made by the Underwriters caused by a breach of the representation contained in the third sentence of Section 1(a)(ii). It is understood, however, that except as provided in this Section, the Underwriters will pay all of their costs and expenses incident to the performance of their obligations hereunder, including fees and disbursements of their counsel, and the travel expenses of their own representatives in connection with any "roadshow" presentation to potential investors. Further, the Underwriters and Company will each pay 50% of the costs of any jointly used and chartered aircraft in the "roadshow", if the use of such chartered aircraft is determined to be cost effective (as regards the Company, such determination shall be approved in advance in writing by the Chairman of the Company).

22

- (b) *Termination of Agreement*. If this Agreement is terminated by the Representatives in accordance with the provisions of Section 5, Section 9(a)(i) or (iii) or Section 10 hereof, the Company shall reimburse the Underwriters for all of their reasonably documented out-of-pocket expenses, including the reasonable fees and disbursements of counsel for the Underwriters.
- (c) Tax Indemnity. The Company will indemnify and hold harmless the Underwriters against any documentary, stamp, registration or similar issuance tax, including any interest and penalties, on the creation and issuance of the Shares by the Company and the sale of the Offered ADSs by the Underwriters, on the execution and delivery of this Agreement and the deposit of the Shares being deposited with the Depositary. All indemnity payments to be made by the Company hereunder in respect of this Section 4(c) shall be made without withholding or deduction for or on account of any present or future Danish taxes, duties or governmental charges whatsoever unless the Company is compelled by law to deduct or withhold such taxes, duties or charges. In that event, and except for any net income, capital gains or franchise taxes imposed on the Underwriters by Denmark or the United States or any political subdivision or taxing authority thereof or therein as a result of any present or former connection (other than any connection resulting from the transactions contemplated by this Agreement) between the Underwriters and the jurisdiction imposing such withholding or deduction, the Company shall pay such additional amounts as may be necessary in order to ensure that the net amounts received after such withholding or deduction shall equal the amounts that would have been received if no withholding or deduction had been made.
- SECTION 5. <u>Conditions of Underwriters' Obligations</u>. The obligations of the several Underwriters hereunder are subject to the accuracy of the representations and warranties of the Company contained herein or in certificates of any officer of the Company or its Subsidiaries delivered pursuant to the provisions hereof, to the performance by the Company of its covenants and other obligations hereunder, and to the following further conditions:
- (a) Effectiveness of Registration Statement, the ADS Registration Statement and the 1934 Act Registration Statement; Rule 430A Information. The Registration Statement, including any Rule 462(b) Registration Statement, the ADS Registration Statement and the 1934 Act Registration Statement have become effective and, at the Closing Time, no stop order suspending the effectiveness of the Registration Statement, the ADS Registration Statement, the 1934 Act Registration Statement or any post-effective amendment thereto has been issued under the 1933 Act or the 1934 Act, as the case may be, no order preventing or suspending the use of any preliminary prospectus or the Prospectus has been issued and no proceedings for any of those purposes have been instituted or are pending or, to the Company's knowledge, contemplated; and the Company has complied with each request (if any) from the Commission for additional information. A prospectus containing the Rule 430A Information shall have been filed with the Commission in the manner and within the time frame required by Rule 424(b) without reliance on Rule 424(b)(8) or a post-effective amendment providing such information shall have been filed with, and declared effective by, the Commission in accordance with the requirements of Rule 430A.
- (b) *Opinion of Counsel for Company; Certificates*. At the Closing Time, the Representatives shall have received the opinion, dated the Closing Time, of Dechert LLP, U.S. counsel for the Company, together with the opinion of Nielsen Nørager Law Firm LLP, Danish counsel for the Company, the opinion of Deloitte, Danish tax counsel for the Company, the opinion of Emmet, Marvin & Martin LLP, counsel for the Depositary, the opinion of Noerr LLP, German counsel for Forward Pharma GmbH, the opinion of Hyman, Phelps & McNamara, P.C., U.S. special counsel for the Company with respect to U.S. regulatory matters, and the certificate of Dr. Ulrich Granzer, E.U. special consultant to the Company with respect to E.U. regulatory matters, each in form and substance satisfactory to counsel for the Underwriters, together with signed or reproduced copies of such letter for each of the other Underwriters to the effect set forth in Exhibits A-1, A-2, A-3, A-4, A-5, A-6 and A-7 hereto, respectively, and such

23

other opinions from such firms and to such further effect as counsel to the Underwriters may reasonably request.

- (c) Opinion of Counsel for Underwriters. At the Closing Time, the Representatives shall have received the opinions, dated the Closing Time, of K&L Gates LLP, U.S. counsel for the Underwriters, and Plesner, Danish counsel for the Underwriters, together with signed or reproduced copies of such letters for each of the other Underwriters with respect to such matters as the Representatives may reasonably request. In giving such opinions such counsel may rely, as to all matters governed by the laws of jurisdictions other than the law of the State of New York and the federal securities laws of the United States or the laws of Denmark (as the case may be), upon the opinions of counsel satisfactory to the Representatives. Such counsel may also state that, insofar as such opinion involves factual matters, they have relied, to the extent they deem proper, upon certificates of officers and other representatives of the Company and its Subsidiaries and certificates of public officials.
- (d) Officers' Certificate. At the Closing Time, there shall not have been, since the date hereof or since the respective dates as of which information is given in the Registration Statement, the General Disclosure Package or the Prospectus, any material adverse change in the condition, financial or otherwise, or in the earnings, business affairs or business prospects of the Company and its Subsidiaries considered as one enterprise, whether or not arising in the ordinary course of business, and the Representatives shall have received a certificate of the Chief Executive Officer of the Company, of the Chief Financial Officer of the Company and of the chairman of the board of directors of the Company, dated the Closing Time, to the effect that (i) there has been no such Material Adverse Change, (ii) the representations and warranties of the Company in this Agreement are true and correct with the same force and effect as though expressly made at and as of the Closing Time, (iii) the Company has complied with all agreements and satisfied all conditions on its part to be performed or satisfied at or prior to the Closing Time, and (iv) no stop order suspending the effectiveness of the Registration Statement, the ADS Registration Statement or the 1934 Act Registration Statement under the 1933 Act or the 1934 Act, as the case may be, has been issued, no order preventing or suspending the use of any preliminary prospectus or the Prospectus has been issued and no proceedings for any of those purposes have been instituted or are pending or, to their knowledge, contemplated.

- (e) Accountant's Comfort Letter. At the time of the execution of this Agreement, the Representatives shall have received from Ernst & Young Denmark P/S a letter, dated such date, in form and substance satisfactory to the Representatives, together with signed or reproduced copies of such letter for each of the other Underwriters containing statements and information of the type ordinarily included in accountants' "comfort letters" to underwriters with respect to the financial statements and certain financial information contained in the Registration Statement, the General Disclosure Package and the Prospectus.
- (f) Bring-down Comfort Letter. At the Closing Time, the Representatives shall have received from Ernst & Young Denmark P/S a letter, dated as of the Closing Time, to the effect that they reaffirm the statements made in the letter furnished pursuant to subsection (e) of this Section, except that the specified date referred to shall be a date not more than three business days prior to the Closing Time.
- (g) *Approval of Listing*. At the Closing Time, the Offered ADSs shall have been approved for listing on the NASDAQ, subject only to official notice of issuance.
- (h) *No Objection.* FINRA shall have confirmed that it has not raised any objection with respect to the fairness and reasonableness of the underwriting terms and arrangements relating to the offering of the Offered ADSs.

24

- (i) *Lock-up Agreements*. At the date of this Agreement, the Representatives shall have received an agreement substantially in the form of Exhibit B hereto signed by the persons listed on Schedule C hereto.
- (j) *Certificates*. The Depositary shall have furnished or caused to be furnished to the Representatives on and as of the Closing Time certificates, dated the date of delivery thereof, and/or evidence reasonably satisfactory to the Representatives evidencing the deposit with it or its nominee of the Shares being so deposited against issuance of the Offered ADSs to be delivered by the Company on and as of the Closing Date, and the execution, countersignature (if applicable), issuance and delivery of such Offered ADSs pursuant to the Deposit Agreement.
- (k) Share Conversion. Prior to the Underwriters' purchase of Offered ADSs pursuant to this Agreement, all shares in the Company shall have been converted to one share class (ordinary shares), any outstanding convertible bridge financings shall have been converted to ordinary shares and a share split shall have been completed; all to the effect that prior to the Underwriters' purchase of Offered ADSs pursuant to this Agreement, the outstanding share capital of the Company will consist of 35,313,760 ordinary shares of DKK 0.10 and there will be no convertible bridge financing outstanding.
- (l) *Conditions to Purchase of Option ADSs.* In the event that the Underwriters exercise their option provided in Section 2(b) hereof to purchase all or any portion of the Option ADSs, the representations and warranties of the Company contained herein and the statements in any certificates furnished by the Company and its Subsidiaries hereunder shall be true and correct as of each Date of Delivery and, at the relevant Date of Delivery, the Representatives shall have received:
  - (i) Officers' Certificate. A certificate, dated such Date of Delivery, of the Chief Executive Officer of the Company, of the Chief Financial Officer of the Company and of the chairman of the board of directors of the Company confirming that the certificate delivered at the Closing Time pursuant to Section 5(d) hereof remains true and correct as of such Date of Delivery.
  - (ii) Opinion of Counsel for Company. If requested by the Representatives, the opinion of Dechert LLP, counsel for the Company, together with the opinion of Nielsen Nørager Law Firm LLP, Danish counsel for the Company, the opinion of Deloitte, Danish tax counsel for the Company, the opinion of Emmet, Marvin & Martin LLP, counsel for the Depositary, the opinion of Noerr LLP, German counsel for Forward Pharma GmbH, the opinion of Hyman, Phelps & McNamara, P.C., U.S. special counsel for the Company with respect to U.S. regulatory matters, and the certificate of Dr. Ulrich Granzer, E.U. special consultant to the Company with respect to E.U. regulatory matters, each in form and substance satisfactory to counsel for the Underwriters, dated such Date of Delivery, relating to the Option ADSs to be purchased on such Date of Delivery and otherwise to the same effect as the opinions required by Section 5(b) hereof, and such other opinions from such firms and to such further effect as counsel to the Underwriters may reasonably request.
  - (iii) <u>Opinion of Counsel for Underwriters</u>. If requested by the Representatives, the opinions of K&L Gates LLP, U.S. counsel for the Underwriters, and Plesner, Danish counsel for the Underwriters, each dated such Date of Delivery, relating to the Option ADSs to be purchased on such Date of Delivery and otherwise to the same effect as the opinion required by Section 5(c) hereof.
  - (iv) <u>Bring-down Comfort Letter</u>. If requested by the Representatives, a letter from Ernst & Young Denmark P/S, in form and substance satisfactory to the Representatives and dated

25

such Date of Delivery, substantially in the same form and substance as the letter furnished to the Representatives pursuant to Section 5(e) hereof, except that the "specified date" in the letter furnished pursuant to this paragraph shall be a date not more than three business days prior to such Date of Delivery.

- (v) <u>Certificates</u>. The Depositary shall have furnished or caused to be furnished to the Representatives on and as of such Date of Delivery certificates, dated such Date of Delivery, and/or evidence reasonably satisfactory to the Representatives evidencing the deposit with it or its nominee of the Shares being so deposited against issuance of the Option ADSs to be delivered by the Company on and as of such Date of Delivery, and the execution, countersignature (if applicable), issuance and delivery of such Option ADSs pursuant to the Deposit Agreement.
- (m) Additional Documents. At the Closing Time and at each Date of Delivery (if any) counsel for the Underwriters shall have been furnished with such documents and opinions as they may reasonably require for the purpose of enabling them to pass upon the issuance and sale of the Offered ADSs as herein contemplated, or in order to evidence the accuracy of any of the representations or warranties, or the fulfillment of any of the conditions, herein contained; and all proceedings taken by the Company in connection with the issuance and sale of the Offered ADSs as herein contemplated shall be reasonably satisfactory in form and substance to the Representatives and counsel for the Underwriters.

(n) *Termination of Agreement.* If any condition specified in this Section shall not have been fulfilled when and as required to be fulfilled, this Agreement, or, in the case of any condition to the purchase of Option ADSs on a Date of Delivery which is after the Closing Time, the obligations of the several Underwriters to purchase the relevant Option ADSs, may be terminated by the Representatives by notice to the Company at any time at or prior to Closing Time or such Date of Delivery, as the case may be, and such termination shall be without liability of any party to any other party except as provided in Section 4 and except that Sections 1, 6, 7, 8, 14, 15 and 16 shall survive any such termination and remain in full force and effect.

## SECTION 6. Indemnification.

- (a) *Indemnification of Underwriters*. The Company agrees to indemnify and hold harmless each Underwriter, its affiliates (as such term is defined in Rule 501(b) under the 1933 Act (each, an "Affiliate")), its selling agents, directors, officers, members and representatives and each person, if any, who controls any Underwriter within the meaning of Section 15 of the 1933 Act or Section 20 of the 1934 Act as follows:
  - (i) against any and all loss, liability, claim, damage and expense whatsoever, as incurred, arising out of any untrue statement or alleged untrue statement of a material fact contained in the ADS Registration Statement, the 1934 Act Registration Statement, or the Registration Statement (or any amendment thereto), including the Rule 430A Information, or the omission or alleged omission therefrom of a material fact required to be stated therein or necessary to make the statements therein not misleading or arising out of any untrue statement or alleged untrue statement of a material fact included (A) in any preliminary prospectus, any Issuer Free Writing Prospectus, any Written Testing-the-Waters Communication, the General Disclosure Package or the Prospectus (or any amendment or supplement thereto), or (B) in any materials or information provided to investors by, or with the approval of, the Company in connection with the marketing of the offering of the ADSs ("Marketing Materials"), including any roadshow or investor presentations made to investors by the Company (whether in person or electronically), or the omission or alleged omission in any preliminary prospectus, Issuer Free Writing Prospectus, any Written Testing-the-Waters Communication, Prospectus or in any

26

Marketing Materials of a material fact necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading;

- (ii) against any and all loss, liability, claim, damage and expense whatsoever, as incurred, to the extent of the aggregate amount paid in settlement of any litigation, or any investigation or proceeding by any governmental agency or body, commenced or threatened, or of any claim whatsoever based upon any such untrue statement or omission, or any such alleged untrue statement or omission; provided that (subject to Section 6(d) below) any such settlement is effected with the written consent of the Company;
- (iii) against any and all expense whatsoever, as incurred (including the fees and disbursements of counsel chosen by the Representatives), reasonably incurred in investigating, preparing or defending against any litigation, or any investigation or proceeding by any governmental agency or body, commenced or threatened, or any claim whatsoever based upon any such untrue statement or omission, or any such alleged untrue statement or omission, to the extent that any such expense is not paid under (i) or (ii) above;

provided, however, that this obligation of indemnity shall not apply to any loss, liability, claim, damage or expense to the extent arising out of any untrue statement or omission or alleged untrue statement or omission made in any preliminary prospectus, the Marketing Materials, the ADS Registration Statement, the 1934 Act Registration Statement, the Registration Statement (or any amendment thereto), including the Rule 430A Information, the General Disclosure Package or the Prospectus (or any amendment or supplement thereto), any Issuer Free Writing Prospectus or any Written Testing-the-Waters Communication, in reliance upon and in conformity with the Underwriter Information.

- (b) *Indemnification of Company, Directors and Officers*. Each Underwriter severally agrees to indemnify and hold harmless the Company, its directors, each of its officers who signed the Registration Statement, and each person, if any, who controls the Company within the meaning of Section 15 of the 1933 Act or Section 20 of the 1934 Act, against any and all loss, liability, claim, damage and expense described in the indemnity contained in subsection (a) of this Section, as incurred, but only with respect to untrue statements or omissions, or alleged untrue statements or omissions, made in the Registration Statement (or any amendment thereto), including the Rule 430A Information, the General Disclosure Package or the Prospectus (or any amendment or supplement thereto) in reliance upon and in conformity with the Underwriter Information.
- (c) Actions against Parties; Notification. Each indemnified party shall give notice as promptly as reasonably practicable to each indemnifying party of any action commenced against it in respect of which indemnity may be sought hereunder, but failure to so notify an indemnifying party shall not relieve such indemnifying party from any liability hereunder to the extent it is not materially prejudiced as a result thereof and in any event shall not relieve it from any liability which it may have otherwise than on account of this obligation of indemnity. In the case of parties indemnified pursuant to Section 6(a) above, counsel to the indemnified parties shall be selected by the Representatives, and, in the case of parties indemnified pursuant to Section 6(b) above, counsel to the indemnified parties shall be selected by the Company. An indemnifying party may participate at its own expense in the defense of any such action; provided, however, that counsel to the indemnifying party shall not (except with the consent of the indemnified party) also be counsel to the indemnified party. In no event shall the indemnifying parties be liable for the reasonable fees and expenses of more than one counsel (in addition to any local counsel) separate from their own counsel for all indemnified parties in connection with any one action or separate but similar or related actions in the same jurisdiction arising out of the same general allegations or circumstances. No indemnifying party shall, without the prior written consent of the indemnified parties, settle or compromise or consent to the entry of any judgment with respect to any

27

litigation, or any investigation or proceeding by any governmental agency or body, commenced or threatened, or any claim whatsoever in respect of which indemnification or contribution could be sought under this Section 6 or Section 7 hereof (whether or not the indemnified parties are actual or potential parties thereto), unless such settlement, compromise or consent (i) includes an unconditional release of each indemnified party from all liability arising out of such litigation, investigation, proceeding or claim and (ii) does not include a statement as to or an admission of fault, culpability or a failure to act by or on behalf of any indemnified party.

(d) Settlement without Consent if Failure to Reimburse. If at any time an indemnified party shall have requested an indemnifying party to reimburse the indemnified party for fees and expenses of counsel, such indemnifying party agrees that it shall be liable for any settlement of the nature contemplated by Section 6(a)(ii) effected without its written consent if (i) such settlement is entered into more than 45 days after receipt by such indemnifying party of the aforesaid request, (ii) such indemnifying party shall have received notice of the terms of such settlement at least 30 days prior to such settlement being entered into and (iii) such indemnifying party shall not have reimbursed such indemnified party in accordance with such request prior to the date of such settlement.

SECTION 7. <u>Contribution</u>. If the indemnification provided for in Section 6 hereof is for any reason unavailable to or insufficient to hold harmless an indemnified party in respect of any losses, liabilities, claims, damages or expenses referred to therein, then each indemnifying party shall contribute to the aggregate amount of such losses, liabilities, claims, damages and expenses incurred by such indemnified party, as incurred, (i) in such proportion as is appropriate to reflect the relative benefits received by the Company, on the one hand, and the Underwriters, on the other hand, from the offering of the Offered ADSs pursuant to this Agreement or (ii) if the allocation provided by clause (i) is not permitted by applicable law, in such proportion as is appropriate to reflect not only the relative benefits referred to in clause (i) above but also the relative fault of the Company, on the one hand, and of the Underwriters, on the other hand, in connection with the statements or omissions which resulted in such losses, liabilities, claims, damages or expenses, as well as any other relevant equitable considerations.

The relative benefits received by the Company, on the one hand, and the Underwriters, on the other hand, in connection with the offering of the Offered ADSs pursuant to this Agreement shall be deemed to be in the same respective proportions as the total net proceeds from the offering of the Offered ADSs pursuant to this Agreement (before deducting expenses) received by the Company, on the one hand, and the total underwriting discount received by the Underwriters, on the other hand, in each case as set forth on the cover of the Prospectus, bear to the aggregate initial public offering price of the Offered ADSs as set forth on the cover of the Prospectus.

The relative fault of the Company, on the one hand, and the Underwriters, on the other hand, shall be determined by reference to, among other things, whether any such untrue or alleged untrue statement of a material fact or omission or alleged omission to state a material fact relates to information supplied by the Company or by the Underwriters and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission.

The Company and the Underwriters agree that it would not be just and equitable if contribution pursuant to this Section 7 were determined by pro rata allocation (even if the Underwriters were treated as one entity for such purpose) or by any other method of allocation which does not take account of the equitable considerations referred to above in this Section 7. The aggregate amount of losses, liabilities, claims, damages and expenses incurred by an indemnified party and referred to above in this Section 7 shall be deemed to include any legal or other expenses reasonably incurred by such indemnified party in investigating, preparing or defending against any litigation, or any investigation or proceeding by any

28

governmental agency or body, commenced or threatened, or any claim whatsoever based upon any such untrue or alleged untrue statement or omission or alleged omission.

Notwithstanding the provisions of this Section 7, no Underwriter shall be required to contribute any amount in excess of the underwriting commissions (including the underwriting discount) received by such Underwriter in connection with the Offered ADSs underwritten by it and distributed to the public.

No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the 1933 Act) shall be entitled to contribution from any person who was not guilty of such fraudulent misrepresentation.

For purposes of this Section 7, each person, if any, who controls an Underwriter within the meaning of Section 15 of the 1933 Act or Section 20 of the 1934 Act and each Underwriter's Affiliates and selling agents shall have the same rights to contribution as such Underwriter, and each director of the Company, each officer of the Company who signed the Registration Statement, and each person, if any, who controls the Company within the meaning of Section 15 of the 1933 Act or Section 20 of the 1934 Act shall have the same rights to contribution as the Company. The Underwriters' respective obligations to contribute pursuant to this Section 7 are several in proportion to the number of Initial ADSs set forth opposite their respective names in Schedule A hereto and not joint.

SECTION 8. <u>Representations, Warranties and Agreements to Survive</u>. All representations, warranties and agreements contained in this Agreement or in certificates of officers of the Company or its Subsidiaries submitted pursuant hereto, shall remain operative and in full force and effect regardless of (i) any investigation made by or on behalf of any Underwriter or its Affiliates or selling agents, any person controlling any Underwriter, its officers or directors or any person controlling the Company and (ii) delivery of and payment for the Offered ADSs.

# SECTION 9. <u>Effective Date of Agreement; Termination of Agreement.</u>

- (a) This Agreement shall become effective when the parties hereto have executed and delivered this Agreement.
- (b) Termination. The Representatives may terminate this Agreement, by prompt notice to the Company in writing, at any time at or prior to the Closing Time (i) if there has been, in the judgment of the Representatives, since the time of execution of this Agreement or since the respective dates as of which information is given in the Registration Statement, the General Disclosure Package or the Prospectus, any material adverse change in the condition, financial or otherwise, or in the earnings, business affairs or business prospects of the Company and its Subsidiaries considered as one enterprise, whether or not arising in the ordinary course of business, or (ii) if there has occurred any material adverse change in the financial markets in the United States or the international financial markets, any outbreak of hostilities or escalation thereof or other calamity or crisis or any change or development involving a prospective change in national or international political, financial or economic conditions, in each case the effect of which is such as to make it, in the judgment of the Representatives, impracticable or inadvisable to proceed with the completion of the offering or to enforce contracts for the sale of the Offered ADSs, or (iii) if trading in any securities of the Company has been suspended or materially limited by the Commission or the NASDAQ, or (iv) if trading generally on the NYSE Amex or the New York Stock Exchange or the NASDAQ has been suspended or materially limited, or minimum or maximum prices for trading have been fixed, or maximum ranges for prices have been required, by any of said exchanges or by order of the Commission, FINRA or any other governmental authority, or (v) a material disruption has occurred in commercial banking or securities settlement or clearance services in the United States, or (vi) if a banking moratorium has been declared by either Federal or New York authorities.

- (c) *Liabilities.* If this Agreement is terminated pursuant to this Section, such termination shall be without liability of any party to any other party except as provided in Section 4 hereof, and provided further that Sections 1, 6, 7, 8, 14, 15 and 16 shall survive such termination and remain in full force and effect. In addition, if this Agreement is terminated for any reason, the Lock-Up Agreements shall be automatically terminated.
- SECTION 10. <u>Default by One or More of the Underwriters</u>. If one or more of the Underwriters shall fail at the Closing Time or a Date of Delivery to purchase the Offered ADSs which it or they are obligated to purchase under this Agreement (the "<u>Defaulted Securities</u>"), the Representatives shall have the right, within 24 hours thereafter, to make arrangements for one or more of the non-defaulting Underwriters, or any other underwriters, to purchase all, but not less than all, of the Defaulted Securities in such amounts as may be agreed upon and upon the terms herein set forth; if, however, the Representatives shall not have completed such arrangements within such 24-hour period, then:
  - (i) if the number of Defaulted Securities does not exceed 10% of the number of Securities to be purchased on such date, each of the non-defaulting Underwriters shall be obligated, severally and not jointly, to purchase the full amount thereof in the proportions that their respective underwriting obligations hereunder bear to the underwriting obligations of all non-defaulting Underwriters, or
  - (ii) if the number of Defaulted Securities exceeds 10% of the number of Offered ADSs to be purchased on such date, this Agreement or, with respect to any Date of Delivery which occurs after the Closing Time, the obligation of the Underwriters to purchase, and the Company to sell, the Option ADSs to be purchased and sold on such Date of Delivery shall terminate without liability on the part of any non-defaulting Underwriter.

No action taken pursuant to this Section shall relieve any defaulting Underwriter from liability in respect of its default.

In the event of any such default which does not result in a termination of this Agreement or, in the case of a Date of Delivery which is after the Closing Time, which does not result in a termination of the obligation of the Underwriters to purchase and the Company to sell the relevant Option ADSs, as the case may be, either the (i) Representatives or (ii) the Company shall have the right to postpone Closing Time or the relevant Date of Delivery, as the case may be, for a period not exceeding seven days in order to effect any required changes in the Registration Statement, the General Disclosure Package or the Prospectus or in any other documents or arrangements. As used herein, the term "Underwriter" includes any person substituted for an Underwriter under this Section 10.

- SECTION 11. Notices. All notices and other communications hereunder shall be in writing and shall be deemed to have been duly given if mailed or transmitted by any standard form of telecommunication. Notices to the Underwriters shall be directed to Leerink at One Federal Street, 37th Floor, Boston, MA 02110, attention: Timothy A.G. Gerhold, General Counsel and Jefferies at 520 Madison Avenue, New York, NY 10022, attention: General Counsel; notices to the Company shall be directed to it at Østergade 24A, 1, 1100 Copenhagen K, Denmark, attention of Florian Schönharting, Chairman.
- SECTION 12. No Advisory or Fiduciary Relationship. The Company acknowledges and agrees that (a) the purchase and sale of the Offered ADSs pursuant to this Agreement, including the determination of the initial public offering price of the Offered ADSs and any related discounts and commissions, is an arm's-length commercial transaction between the Company, on the one hand, and the several Underwriters, on the other hand, (b) in connection with the offering of the Offered ADSs and the

30

process leading thereto, each Underwriter is and has been acting solely as a principal and is not the agent or fiduciary of the Company, its Subsidiaries or their respective stockholders, creditors, employees or any other party, (c) no Underwriter has assumed or will assume an advisory or fiduciary responsibility in favor of the Company with respect to the offering of the Offered ADSs or the process leading thereto (irrespective of whether such Underwriter has advised or is currently advising the Company or its Subsidiaries on other matters) and no Underwriter has any obligation to the Company with respect to the offering of the Offered ADSs except the obligations expressly set forth in this Agreement, (d) the Underwriters and their respective affiliates may be engaged in a broad range of transactions that involve interests that differ from those of the Company and (e) the Underwriters have not provided any legal, accounting, regulatory or tax advice with respect to the offering of the Offered ADSs and the Company has consulted its own respective legal, accounting, regulatory and tax advisors to the extent it deemed appropriate.

- SECTION 13. <u>Parties</u>. This Agreement shall each inure to the benefit of and be binding upon the Underwriters and the Company and their respective successors. Nothing expressed or mentioned in this Agreement is intended or shall be construed to give any person, firm or corporation, other than the Underwriters and the Company and their respective successors and the controlling persons and officers and directors referred to in Sections 6 and 7 and their heirs and legal representatives, any legal or equitable right, remedy or claim under or in respect of this Agreement or any provision herein contained. This Agreement and all conditions and provisions hereof are intended to be for the sole and exclusive benefit of the Underwriters and the Company and their respective successors, and said controlling persons and officers and directors and their heirs and legal representatives, and for the benefit of no other person, firm or corporation. No purchaser of Offered ADSs from any Underwriter shall be deemed to be a successor by reason merely of such purchase.
- SECTION 14. <u>Waiver of Trial by Jury</u>. The Company (on its behalf and, to the extent permitted by applicable law, on behalf of its stockholders and affiliates) and each of the Underwriters hereby irrevocably waives, to the fullest extent permitted by applicable law, any and all right to trial by jury in any legal proceeding arising out of or relating to this Agreement or the transactions contemplated hereby.
- SECTION 15. <u>GOVERNING LAW</u>. THIS AGREEMENT AND ANY CLAIM, CONTROVERSY OR DISPUTE BASED UPON, ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF, THE STATE OF NEW YORK WITHOUT REGARD TO ITS CONFLICTS OF LAWS PROVISIONS.
- SECTION 16. <u>Consent to Jurisdiction; Waiver of Immunity.</u> Any legal suit, action or proceeding based upon, arising out of or related to this Agreement or the transactions contemplated hereby shall be instituted in (i) the United States Federal District Court for the Southern District of New York or (ii) the Supreme Court of the State of New York, New York County (collectively, the "Specified Courts"), and each party irrevocably submits to the exclusive jurisdiction (except for proceedings instituted in regard to the enforcement of a judgment of any such court, as to which such jurisdiction is non-exclusive) of

the Specified Courts in any such suit, action or proceeding. The parties irrevocably and unconditionally waive and agree not to plead or assert any objection
to personal jurisdiction or venue with respect to any such suit, action or other proceeding in the Specified Courts, and irrevocably and unconditionally waive
and agree not to plead or assert that any such suit, action or proceeding brought in any Specified Court has been brought in an inconvenient forum. The
Company irrevocably appoints CT Corporation System, located at 1015 15th Street, NW, Suite 1000, Washington, DC 20005, as its authorized agent for
service of process in any such suit, action or proceeding and agrees that service of process upon such authorized agent, and written notice of such service to
the Company, by the person serving the same to the address provided in this Section 16, shall be deemed in every respect effective

31

service of process upon the Company in any such suit, action or proceeding if delivered by overnight courier. The Company irrevocably and unconditionally waives and agrees not to plead or assert any objection to service of process in accordance with the foregoing provision. The Company hereby represents and warrants that such authorized agent has accepted such appointment and has agreed to act as such authorized agent for service of process.

To the extent that the Company may otherwise be entitled, with respect to any claim, controversy, dispute, legal suit, action or proceeding based upon, arising out of or related to this Agreement or the transactions contemplated hereby, to claim for itself or its revenues or assets any immunity, including sovereign immunity, from such claim, controversy, dispute, legal suit, action or proceeding, or from jurisdiction, forum, venue, attachment in aid of execution of a judgment or prior to a judgment, execution of a judgment or any other legal process, or to the extent that there may be attributed to the Company such an immunity whether or not claimed, the Company hereby irrevocably and unconditionally agrees not to claim and irrevocably and unconditionally waives any such immunity and the benefit of any such immunity to the maximum extent permitted by law.

SECTION 17. TIME. TIME SHALL BE OF THE ESSENCE OF THIS AGREEMENT. EXCEPT AS OTHERWISE SET FORTH HEREIN, SPECIFIED TIMES OF DAY REFER TO NEW YORK CITY TIME.

SECTION 18. <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original, but all such counterparts shall together constitute one and the same agreement.

SECTION 19. Effect of Headings. The Section headings herein are for convenience only and shall not affect the construction hereof.

(Signature page follows)

32

If the foregoing is in accordance with your understanding of our agreement, please sign and return to the Company a counterpart hereof, whereupon this instrument, along with all counterparts, will become a binding agreement among the Underwriters and the Company in accordance with its terms.

Very truly yours,

FORWARD PHARMA A/S

By Name: Title:

CONFIRMED AND ACCEPTED,

as of the date first above written:

LEERINK PARTNERS LLC JEFFERIES LLC

LEERINK PARTNERS LLC

By
Name:
Title:

JEFFERIES LLC

Title:

By Name:

For themselves and as Representatives of the other Underwriters named in <u>Schedule A</u> hereto.

Signature Page to Underwriting Agreement

# SCHEDULE A

The initial public offering price per Initial ADS (representing one Underwritten Share) shall be  $\S[\cdot]$ .

The purchase price per share for the Underwritten Shares (to be delivered in the form of Initial ADSs) to be paid by the several Underwriters shall be  $\{\cdot\}$ , being an amount equal to the initial public offering price set forth above less  $\{\cdot\}$  per share, subject to adjustment in accordance with Section 2(b) for dividends or distributions declared by the Company and payable on the Underwritten Shares but not payable on the Option Shares.

Name of Underwriter		Number of Initial ADSs	
	nk Partners LLC		
RBC (	ries LLC Capital Markets, LLC		
JMP S	Securities LLC		
Tot	tal		
	SCHEDULE B-1		
	<u>Pricing Terms</u>		
1.	The Company is selling $[\cdot]$ Offered ADSs.		
2.	. The Company has granted an option to the Underwriters, severally and not jointly, to purchase up to an additional $[\cdot]$ Option ADSs.		
3.	The initial public offering price per Underwritten Share (to be delivered in the form of Initial ADSs) shall be $\{\cdot\}$ .		
	SCHEDULE B-2		
	Free Writing Prospectuses		
	E)		
	SCHEDULE B-3		
	Written Testing-the-Waters Communications		
	SCHEDULE C		
	List of Persons and Entities Subject to Lock-up		
		Exhibit A-1	
	FORM OF OPINION OF COMPANY'S U.S. COUNSEL		
	TO BE DELIVERED PURSUANT TO SECTION 5(b)		
	[·]		

Exhibit A-2

Exhibit A-3 FORM OF OPINION OF COMPANY'S DANISH TAX COUNSEL TO BE DELIVERED PURSUANT TO SECTION 5(b)  $[\cdot]$ Exhibit A-4 FORM OF OPINION OF COUNSEL TO DEPOSITARY TO BE DELIVERED PURSUANT TO SECTION 5(b)  $[\cdot]$ Exhibit A-5 FORM OF OPINION OF SUBSIDIARY'S GERMAN COUNSEL TO BE DELIVERED PURSUANT TO SECTION 5(b)  $[\cdot]$ Exhibit A-6 FORM OF OPINION OF COMPANY'S U.S. REGULATORY MATTERS SPECIAL COUNSEL TO BE DELIVERED PURSUANT TO SECTION 5(b)  $[\cdot]$ Exhibit A-7 FORM OF CERTIFICATE OF COMPANY'S E.U. REGULATORY CONSULTANT TO BE DELIVERED PURSUANT TO SECTION 5(b)  $[\cdot]$ Exhibit B FORM OF LOCK-UP FROM DIRECTORS, OFFICERS OR OTHER SECURITYHOLDERS PURSUANT TO SECTION 5(I)

[·]

Leerink Partners LLC One Federal Street, 37<sup>th</sup> Floor Boston, MA 02110

Jefferies LLC 520 Madison Avenue New York, New York 10022

, as Representatives of the several Underwriters

Re: <u>Proposed Public Offering by Forward Pharma A/S</u>

#### Ladies and Gentlemen:

The undersigned, a securityholder, officer and/or director of Forward Pharma A/S, a limited liability company organized and existing under the laws of the Kingdom of Denmark (the "Company"), understands that Leerink Partners LLC and Jefferies LLC (together, the "Representatives") propose to enter into an Underwriting Agreement (the "Underwriting Agreement") with the Company providing for the public offering (the "Public Offering") of the Company's equity securities. In recognition of the benefit that such an offering will confer upon the undersigned as a securityholder, an officer and/or a director of the Company, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the undersigned agrees with each underwriter to be named in the Underwriting Agreement that, during the period beginning on the date hereof and ending on the date that is 180 days from the date of the Underwriting Agreement (the "Lock-Up Period"), the undersigned will not, without the prior written consent of each Representative, directly or indirectly, (i) offer, pledge, sell, contract to sell, sell any option or contract to purchase any option or contract to sell, grant any option, right or warrant for the purchase or sale of, or otherwise dispose of or transfer, directly or indirectly, any equity securities of the Company ("Company Equity Securities"), or any securities convertible into or exchangeable or exercisable for Company Equity Securities (including, without limitation, such securities which may be deemed to be beneficially owned by the undersigned in accordance with the rules and regulations of the Securities and Exchange Commission ("SEC")), whether now owned or hereafter acquired by the undersigned or with respect to which the undersigned has or hereafter acquires the power of disposition (collectively, the "Lock-Up Securities"), or publicly disclose the intention to do any of the foregoing, or exercise any right with respect to the registration of any of the Lock-Up Securities, or file or cause to be filed any registration statement in connection therewith, under the Securities Act of 1933, as amended, or (ii) enter into any swap or any other agreement or any transaction that transfers, in whole or in part, directly or indirectly, any of the economic consequences of ownership of the Lock-Up Securities, whether any such swap or transaction is to be settled by delivery of Company Equity Securities or other securities, in cash or otherwise. If the undersigned is an officer or director of the Company, the undersigned further agrees that the foregoing provisions shall be equally applicable to any issuer-directed Company Equity Securities the undersigned may purchase in the Public Offering.

The restrictions set forth in this lock-up agreement shall not apply to (i) the conversion of currently outstanding Company Equity Securities into another class or series of Company Equity Securities, (ii) the conversion of currently outstanding warrants exercisable for Company Equity Securities into warrants exercisable for another class or series of Company Equity Securities, (iii) the conversion of notes convertible into Company Equity Securities into Company Equity Securities, in each case whether prior to, immediately upon or following consummation of the Public Offering.

If the undersigned is an officer or director of the Company, (1) the Representatives agree that, at least three business days before the effective date of any release or waiver of the foregoing restrictions in connection with a transfer of Company Equity Securities, the Representatives will notify the Company of the impending release or waiver, and (2) the Company has agreed in the Underwriting Agreement to announce the impending release or waiver by press release through a major news service at least two business days before the effective date of the release or waiver. Any release or waiver granted by the Representatives hereunder to any such officer or director shall only be effective two business days after the publication date of such press release. The provisions of this paragraph will not apply if (i) the release or waiver is effected solely to permit a transfer not for consideration and (ii) the transferee has agreed in writing to be bound by the same terms described in this letter to the extent and for the duration that such terms remain in effect at the time of the transfer.

Notwithstanding the foregoing, and subject to the conditions below, the undersigned may transfer the Lock-Up Securities without the prior written consent of the Representatives:

- (i) as a bona fide gift or gifts; or
- (ii) to any trust for the direct or indirect benefit of the undersigned or the immediate family of the undersigned (for purposes of this lock-up agreement, "immediate family" shall mean any relationship by blood, marriage or adoption, not more remote than first cousin); or
- (iii) by will or intestate succession upon the death of the undersigned; or
- (iv) as a distribution to limited partners, members or shareholders or other equity holders of the undersigned; or
- (v) to the undersigned's affiliates or to any investment fund or other entity controlled or managed by the undersigned,

*provided*, that for any such transfers, (1) the Representatives shall receive a signed lock-up agreement for the balance of the Lock-Up Period from each donee, trustee, distributee, or transferee, as the case may be, (2) any such transfer shall not involve a disposition for value, (3) such transfers are not required to be reported with the SEC on Form 4 in accordance with Section 16 of the Securities Exchange Act of 1934, as amended, and (4) the undersigned shall not be required to and shall not otherwise voluntarily effect any public filing or report regarding such transfers.

Furthermore, during the Lock-Up Period, the undersigned may sell Company Equity Securities purchased by the undersigned on the open market following the Public Offering if and only if (i) such sales are not required during the Lock-Up Period to be reported in any press release or public report or filing with the SEC, or otherwise and (ii) the undersigned does not otherwise voluntarily effect any press release, public filing or report regarding such sales during the Lock-Up Period.

The undersigned also agrees and consents to the entry of stop transfer instructions with the Company's transfer agent and registrar against the transfer of the Lock-Up Securities except in compliance with the foregoing restrictions.

The undersigned hereby represents and warrants that the undersigned has full power and authority to enter into this lock-up agreement. This lock-up agreement is irrevocable and all authority herein conferred or agreed to be conferred shall survive the death or incapacity or dissolution of the undersigned and any obligations of the undersigned shall be binding on the undersigned and the successors, heirs, personal representatives and assigns of the undersigned. This lock-up agreement shall automatically terminate upon the earliest to occur, if any, of (1) the termination of the Underwriting Agreement before the sale of any Company Equity Securities to the Underwriters or (2) December 31, 2014, in the event that the Underwriting Agreement has not been executed by that

date; provided, however, that the Representatives and the Company may jointly agree, by written notice to you prior to such date, to extend such date for a
period of up to three additional months. This lock-up agreement shall be governed by, and construed in accordance with, the laws of the State of New York
without regard to the conflict of law principles thereof.

## [Signature page follows]

In witness whereof, the undersigned has executed this Lock-Up as of the date first set forth above.		
	Very truly yours,	
	Cignothyro	
	Signature:	
	Print Name:	
[Lock Up Agreement Signature Page]		

Exhibit C

# FORM OF PRESS RELEASE TO BE ISSUED PURSUANT TO SECTION 3(1)

[Date]

Forward Pharma A/S (the "Company") announced today that Leerink Partners LLC and Jefferies LLC, the joint book-running managers in the Company's recent public sale of  $[\cdot]$  ADSs, are [waiving] [releasing] a lock-up restriction with respect to of the Company's ordinary shares held by [certain officers or directors] [an officer or director] of the Company. The [waiver] [release] will take effect on , 20 , and the shares may be sold on or after such date.

This press release is not an offer for sale of the securities in the United States or in any other jurisdiction where such offer is prohibited, and such securities may not be offered or sold in the United States absent registration or an exemption from registration under the United States Securities Act of 1933, as amended.

# Articles of Association Forward Pharma A/S

The English part of this parallel document in Danish and English is an unofficial translation of the original Danish text. In the event of disputes or misunderstandings arising from the interpretation of the translation, the Danish language shall prevail.

	VEDTÆGTER FOR FORWARD PHARMA A/S CVR-NR. 28865880	ARTICLES OF ASSOCIATION OF FORWARD PHARMA A/S CBR-NO. 28865880
1	NAVN OG FORMÅL	NAME AND OBJECTS
1.1	Selskabets navn er Forward Pharma A/S.	The name of the company is Forward Pharma A/S.
1.2	Selskabets formål er direkte eller indirekte via datterselskaber at drive aktiviteter med udvikling, fremstilling, distribution og salg af lægemidler, og enhver anden relateret virksomhed efter bestyrelsens skøn. Herudover kan selskabet deltage i samarbejder eller indgå i partnerskaber med andre virksomheder inden for sit forretningsområde, herunder udlicensiere rettigheder inden for sit forretningsområde.	The object of the company is, directly or indirectly through subsidiaries, to conduct business within development, manufacturing, distribution and sale of drugs and medicaments, as well as any other related activities at the discretion of the board of directors. Furthermore, the company may, within its line of business, participate in partnerships or co-operate with other businesses, including by licensing out rights within its line of business.
2	AKTIEKAPITAL OG AKTIER	SHARE CAPITAL AND SHARES
2.1	Selskabets aktiekapital udgør nominelt kr. 44.837.570, fordelt i aktier à nominelt kr. 0,10 eller multipla heraf.	The company's nominal share capital is DKK 44,837,570, divided into shares of DKK 0.10 each or multiples thereof.
2.2	Aktiekapitalen er fuldt indbetalt.	The share capital has been fully paid up.
2.3	Aktierne skal lyde på navn og skal noteres på navn i selskabets ejerbog.	The shares shall be issued in the name of the holder and shall be recorded in the name of the holder in the company's register of shareholders.
2.4	Ejerbogen føres af Computershare	The register of shareholders is kept
2.5	A/S (CVR-nr. 27088899).	by Computershare A/S (Company Registration (CVR) no. 27088899).
2.5	Aktierne er ikke-omsætningspapirer. Der gælder ingen indskrænkninger i aktiernes omsættelighed.	The shares are non-negotiable instruments. No restrictions shall apply to the transferability of the shares.
2.6	Ingen aktier har særlige rettigheder.	No shares shall carry special rights.
2.7	Ingen aktionær skal være forpligtet til at lade sine aktier indløse helt eller delvist af selskabet eller andre.	No shareholder shall be under an obligation to have his shares redeemed in whole or in part by the company or by any third party.
2.8	Der udstedes ikke ejerbeviser for aktier i selskabet.	No share certificates are issued for the shares in the company.
3	UDSTEDELSE AF WARRANTS OG FORHØJELSE AF AKTIEKAPITALEN	ISSUE OF WARRANTS AND INCREASE OF THE SHARE CAPITAL
	Warrants til medarbejdere m.v.	Warrants to employees etc.
3.1	Selskabet har frem til 30. juni 2014 udstedt warrants til selskabets medarbejdere og konsulenter og medarbejdere og konsulenter i dets datterselskab, Forward Pharma GmbH, i et sådant omfang og på sådanne vilkår, som fremgår af <u>bilag 1</u> , der udgør en integreret del af disse vedtægter.	The company has up until 30 June 2014 issued warrants to the company's employees and consultants and employees and consultants of its subsidiary, Forward Pharma GmbH, to the extent and on such terms and conditions as set forth in <a href="mailto:appendix1">appendix 1</a> which forms an integral part of these articles of association.
3.2	Bestyrelsen er i perioden indtil 1. juni 2019 bemyndiget til, ad én eller flere gange, uden fortegningsret for selskabets eksisterende aktionærer,	"In the period until 1 June 2019, the board of directors is authorized, in one or more rounds, without pre-emption rights for the company's
	2	

	direktionsmedlemmer, bestyrelsesmedlemmer og konsulenter og/eller medarbejdere, direktionsmedlemmer, bestyrelsesmedlemmer og konsulenter i dets datterselskaber, idet bestyrelsen samtidig bemyndiges til at foretage de dertilhørende kapitalforhøjelser med op til nominelt DKK 214.000 aktier. De nye aktier, som kan tegnes ved udnyttelse af warrants, udstedes til en tegningskurs, der fastsættes af bestyrelsen og som kan være lavere end markedskursen på tidspunktet for udstedelsen af de pågældende warrants. Øvrige vilkår for warrants fastsættes af bestyrelsen i forbindelse med bestyrelsens udnyttelse af bemyndigelsen.	DKK 0.10, to the company's employees, members of the management, members of the board of directors, and consultants and/or employees, members of the management, members of the board of directors and consultants of its subsidiaries. The board of directors is further authorized to implement the capital increases required for this purpose by up to nominally DKK 214,000 shares. The subscription rate for the new shares that may be subscribed for by exercise of the warrants in question shall be fixed by the board of directors and may be lower than the market price at the time of issue of warrants. Other terms and conditions for the warrants, which can be issued by the board of directors according to the authorization, shall be fixed by the board of directors.
3.3	For aktier udstedt på baggrund af bemyndigelsen i punkt 3.2 skal i øvrigt gælde:	For shares issued pursuant to the authorization in article 3.2 the following shall apply:
	at der ikke kan ske delvis indbetaling,	that no partial payment may take place;
	<u>at</u> tegningen af aktier foretages uden fortegningsret for de eksisterende aktionærer,	that the subscription shall be effected without pre-emption rights of the existing shareholders;
	at aktierne skal tegnes ved kontant	that the shares shall be subscribed
	3	
	indbetaling,	for against payment of cash;
	at aktierne skal være ikke-omsætningspapirer,	that the shares shall be non-negotiable instruments
	<u>at</u> aktierne skal lyde på navn og noteres i selskabets ejerbog, og	that the shares shall be made out in the name of the holder and registered in the name of the holder in the company's register of shareholders; and
	<u>at</u> aktierne i øvrigt i enhver henseende har samme rettigheder som de eksisterende aktier.	that the shares in every respect shall carry the same rights as the existing shares.
	Bestyrelsen kan foretage de ændringer i selskabets vedtægter, der måtte være en følge af kapitalforhøjelsen.	The board of directors is entitled to make such changes amendments to the articles of association as may be required as a result of the capital increase.
	Aktier til medarbejdere m.v.	Shares to employees etc.
3.4	Bestyrelsen er i perioden indtil 1. juni 2019 bemyndiget til uden fortegningsret for selskabets eksisterende aktionærer at forhøje Selskabets aktiekapital, ad en eller flere omgange, med op til nominelt DKK 214.000 aktier ved udstedelse af aktier til dets	In the period until 1 June 2019, the board of directors is authorized to increase the share capital of the Company, in one or more rounds and without pre-emptive subscription rights for the existing shareholders, by up to nominally DKK 214,000 shares by issuance
	medarbejdere, direktionsmedlemmer, bestyrelsesmedlemmer og konsulenter og/eller medarbejdere, direktionsmedlemmer, bestyrelsesmedlemmer og konsulenter i dets datterselskaber. De nye aktier udstedes til en kurs, der fastsættes af bestyrelsen og som kan	of shares to the company's employees, members of the management, members of the board of directors, and consultants and/or employees, members of the management, members of the board of directors and consultants of its subsidiaries. The new shares are issued at a price determined by the
	konsulenter og/eller medarbejdere, direktionsmedlemmer, bestyrelsesmedlemmer og konsulenter i dets datterselskaber. De nye aktier udstedes til en kurs, der fastsættes af bestyrelsen og som	management, members of the board of directors, and consultants and/or employees, members of the management, members of the board of directors and consultants of its subsidiaries. The new
	konsulenter og/eller medarbejdere, direktionsmedlemmer, bestyrelsesmedlemmer og konsulenter i dets datterselskaber. De nye aktier udstedes til en kurs, der fastsættes af bestyrelsen og som kan	management, members of the board of directors, and consultants and/or employees, members of the management, members of the board of directors and consultants of its subsidiaries. The new
	konsulenter og/eller medarbejdere, direktionsmedlemmer, bestyrelsesmedlemmer og konsulenter i dets datterselskaber. De nye aktier udstedes til en kurs, der fastsættes af bestyrelsen og som kan	management, members of the board of directors, and consultants and/or employees, members of the management, members of the board of directors and consultants of its subsidiaries. The new
3.5	konsulenter og/eller medarbejdere, direktionsmedlemmer, bestyrelsesmedlemmer og konsulenter i dets datterselskaber. De nye aktier udstedes til en kurs, der fastsættes af bestyrelsen og som kan  4  være lavere end markedskursen. Øvrige vilkår for en sådan udstedelse af aktier fastsættes af bestyrelsen i forbindelse med	management, members of the board of directors, and consultants and/or employees, members of the management, members of the board of directors and consultants of its subsidiaries. The new shares are issued at a price determined by the  board of directors, which may be lower than the market price.  Other terms and conditions for such issue of shares, which can be issued by the board of directors according to the authorization, shall
3.5	konsulenter og/eller medarbejdere, direktionsmedlemmer, bestyrelsesmedlemmer og konsulenter i dets datterselskaber. De nye aktier udstedes til en kurs, der fastsættes af bestyrelsen og som kan  4  være lavere end markedskursen. Øvrige vilkår for en sådan udstedelse af aktier fastsættes af bestyrelsen i forbindelse med bestyrelsens udnyttelse af bemyndigelsen.  For aktier udstedt på baggrund af bemyndigelsen i punkt 3.4 skal i	management, members of the board of directors, and consultants and/or employees, members of the management, members of the board of directors and consultants of its subsidiaries. The new shares are issued at a price determined by the  board of directors, which may be lower than the market price. Other terms and conditions for such issue of shares, which can be issued by the board of directors according to the authorization, shall be fixed by the board of directors.  For shares issued pursuant to the authorization in article 3.4 the
3.5	konsulenter og/eller medarbejdere, direktionsmedlemmer, bestyrelsesmedlemmer og konsulenter i dets datterselskaber. De nye aktier udstedes til en kurs, der fastsættes af bestyrelsen og som kan  4  være lavere end markedskursen. Øvrige vilkår for en sådan udstedelse af aktier fastsættes af bestyrelsen i forbindelse med bestyrelsens udnyttelse af bemyndigelsen.  For aktier udstedt på baggrund af bemyndigelsen i punkt 3.4 skal i øvrigt gælde:	management, members of the board of directors, and consultants and/or employees, members of the management, members of the board of directors and consultants of its subsidiaries. The new shares are issued at a price determined by the  board of directors, which may be lower than the market price.  Other terms and conditions for such issue of shares, which can be issued by the board of directors according to the authorization, shall be fixed by the board of directors.  For shares issued pursuant to the authorization in article 3.4 the following shall apply:
3.5	konsulenter og/eller medarbejdere, direktionsmedlemmer, bestyrelsesmedlemmer og konsulenter i dets datterselskaber. De nye aktier udstedes til en kurs, der fastsættes af bestyrelsen og som kan  4  være lavere end markedskursen. Øvrige vilkår for en sådan udstedelse af aktier fastsættes af bestyrelsen i forbindelse med bestyrelsens udnyttelse af bemyndigelsen.  For aktier udstedt på baggrund af bemyndigelsen i punkt 3.4 skal i øvrigt gælde:  at der ikke kan ske delvis indbetaling,  at tegningen af aktier foretages uden fortegningsret for de	management, members of the board of directors, and consultants and/or employees, members of the management, members of the board of directors and consultants of its subsidiaries. The new shares are issued at a price determined by the  board of directors, which may be lower than the market price.  Other terms and conditions for such issue of shares, which can be issued by the board of directors according to the authorization, shall be fixed by the board of directors.  For shares issued pursuant to the authorization in article 3.4 the following shall apply:  that no partial payment may take place;  that the subscription shall be effected without pre-emption rights of
3.5	konsulenter og/eller medarbejdere, direktionsmedlemmer, bestyrelsesmedlemmer og konsulenter i dets datterselskaber. De nye aktier udstedes til en kurs, der fastsættes af bestyrelsen og som kan  4  være lavere end markedskursen. Øvrige vilkår for en sådan udstedelse af aktier fastsættes af bestyrelsen i forbindelse med bestyrelsens udnyttelse af bemyndigelsen.  For aktier udstedt på baggrund af bemyndigelsen i punkt 3.4 skal i øvrigt gælde:  at der ikke kan ske delvis indbetaling,  at tegningen af aktier foretages uden fortegningsret for de eksisterende aktionærer,	management, members of the board of directors, and consultants and/or employees, members of the management, members of the board of directors and consultants of its subsidiaries. The new shares are issued at a price determined by the  board of directors, which may be lower than the market price.  Other terms and conditions for such issue of shares, which can be issued by the board of directors according to the authorization, shall be fixed by the board of directors.  For shares issued pursuant to the authorization in article 3.4 the following shall apply:  that no partial payment may take place;  that the subscription shall be effected without pre-emption rights of the existing shareholders;
3.5	konsulenter og/eller medarbejdere, direktionsmedlemmer, bestyrelsesmedlemmer og konsulenter i dets datterselskaber. De nye aktier udstedes til en kurs, der fastsættes af bestyrelsen og som kan  4  være lavere end markedskursen. Øvrige vilkår for en sådan udstedelse af aktier fastsættes af bestyrelsen i forbindelse med bestyrelsens udnyttelse af bemyndigelsen.  For aktier udstedt på baggrund af bemyndigelsen i punkt 3.4 skal i øvrigt gælde:  at der ikke kan ske delvis indbetaling,  at tegningen af aktier foretages uden fortegningsret for de eksisterende aktionærer,  at aktierne skal tegnes ved kontant indbetaling,	management, members of the board of directors, and consultants and/or employees, members of the management, members of the board of directors and consultants of its subsidiaries. The new shares are issued at a price determined by the  board of directors, which may be lower than the market price.  Other terms and conditions for such issue of shares, which can be issued by the board of directors according to the authorization, shall be fixed by the board of directors.  For shares issued pursuant to the authorization in article 3.4 the following shall apply:  that no partial payment may take place;  that the subscription shall be effected without pre-emption rights of the existing shareholders;  that the shares shall be subscribed for against payment of cash;

		registered in the name of the holder in the company's register of shareholders; and
	<u>at</u> aktierne i øvrigt i enhver henseende har samme rettigheder som de eksisterende aktier.	that the shares in every respect shall carry the same rights as the existing shares.
	Bestyrelsen kan foretage de ændringer i selskabets vedtægter, der måtte være en følge af kapitalforhøjelsen.	The board of directors is entitled to make such changes amendments to the articles of association as may be required as a result of the capital
	5	
		increase.
	Øvrige kapitalforhøjelser	Other capital increases
3.6	Bestyrelsen er indtil 1. oktober 2019 bemyndiget til at beslutte at forhøje Selskabets aktiekapital, ad én eller flere gange, med et nominelt beløb på i alt op til DKK 3.500.000 ved udstedelse af aktier til en kurs fastsat af bestyrelsen, der kan være lavere end markedskursen.	The board of directors is authorised in the period until 1 october 2019 to resolve to increase the Company's share capital in one or more issues by up to a total nominal amount of DKK 3,500,000 at a price determined by the board of directors, which may be lower than the market price.
3.7	For aktier udstedt på baggrund af bemyndigelsen i punkt 3.6 skal i øvrigt gælde:	For shares issued pursuant to the authorization in article 3.6 the following shall apply:
	at der ikke kan ske delvis indbetaling,	that no partial payment may take place;
	<u>at</u> tegningen af aktier foretages uden fortegningsret for de eksisterende aktionærer,	<u>that</u> the subscription shall be effected without pre-emption rights of the existing shareholders;
	at aktierne skal tegnes ved kontant indbetaling, indbetaling i andre værdier end kontanter eller gældskonvertering,	that the shares shall be subscribed for against payment of cash, contribution in kind or conversion of debt;
	at aktierne skal være ikke-omsætningspapirer, og	that the shares shall be non-negotiable instruments; and
	<u>at</u> aktierne skal lyde på navn og noteres i selskabets ejerbog.	that the shares shall be made out in the name of the holder and registered in the name of the holder in the company's register of shareholders.
	Bestyrelsen kan foretage de	The board of directors is entitled to
	6	
	ændringer i selskabets vedtægter, der måtte være en følge af kapitalforhøjelsen.	make such changes amendments to the articles of association as may be required as a result of the capital increase.
3.8	kapitalforhøjelsen.	may be required as a result of the capital increase.
3.8	kapitalforhøjelsen.  IPO aktier	may be required as a result of the capital increase.  IPO shares
	kapitalforhøjelsen.  IPO aktier  [Slettet]	may be required as a result of the capital increase.  IPO shares  [Deleted]
	kapitalforhøjelsen.  IPO aktier  [Slettet]  [Slettet]	may be required as a result of the capital increase.  IPO shares  [Deleted]
3.9	kapitalforhøjelsen.  IPO aktier  [Slettet]  [Slettet]  Overallokeringsaktier  Bestyrelsen er indtil 31. december 2014 bemyndiget til at beslutte at forhøje selskabets aktiekapital, ad én eller flere gange, med et nominelt beløb på i alt op til DKK 225.000 ved udstedelse af aktier til en kurs fastsat af bestyrelsen, der kan være lavere end	may be required as a result of the capital increase.  IPO shares  [Deleted]  [Deleted]  Over-Allotment Shares  The board of directors is authorised in the period until 31 December 2014 to resolve to increase the company's share capital in one or more issues by up to a total nominal amount of DKK 225,000 at a price determined by the board of directors, which may
3.9	kapitalforhøjelsen.  IPO aktier  [Slettet]  [Slettet]  Overallokeringsaktier  Bestyrelsen er indtil 31. december 2014 bemyndiget til at beslutte at forhøje selskabets aktiekapital, ad én eller flere gange, med et nominelt beløb på i alt op til DKK 225.000 ved udstedelse af aktier til en kurs fastsat af bestyrelsen, der kan være lavere end markedskursen, ved kontant indbetaling.  For aktier udstedt på baggrund af bemyndigelsen i punkt 3.10 skal i	may be required as a result of the capital increase.  IPO shares  [Deleted]  [Deleted]  Over-Allotment Shares  The board of directors is authorised in the period until 31  December 2014 to resolve to increase the company's share capital in one or more issues by up to a total nominal amount of DKK 225,000 at a price determined by the board of directors, which may be lower than the market price, against payment of cash.  For shares issued pursuant to the authorization in article 3.10 the
3.9	kapitalforhøjelsen.  IPO aktier  [Slettet]  [Slettet]  Overallokeringsaktier  Bestyrelsen er indtil 31. december 2014 bemyndiget til at beslutte at forhøje selskabets aktiekapital, ad én eller flere gange, med et nominelt beløb på i alt op til DKK 225.000 ved udstedelse af aktier til en kurs fastsat af bestyrelsen, der kan være lavere end markedskursen, ved kontant indbetaling.  For aktier udstedt på baggrund af bemyndigelsen i punkt 3.10 skal i øvrigt gælde:	may be required as a result of the capital increase.  IPO shares  [Deleted]  Over-Allotment Shares  The board of directors is authorised in the period until 31 December 2014 to resolve to increase the company's share capital in one or more issues by up to a total nominal amount of DKK 225,000 at a price determined by the board of directors, which may be lower than the market price, against payment of cash.  For shares issued pursuant to the authorization in article 3.10 the following shall apply:
3.9	kapitalforhøjelsen.  IPO aktier  [Slettet]  Overallokeringsaktier  Bestyrelsen er indtil 31. december 2014 bemyndiget til at beslutte at forhøje selskabets aktiekapital, ad én eller flere gange, med et nominelt beløb på i alt op til DKK 225.000 ved udstedelse af aktier til en kurs fastsat af bestyrelsen, der kan være lavere end markedskursen, ved kontant indbetaling.  For aktier udstedt på baggrund af bemyndigelsen i punkt 3.10 skal i øvrigt gælde:  at der ikke kan ske delvis indbetaling,  at tegningen af aktier foretages uden fortegningsret for de	may be required as a result of the capital increase.  IPO shares  [Deleted]  Over-Allotment Shares  The board of directors is authorised in the period until 31 December 2014 to resolve to increase the company's share capital in one or more issues by up to a total nominal amount of DKK 225,000 at a price determined by the board of directors, which may be lower than the market price, against payment of cash.  For shares issued pursuant to the authorization in article 3.10 the following shall apply:  that no partial payment may take place;  that the subscription shall be effected without pre-emption rights of

	omsætningspapirer,	negotiable instruments;
	<u>at</u> aktierne skal lyde på navn og noteres i selskabets ejerbog, og	that the shares shall be made out in the name of the holder and registered in the name of the holder in the company's register of shareholders; and
	<u>at</u> aktierne i øvrigt i enhver henseende har samme rettigheder som de eksisterende aktier.	that the shares in every respect shall carry the same rights as the existing shares.
	Bestyrelsen kan foretage de ændringer i selskabets vedtægter, der måtte være en følge af kapitalforhøjelsen. Derudover er bestyrelsen bemyndiget til at foranledige sådanne aktiers deponering hos en depotbank og samtidig udstedelse af American Depository Shares.	The board of directors is entitled to make such changes amendments to the articles of association as may be required as a result of the capital increase. Further, the board of directors is authorized to cause such shares to be deposited with a depository bank and simultaneous issuance of American Depository Shares.
4	BEMYNDIGELSE TIL AT UDLODDE EKSTRAORDINÆRT UDBYTTE OG KØBE EGNE AKTIER	AUTHORIZATION TO DISTRIBUTE EXTRAORDINARY DIVIDENDS AND ACQUIRE OWN SHARES
4.1	Bestyrelsen er af generalforsamlingen bemyndiget til at træffe beslutning om uddeling af ekstraordinært udbytte, såfremt Selskabets økonomiske situation giver grundlag for dette.	The board of directors is authorized to resolve to distribute extraordinary dividends if the company's financial situation warrants such distribution.
4,2	Bestyrelsen er i perioden indtil 1. oktober 2019 bemyndiget til at lade Selskabet erhverve egne aktier i et omfang således, at den pålydende værdi af Selskabets samlede	In the period until 1 October 2019, the board of directors is authorized to have the company acquire own shares to such extent that the nominal value of the company's aggregate holding of
	8	
	beholdning af egne aktier ikke på noget tidspunkt overstiger 10 procent af aktiekapitalen. Vederlaget for de pågældende aktier må ikke afvige mere end 20 procent fra følgende kurs: Den ved erhvervelsen noterede kurs for de på NASDAQ Global Select Market, New York, under fondskode US34986J1051 handlede American Depositary Shares relateret til selskabets aktier divideret med 1 (svarende til antallet af underlæggende aktier i selskabet per American Depositary Share). Autorisationen kan benyttes til at (i) erhverve egne aktier direkte, og/eller (ii) erhverve American Depositary Shares som derefter kan overleveres til depotbanken mod levering af de underliggende aktier repræsenteret af American Depositary Shares.	own shares at no time may exceed 10 percent of the share capital. The price payable for the shares in question may not deviate by more than 20 percent from the following price: The prevailing quoted price at the time of the acquisition applicable to the American Depositary Shares related to the company's shares traded under ISIN code US34986J1051 at NASDAQ Global Select Market, New York, divided by 1 (equaling the number of underlying shares in the company per American Depositary Share). The authorization can be utilized to (i) acquire own shares directly, and/or (ii) acquire American Depositary Shares which can then be surrendered to the depository bank enabling the company to take delivery of the underlying shares represented by such American Depository Shares.
5	GENERALFORSAMLINGEN, AFHOLDELSESSTED OG INDKALDELSE	GENERAL MEETING, VENUE AND NOTICE
5.1	Generalforsamlingen er inden for de ved lovgivningen og vedtægterne fastsatte grænser den højeste myndighed i selskabet.	The general meeting has the supreme authority in all matters relating to the company subject to law and these articles of association.
5.2	Selskabets generalforsamlinger afholdes i Region Hovedstaden, Danmark.	The general meetings of the company shall be held in the Capital Region of Denmark.
5.3	Selskabets ordinære generalforsamling afholdes i så god	The annual general meeting of the company shall be held well in advance
	9	
	tid, at den reviderede og godkendte årsrapport kan indsendes til Erhvervsstyrelsen, så den er modtaget i styrelsen inden 5 måneder efter udløbet af hvert regnskabsår.	in order for the revised and adopted annual report to be sent to and received by the Danish Business Authority within 5 months after the expiry of each financial year.
4.4	Ekstraordinær generalforsamling afholdes, når bestyrelsen eller revisor forlanger det. Ekstraordinær generalforsamling skal endvidere afholdes, når det forlanges af aktionærer, der tilsammen ejer mindst fem procent af aktiekapitalen. Sådan begæring skal ske skriftligt til bestyrelsen og være ledsaget af et bestemt angivet forslag til dagsordenspunkt. Bestyrelsen indkalder til en	Extraordinary general meetings shall be held when determined by the board of directors or requested by the company's auditor. Furthermore, an extraordinary general meeting shall be held when requested by shareholders possessing no less than five per cent of the share capital. Such request shall be submitted in writing to the board of directors and be accompanied by a specific proposal for

	ekstraordinær generalforsamling senest to uger efter, at det er forlangt.	the business to be transacted. The board of directors convenes an extraordinary general meeting no later than two weeks after such request has been made.
5.5	Generalforsamlinger indkaldes af bestyrelsen med mindst to ugers og højst fire ugers varsel. Indkaldelsen offentliggøres på selskabets hjemmeside og i øvrigt på den måde og i den form, som de børser, på hvilke selskabets aktier er noteret, til enhver tid måtte forlange. Indkaldelse sendes endvidere til alle i ejerbogen noterede aktionærer, som har fremsat begæring herom.	General meetings shall be convened by the board of directors with at least two weeks' and not more than four weeks' notice. The notice shall be published on the company's website and moreover in such way and in such form as required from time to time by the stock exchanges on which the company's shares are listed. Furthermore, a notice of the general meeting shall be sent to all shareholders recorded in the company's register of shareholders who have so requested.
5.6	I indkaldelsen skal angives tid og sted for generalforsamlingen samt	The notice shall specify the time and place of the general meeting and the
	10	
	dagsorden, hvoraf det fremgår, hvilke anliggender der skal behandles på generalforsamlingen. Såfremt forslag til vedtægtsændringer skal behandles på generalforsamlingen, skal forslagets væsentligste indhold angives i indkaldelsen. Indkaldelse til generalforsamlingen, hvor der skal træffes beslutning efter selskabslovens § 77, stk. 2, § 92, stk. 1 eller 5, eller § 107, stk. 1 eller 2, skal indeholde den fulde ordlyd af forslaget.	agenda containing the business to be transacted at the general meeting. If a proposal to amend the articles of association is to be considered at the general meeting, the main contents of the proposal must be specified in the notice. Notices convening general meetings at which a resolution shall be passed pursuant to Section 77(2), Section 92(1) or (5), or Section 107(1) or (2) of the Danish Companies Act must set out the full wording of the proposals.
5.7	I en periode på to uger før en generalforsamling, inklusive datoen for generalforsamlingens afholdelse, gøres følgende oplysninger tilgængelige på selskabets hjemmeside:	For a period of two weeks prior to the general meeting, including the date of the general meeting, the following information shall be available on the company's website:
	(a) Indkaldelsen	(a) The notice convening the general meeting;
	(b) Det samlede antal aktier og stemmerettigheder på datoen for indkaldelsen	(b) The total number of shares and voting rights on the date of the notice;
	(c) De dokumenter, der skal fremlægges på generalforsamlingen	(c) The documents to be presented at the general meeting;
	(d) Dagsordenen og de fuldstændige forslag samt for den ordinære generalforsamlings vedkommende tillige revideret årsrapport	(d) The agenda and the complete proposals as well as, for annual general meetings, the audited annual report;
	(e) De formularer, der skal anvendes ved stemmeafgivelse pr. fuldmagt eller skriftligt ved brevstemme.	(e) The forms to be used for voting by proxy or voting by correspondence.
	11	
6	DAGSORDEN FOR DEN ORDINÆRE	AGENDA FOR THE ANNUAL GENERAL MEETING,
	GENERALFORSAMLING, DIRIGENT og PROTOKOL	CHAIRMAN AND PROTOCOL
6.1	Enhver aktionær har ret til at få et bestemt emne behandlet på den ordinære generalforsamling. Begæring herom skal fremsættes skriftligt over for bestyrelsen senest seks uger før generalforsamlingens afholdelse.	Every shareholder shall be entitled to have a specific subject considered at the annual general meeting. Such proposals must be submitted in writing to the board of directors not later than six weeks prior to the general meeting.
6.2	Dagsordenen for den ordinære generalforsamling skal omfatte følgende:	The agenda for the annual general meeting shall include the following:
	(a) Bestyrelsens beretning om selskabets virksomhed i det forløbne regnskabsår	(a) The board of directors' report on the company's activities in the past financial year;
	(b) Fremlæggelse og godkendelse af revideret årsrapport	(b) Presentation and adoption of the audited annual report;
	(c) Anvendelse af overskud eller dækning af underskud i henhold til den godkendte årsrapport	(c) Distribution of profit or covering of loss according to the adopted annual report;
	(d) Meddelelse af decharge til bestyrelsen og direktionen	(d) Discharge of the board of directors and the management board;

	(e) Valg af medlemmer til bestyrelsen	(e) Election of members to the board of directors;
	(f) Valg af revisor	(f) Appointment of auditor;
	(g) Eventuelle forslag fra bestyrelse og aktionærer	(g) Any proposals from the board of directors or shareholders;
	(h) Eventuelt	(h) Any other business.
6.3	Generalforsamlingen ledes af en af bestyrelsen valgt dirigent, der afgør alle spørgsmål vedrørende behandling af dagsordenspunkterne, stemmeafgivning og resultaterne	The general meeting shall be presided over by a chairman elected by the board of directors. The chairman shall decide all questions regarding the business transacted, the casting of
	12	
	heraf.	votes and the results of voting.
6.4	Der føres en protokol over generalforsamlingen, der underskrives	Minutes of the proceedings of the general meeting shall be entered
0.4	af dirigenten.	into a minute book to be signed by the chairman.
7	AKTIONÆRERNES MØDE- OG STEMMERET PÅ GENERALFORSAMLINGEN	SHAREHOLDERS' ATTENDANCE AND VOTING RIGHTS AT THE GENERAL MEETING
7.1	En aktionærs ret til at deltage i en generalforsamling og til at afgive stemme fastsættes i forhold til de aktier, aktionæren besidder på registreringsdatoen. Registreringsdatoen ligger en uge før generalforsamlingen. De aktier, den enkelte aktionær besidder, opgøres på registreringsdatoen på baggrund af notering af aktionærens ejerforhold i ejerbogen samt eventuelle meddelelser om ejerforhold, som selskabet har modtaget med henblik på indførsel i ejerbogen, men som endnu ikke er indført i ejerbogen.	The right of a shareholder to attend and vote at a general meeting is determined by the shares held by the shareholder at the record date. The record date is one week prior to the general meeting. The shares held by each shareholder at the record date is calculated based on the registration of the number of shares held by that shareholder in the company's register of shareholders as well as on any notification of ownership received by the company for the purpose of registration in the Company's register of shareholders, but which have not yet been registered.
7.2	En aktionær, der er berettiget til at deltage i generalforsamlingen i henhold til punkt 6.1, og som ønsker at deltage i generalforsamlingen, skal senest tre dage før dens afholdelse anmode om adgangskort.	A shareholder who is entitled to attend the general meeting pursuant to article 6.1 and who wants to attend the general meeting shall request to receive an admission card no later than three days prior to the date of the general meeting.
7.3	En aktionær kan møde personligt eller ved fuldmægtig, og både aktionæren	A shareholder may attend in person or by proxy, and the shareholder or the
	13	
	og fuldmægtigen kan møde med en rådgiver.	proxy may attend together with an adviser.
7.4	Stemmeret kan udøves i henhold til skriftlig og dateret fuldmagt i overensstemmelse med den til enhver tid gældende lovgivning herom.	The right to vote may be exercised by a written and dated proxy in accordance with applicable laws.
7.5	En aktionær, der er berettiget til at deltage i en generalforsamling i henhold til punkt 6.1, kan endvidere stemme skriftligt ved brevstemme i overensstemmelse med selskabslovens regler herom. Brevstemmer skal være selskabet i hænde senest dagen før generalforsamlingen. Brevstemmer kan ikke tilbagekaldes.	A shareholder who is entitled to participate in the general meeting pursuant to article 6.1 may vote by correspondence in accordance with the provisions of the Danish Companies Act. Such votes by correspondence shall be received by the Company not later than the day before the general meeting. Votes by correspondence cannot be withdrawn.
7.6	Hvert aktiebeløb på nominelt kr. 0,10 giver én stemme.	Each share of the nominal value of DKK 0.10 shall carry one vote.
7.7	Enhver aktionær er berettiget til at afgive forskellige stemmer på sine aktier. Kravet i selskabslovens § 104, stk. 1, hvorefter en kapitalejer skal stemme samlet på sine kapitalandele, er således fraveget ved denne bestemmelse.	Any shareholder is entitled to cast different votes on his shares. Accordingly, the requirement set out in Section 104 (1) of the Danish Companies Act according to which a shareholder must vote on his shares in aggregate, is deviated from by virtue of this provision.
8	BESLUTNINGER PÅ GENERALFORSAMLINGEN	RESOLUTIONS AT GENERAL MEETINGS
8.1	De på generalforsamlingen behandlede anliggender afgøres ved simpelt stemmeflertal blandt afgivne stemmer, medmindre andet følger af	Resolutions by the general meeting shall be passed by a simple majority of votes cast unless otherwise prescribed by law or by these articles of

	lovgivningen eller disse vedtægter.	association.
8.2	Til vedtagelse af beslutning om vedtægtsændringer, selskabets opløsning, fusion eller spaltning kræves, at beslutningen vedtages med mindst 2/3 af såvel de afgivne stemmer som af den på generalforsamlingen repræsenterede aktiekapital, medmindre der i medfør af lovgivningen stilles strengere eller lempeligere vedtagelseskrav eller tillægges bestyrelsen eller andre organer selvstændig kompetence.	Adoption of changes to these articles of association, dissolution of the company, merger or demerger requires that the decision is adopted with at least 2/3 of the votes cast as well as the share capital represented at the general meeting, unless applicable laws prescribe stricter or less strict adoption requirements or applicable laws confer independent competence to the board of directors or other bodies.
9	ELEKTRONISK KOMMUNIKATION	ELECTRONIC COMMUNICATION
9.1	Al kommunikation fra selskabet til de enkelte aktionærer, herunder indkaldelse til generalforsamlinger, kan ske elektronisk via offentliggørelse på selskabets hjemmeside eller ved udsendelse via e-mail. Generelle meddelelser gøres tilgængelige på selskabets hjemmeside og på sådan anden måde, som måtte være foreskrevet i henhold til lov. Selskabet kan til enhver tid vælge i stedet at fremsende meddelelser mv. med almindelig post.	All communication from the company to the individual shareholders, including notices convening general meetings, may take place electronically by posting on the company's website or by email. General notices shall be published on the company's website and in such other manner as may be prescribed by applicable laws. The company may at all times choose to send notices, etc., by ordinary post instead.
9.2	Kommunikation fra aktionærer til selskabet kan ske ved e-mail eller med almindelig post.	Communication from a shareholder to the company may take place by email or by ordinary post.
9.3	Selskabet anmoder de navnenoterede	The company shall request all share-
	15	
	aktionærer om en e-mail adresse, hvortil meddelelser mv. kan sendes. Det er den enkelte aktionærs ansvar at sikre, at selskabet til stadighed er i besiddelse af korrekte oplysninger om e-mail adresse. Selskabet har ingen pligt til at søge oplysningerne berigtiget eller til at fremsende meddelelser på anden måde.	holders registered by name to submit an email address to which notices, etc., may be sent. Each shareholder is responsible for ensuring that the company has the correct email address at all times. The company is not obliged to verify such contact information or to send notices in any other way.
9.4	Oplysninger om kravene til anvendte systemer samt om fremgangsmåden ved elektronisk kommunikation findes på selskabets hjemmeside, www.forward-pharma.com.	The company's website, www.forward-pharma.com, contains information about system requirements and electronic communication procedures.
10	BESTYRELSEN	BOARD OF DIRECTORS
10.1	Bestyrelsen varetager den overordnede ledelse af selskabet.	The board of directors shall be in charge of the overall management of the company.
10.2	Bestyrelsen består af mindst tre og højst seks medlemmer, der vælges af generalforsamlingen.	The board of directors consists of not less than three and not more than six members elected by the general meeting.
10.3	Bestyrelsen vælger en formand blandt sine medlemmer.	The board of directors elects a chairman among its members.
10.4	De af generalforsamlingen valgte bestyrelsesmedlemmer vælges for en periode på ét år. Genvalg af bestyrelsesmedlemmer kan finde sted. Til selskabets bestyrelse kan kun vælges personer, som er yngre end 70 år på valgtidspunktet.	The members of the board of directors elected by the general meeting are elected for a term of one year. Re-election of board members may take place. Only persons who are younger than 70 years at the time of election may be elected to the board of directors.
	16	
10.5	Bestyrelsen er beslutningsdygtig, når over halvdelen af bestyrelsesmedlemmerne, herunder formanden, er repræsenteret.	The board of directors forms a quorum when more than half of its members are represented, including the chairman.
10.6	De i bestyrelsen behandlede anliggender afgøres ved simpelt stemmeflertal. I tilfælde af stemmelighed er formandens stemme udslagsgivende.	Resolutions of the board of directors are passed by simple majority. In the event of equal votes, the chairman shall have a casting vote.
10.7	Bestyrelsen skal ved sin forretningsorden træffe nærmere bestemmelse om udførelsen af sit hverv.	The board of directors shall adopt rules of procedure containing detailed provisions for the performance of its duties.
10.8	Over det på bestyrelsesmøderne passerede føres en protokol, der underskrives af samtlige bestyrelsesmedlemmer.	Minutes of the proceedings of the board meetings shall be recorded in a minute book to be signed by all members of the board of directors.

11	DIREKTIONEN	EXECUTIVE MANAGEMENT
11.1	Bestyrelsen ansætter en direktion bestående af ét til tre medlemmer til at varetage den daglige ledelse af selskabet.	The board of directors appoints a management board consisting of one to three members to be in charge of the day-to-day management of the company.
12	TEGNINGSREGEL	RULES OF SIGNATURE
12.1	Selskabet tegnes (i) af bestyrelsens formand i forening med et bestyrelsesmedlem, (ii) af bestyrelsens formand i forening med et medlem af direktionen eller (iii) af den samlede bestyrelse.	The company shall be bound (i) by the joint signatures of the chairman and a member of the board of directors, (ii) by the joint signatures of the chairman and a member of the management board, or (iii) by the joint signatures
	17	
		of all members of the board of directors.
13	REVISION	AUDIT
13.1	Selskabets årsrapport revideres af en statsautoriseret revisor, der vælges af generalforsamlingen for ét år ad gangen. Genvalg kan finde sted.	The company's annual report shall be audited by a state-authorized public accountant elected by the general meeting for a one-year term. Re-election may take place.
14	REGNSKABSÅR	FINANCIAL YEAR
14.1	Selskabets regnskab er kalenderåret.	The company's financial year follows the calendar year.
15	BILAG	APPENDICES
15.1	Bilag 1: Warrants udstedt senest 30. juni 2014 samt vilkårene for disse.	Appendix 1: Warrants issued on or prior to 30 June 2014 the rules applicable to these.
	Således vedtaget den [∙].	As adopted on [·].

# FORWARD PHARMA A/S

# AND

# THE BANK OF NEW YORK MELLON

# As Depositary

# AND

# OWNERS AND HOLDERS OF AMERICAN DEPOSITARY SHARES

# Deposit Agreement

Dated as of , 2014

# TABLE OF CONTENTS

ARTICLE 1.	DEFINITIONS	1
SECTION 1.1	American Depositary Shares	1
SECTION 1.2	Commission	2
SECTION 1.3	Company	2
SECTION 1.4	Custodian	2
SECTION 1.5	Deliver; Surrender	2
SECTION 1.6	Deposit Agreement	3
SECTION 1.7	Depositary; Corporate Trust Office	3
SECTION 1.8	Deposited Securities	3
SECTION 1.9	Dollars	3
SECTION 1.10	DTC	3
SECTION 1.11	Foreign Registrar	3
SECTION 1.12	Holder	4
SECTION 1.13	Owner	4
SECTION 1.14	Receipts	4
SECTION 1.15	Registrar	4
SECTION 1.16	Restricted Securities	4
SECTION 1.17	Securities Act of 1933	5
SECTION 1.18	Shares	5
ARTICLE 2.	FORM OF RECEIPTS, DEPOSIT OF SHARES, DELIVERY, TRANSFER AND SURRENDER OF AMERICAN	
	DEPOSITARY SHARES	5
SECTION 2.1	Form of Receipts; Registration and Transferability of American Depositary Shares	5
SECTION 2.2	Deposit of Shares	6
SECTION 2.3	Delivery of American Depositary Shares	7
SECTION 2.4	Registration of Transfer of American Depositary Shares; Combination and Split-up of Receipts; Interchange of	,
5EC11011 2.4	Certificated and Uncertificated American Depositary Shares	8
SECTION 2.5	Surrender of American Depositary Shares and Withdrawal of Deposited Securities	9
SECTION 2.6	Limitations on Delivery, Transfer and Surrender of American Depositary Shares	10
SECTION 2.7	Lost Receipts, etc.	10
	Cancellation and Destruction of Surrendered Receipts	
SECTION 2.8		11
SECTION 2.9	Pre-Release of American Depositary Shares	11
SECTION 2.10	DTC Direct Registration System and Profile Modification System	12
ARTICLE 3.	CERTAIN OBLIGATIONS OF OWNERS AND HOLDERS OF AMERICAN DEPOSITARY SHARES	12
SECTION 3.1	Filing Proofs, Certificates and Other Information	12
SECTION 3.2	Liability of Owner for Taxes	13
SECTION 3.3	Warranties on Deposit of Shares	13
SECTION 3.4	Disclosure of Interests	13
5ECTION 5.4	Disclosure of interests	13
ARTICLE 4.	THE DEPOSITED SECURITIES	14
SECTION 4.1	Cash Distributions	14
SECTION 4.2	Distributions Other Than Cash, Shares or Rights	14
SECTION 4.3	Distributions in Shares	15
SECTION 4.4	Rights	16
SECTION 4.5	Conversion of Foreign Currency	17

SECTION 4.6	Fixing of Record Date	18
SECTION 4.7	Voting of Deposited Securities	19
SECTION 4.8	Changes Affecting Deposited Securities	20
SECTION 4.9	Reports	20
SECTION 4.10	Lists of Owners	20
SECTION 4.11	Withholding	21
ARTICLE 5.	THE DEPOSITARY, THE CUSTODIANS AND THE COMPANY	21
SECTION 5.1	Maintenance of Office and Transfer Books by the Depositary	21
SECTION 5.2		
SECTION 5.3	Obligations of the Depositary, the Custodian and the Company	22
SECTION 5.4	Resignation and Removal of the Depositary	23
SECTION 5.5	The Custodians	24
SECTION 5.6	Notices and Reports	25
SECTION 5.7	Distribution of Additional Shares, Rights, etc.	25
SECTION 5.8	Indemnification	26
SECTION 5.9	Charges of Depositary	28
SECTION 5.10	Retention of Depositary Documents	29
SECTION 5.11	Exclusivity	29
SECTION 5.12	List of Restricted Securities Owners	29
SECTION 5.13	Registration of Shares; Share Register	29
	2	
ARTICLE 6.	AMENDMENT AND TERMINATION	31
SECTION 6.1	AMENDMENT AND TERMINATION  Amendment	31
SECTION 6.1 SECTION 6.2	Termination	31
SECTION 0.2	remination	31

Owners and Holders as Parties; Binding Effect 33 33 Arbitration; Settlement of Disputes 34 Submission to Jurisdiction; Jury Trial Waiver 35 36 36

32 32

33

33

3

# **DEPOSIT AGREEMENT**

, 2014, among FORWARD PHARMA A/S, a company incorporated under the laws of DEPOSIT AGREEMENT dated as of Denmark (herein called the Company), THE BANK OF NEW YORK MELLON, a New York banking corporation (herein called the Depositary), and all Owners and Holders (each as hereinafter defined) from time to time of American Depositary Shares issued hereunder.

# WITNESSETH:

WHEREAS, the Company desires to provide, as hereinafter set forth in this Deposit Agreement, for the deposit of Shares (as hereinafter defined) of the Company from time to time with the Depositary or with the Custodian (as hereinafter defined) as agent of the Depositary for the purposes set forth in this Deposit Agreement, for the creation of American Depositary Shares representing the Shares so deposited and for the execution and delivery of American Depositary Receipts evidencing the American Depositary Shares; and

WHEREAS, the American Depositary Receipts are to be substantially in the form of Exhibit A annexed hereto, with appropriate insertions, modifications and omissions, as hereinafter provided in this Deposit Agreement;

NOW, THEREFORE, in consideration of the premises, it is agreed by and between the parties hereto as follows:

### ARTICLE 1. **DEFINITIONS**

ARTICLE 7.

SECTION 7.1

SECTION 7.2

SECTION 7.3

SECTION 7.4

**SECTION 7.5** 

SECTION 7.6

SECTION 7.7

**SECTION 7.8** 

SECTION 7.9

**MISCELLANEOUS** 

Severability

Notices

Counterparts; Signatures

Waiver of Immunities

Governing Law

No Third Party Beneficiaries

The following definitions shall for all purposes, unless otherwise clearly indicated, apply to the respective terms used in this Deposit Agreement:

SECTION 1.1 American Depositary Shares.

The term "American Depositary Shares" shall mean the securities created under this Deposit Agreement representing rights with respect to the Deposited Securities. American Depositary Shares may be certificated securities evidenced by Receipts or uncertificated securities. The form of Receipt annexed as Exhibit A to this Deposit Agreement shall be the prospectus required under the Securities Act of 1933 for sales of both certificated and uncertificated American Depositary Shares. Except for those provisions of this Deposit Agreement that refer specifically to Receipts, all the provisions of

this Deposit Agreement shall apply to both certificated and uncertificated American Depositary Shares. Each American Depositary Share shall represent the number of Shares specified in Exhibit A to this Deposit Agreement, until there shall occur a distribution upon Deposited Securities covered by Section 4.3 or a change in Deposited Securities covered by Section 4.8 with respect to which additional American

Depositary Shares are not delivered, and thereafter American Depositary Shares shall represent the amount of Shares or Deposited Securities specified in such Sections.

SECTION 1.2 Commission.

The term "Commission" shall mean the Securities and Exchange Commission of the United States or any successor governmental agency in the United States.

SECTION 1.3 Company.

The term "Company" shall mean Forward Pharma A/S, a company organized under the laws of Denmark, and its successors.

SECTION 1.4 Custodian.

The term "Custodian" shall mean the London Branch of The Bank of New York Mellon, as agent of the Depositary for the purposes of this Deposit Agreement, and any other firm or corporation which may hereafter be appointed by the Depositary pursuant to the terms of Section 5.5, as successor, substitute or additional custodian or custodians hereunder, as the context shall require and the term "Custodian" shall also mean all of them, collectively.

SECTION 1.5 Deliver; Surrender.

- (a) The term "deliver", or its noun form, when used with respect to Shares or other Deposited Securities, shall mean recordation of transfer of such Shares or other Deposited Securities in the share or other relevant register of the Company in the name of the person entitled to that delivery or, in the case of other Deposited Securities that are not in the form of securities of the Company, shall mean delivery of such Deposited Securities in such a way as is necessary under applicable law to effect transfers of such Deposited Securities to the person entitled to that delivery, including, without limitation, (i) book-entry transfer of those Shares or other Deposited Securities to an account maintained by an institution authorized under applicable law to effect transfers of such securities designated by the person entitled to that delivery, or (ii) physical transfer of certificates evidencing those Shares or other Deposited Securities registered in the name of, or duly endorsed or accompanied by proper instruments of transfer to, the person entitled to that delivery.
- (b) The term "deliver", or its noun form, when used with respect to American Depositary Shares, shall mean (i) book-entry transfer of American Depositary Shares to an account at DTC designated by the person entitled to such delivery, evidencing American Depositary Shares registered in the name requested by that person, (ii) registration of American Depositary Shares not evidenced by a Receipt on the books of the Depositary in the name requested by the person entitled to such delivery and

2

mailing to that person of a statement confirming such registration or (iii) if requested by the person entitled to such delivery, delivery at the Corporate Trust Office of the Depositary to the person entitled to such delivery of one or more Receipts.

(c) The term "surrender", when used with respect to American Depositary Shares, shall mean (i) one or more book-entry transfers of American Depositary Shares to the DTC account of the Depositary, (ii) delivery to the Depositary at its Corporate Trust Office of an instruction to surrender American Depositary Shares not evidenced by a Receipt or (iii) surrender to the Depositary at its Corporate Trust Office of one or more Receipts evidencing American Depositary Shares.

SECTION 1.6 Deposit Agreement.

The term "Deposit Agreement" shall mean this Deposit Agreement, as the same may be amended from time to time in accordance with the provisions hereof.

SECTION 1.7 Depositary; Corporate Trust Office.

The term "Depositary" shall mean The Bank of New York Mellon, a New York banking corporation, and any successor as depositary hereunder. The term "Corporate Trust Office", when used with respect to the Depositary, shall mean the office of the Depositary which at the date of this Deposit Agreement is 101 Barclay Street, New York, New York 10286.

SECTION 1.8 Deposited Securities.

The term "Deposited Securities" as of any time shall mean Shares at such time deposited or deemed to be deposited under this Deposit Agreement, including without limitation Shares that have not been successfully delivered upon surrender of American Depositary Shares, and any and all other securities, property and cash received by the Depositary or the Custodian in respect thereof and at such time held under this Deposit Agreement, subject as to cash to the provisions of Section 4.5.

SECTION 1.9 Dollars.

The term "Dollars" shall mean United States dollars.

SECTION 1.10 DTC.

The term "DTC" shall mean The Depository Trust Company or its successor.

SECTION 1.11 Foreign Registrar.

The term "Foreign Registrar" shall mean the entity that presently carries out the duties of registrar for the Shares or any successor as registrar for the Shares and

3

any other agent of the Company for the transfer and registration of Shares, including without limitation any securities depository for the Shares.

SECTION 1.12 Holder.

The term "Holder" shall mean any person holding a Receipt or a security entitlement or other interest in American Depositary Shares, whether for its own account or for the account of another person, but that is not the Owner of that Receipt or those American Depositary Shares.

SECTION 1.13 Owner.

The term "Owner" shall mean the person in whose name American Depositary Shares are registered on the books of the Depositary maintained for such purpose.

SECTION 1.14 Receipts.

The term "Receipts" shall mean the American Depositary Receipts issued hereunder evidencing certificated American Depositary Shares, as the same may be amended from time to time in accordance with the provisions hereof.

SECTION 1.15 Registrar.

The term "Registrar" shall mean any bank or trust company having an office in the Borough of Manhattan, The City of New York, that is appointed by the Depositary to register American Depositary Shares and transfers of American Depositary Shares as herein provided.

SECTION 1.16 Restricted Securities.

The term "Restricted Securities" shall mean Shares, or American Depositary Shares representing Shares, that are acquired directly or indirectly from the Company or its affiliates (as defined in Rule 144 under the Securities Act of 1933) in a transaction or chain of transactions not involving any public offering, or that are subject to resale limitations under Regulation D under the Securities Act of 1933 or both, or which are held directly or indirectly by an officer, director (or persons performing similar functions) or other affiliate of the Company, or that would require registration under the Securities Act of 1933 in connection with the offer and sale thereof in the United States, or that are subject to other restrictions on sale or deposit under the laws of the United States or Denmark, or under a shareholder agreement or the articles of association or similar document of the Company.

4

SECTION 1.17 Securities Act of 1933.

The term "Securities Act of 1933" shall mean the United States Securities Act of 1933, as from time to time amended.

SECTION 1.18 Shares.

The term "Shares" shall mean ordinary shares of the Company that are validly issued and outstanding and fully paid, nonassessable and that were not issued in violation of any pre-emptive or similar rights of the holders of outstanding securities of the Company; provided, however, that, if there shall occur any change in nominal value, a split-up or consolidation or any other reclassification or, upon the occurrence of an event described in Section 4.8, an exchange or conversion in respect of the Shares of the Company, the term "Shares" shall thereafter also mean the successor securities resulting from such change in nominal value, split-up or consolidation or such other reclassification or such exchange or conversion.

ARTICLE 2. FORM OF RECEIPTS, DEPOSIT OF SHARES, DELIVERY, TRANSFER AND SURRENDER OF AMERICAN DEPOSITARY SHARES

SECTION 2.1 Form of Receipts; Registration and Transferability of American Depositary Shares.

Definitive Receipts shall be substantially in the form set forth in Exhibit A annexed to this Deposit Agreement, with appropriate insertions, modifications and omissions, as hereinafter provided. No Receipt shall be entitled to any benefits under this Deposit Agreement or be valid or obligatory for any purpose, unless such Receipt shall have been (i) executed by the Depositary by the manual signature of a duly authorized officer of the Depositary or (ii) executed by the facsimile signature of a duly authorized officer of the Depositary and countersigned by the manual signature of a duly authorized signatory of the Depositary or a Registrar. The Depositary shall maintain books on which (x) each Receipt so executed and delivered as hereinafter provided and the transfer of each such Receipt shall be registered and (y) all American Depositary Shares delivered as hereinafter provided and all registrations of transfer of American Depositary Shares shall be registered. A Receipt bearing the facsimile signature of a person that was at any time a proper officer of the Depositary shall, subject to the other provisions of this paragraph, bind the Depositary, notwithstanding that such person was not a proper officer of the Depositary on the date of issuance of that Receipt.

The Receipts may be endorsed with or have incorporated in the text thereof such legends or recitals or modifications not inconsistent with the provisions of this Deposit Agreement as may be required by the Depositary or required to comply with any applicable law or regulations thereunder or with the rules and regulations of any securities exchange upon which American Depositary Shares may be listed or to conform with any usage with respect

5

American Depositary Shares evidenced by a Receipt, when properly endorsed or accompanied by proper instruments of transfer, shall be transferable as certificated registered securities under the laws of the State of New York. American Depositary Shares not evidenced by Receipts shall be transferable as uncertificated registered securities under the laws of the State of New York. The Depositary, notwithstanding any notice to the contrary, may treat the Owner of American Depositary Shares as the absolute owner thereof for the purpose of determining the person entitled to distribution of dividends or other distributions or to any notice provided for in this Deposit Agreement and for all other purposes, and neither the Depositary nor the Company shall have any obligation or be subject to any liability under this Deposit Agreement to any Holder of American Depositary Shares unless that Holder is the Owner of those American Depositary Shares.

# SECTION 2.2 Deposit of Shares.

Subject to the terms and conditions of this Deposit Agreement, Shares or evidence of rights to receive Shares may be deposited by delivery thereof to any Custodian hereunder, accompanied by any appropriate instruments or instructions for transfer, or endorsement, in form satisfactory to the Custodian, together with all such certifications as may reasonably be required by the Depositary or the Custodian in accordance with the provisions of this Deposit Agreement, and, if the Depositary reasonably requires, together with a written order directing the Depositary to deliver to, or upon the written order of, the person or persons stated in such order, the number of American Depositary Shares representing such deposit.

No Share shall be accepted for deposit unless accompanied by evidence satisfactory to the Depositary that any necessary approval, exemption or derogation has been granted by any governmental body in each applicable jurisdiction that is then performing the function of the regulation of currency exchange. If required by the Depositary, Shares presented for deposit at any time, whether or not the transfer books of the Company or the Foreign Registrar, if applicable, are closed, shall also be accompanied by an agreement or assignment, or other instrument reasonably satisfactory to the Depositary, which will provide for the prompt transfer to the Custodian of any dividend, or right to subscribe for additional Shares or to receive other property which any person in whose name the Shares are or have been recorded may thereafter receive upon or in respect of such deposited Shares, or in lieu thereof, such agreement of indemnity or other agreement as shall be reasonably satisfactory to the Depositary.

The Depositary and the Custodian shall refuse to accept Shares for deposit if the Depositary has received a notice from the Company that the Company has restricted transfer of those Shares under the Company's Articles of Association or any applicable laws or that the deposit would result in any violation of the Company's Articles of Association or any applicable laws. The Company shall notify the Depositary

6

in writing with respect to any restrictions on transfer of its Shares for deposit under this Deposit Agreement.

At the request and risk and expense of any person proposing to deposit Shares, and for the account of such person, the Depositary may receive certificates for Shares or receive Shares by way of Share registration with the Transfer Agent and Registrar to be deposited, together with the other instruments herein specified, for the purpose of forwarding such Share certificates to the Custodian for deposit hereunder.

Upon each delivery to a Custodian of a certificate or certificates for Shares to be deposited hereunder, together with the other documents specified above, such Custodian shall, as soon as transfer and recordation can be accomplished, present such certificate or certificates to the Company or the Foreign Registrar, if applicable, for transfer and recordation of the Shares being deposited in the name of the Depositary or its nominee or such Custodian or its nominee.

Deposited Securities shall be held by the Depositary or by a Custodian for the account and to the order of the Depositary or at such other place or places as the Depositary shall determine.

# SECTION 2.3 Delivery of American Depositary Shares.

Upon receipt by any Custodian of any deposit pursuant to Section 2.2 hereunder, together with the other documents required as specified above, such Custodian shall notify the Depositary of such deposit and the person or persons to whom or upon whose written order American Depositary Shares are deliverable in respect thereof and the number of American Depositary Shares to be so delivered. Such notification shall be made by letter or, at the request, risk and expense of the person making the deposit, by cable, telex or facsimile transmission (and in addition, if the transfer books of the Company or the Foreign Registrar, if applicable, are open, the Depositary may in its sole discretion require a proper acknowledgment or other evidence from the Company or the Foreign Registrar that any Deposited Securities have been recorded upon the books of the Company or the Foreign Registrar, if applicable, in the name of the Depositary or its nominee or such Custodian or its nominee). Upon receiving such notice from such Custodian, or upon the receipt of Shares or evidence of the right to receive Shares by the Depositary, the Depositary, subject to the terms and conditions of this Deposit Agreement, shall deliver, to or upon the order of the person or persons entitled thereto, the number of American Depositary Shares issuable in respect of that deposit, but only upon payment to the Depositary of the fees and expenses of the Depositary for the delivery of such American Depositary Shares as provided in Section 5.9, and of all taxes and governmental charges and fees payable in connection with such deposit and the transfer of the Deposited Securities.

The Depositary, subject to the terms and conditions of this Deposit Agreement, shall register transfers of American Depositary Shares on its transfer books from time to time, upon (i) in the case of certificated American Depositary Shares, surrender of the Receipt evidencing those American Depositary Shares, by the Owner in person or by a duly authorized attorney, properly endorsed or accompanied by proper instruments of transfer or (ii) in the case of uncertificated American Depositary Shares, receipt from the Owner of a proper instruction (including, for the avoidance of doubt, instructions through DRS and Profile as provided in Section 2.10), and, in either case, duly stamped as may be required by the laws of the State of New York and of the United States of America. Thereupon the Depositary shall deliver those American Depositary Shares to or upon the order of the person entitled thereto.

The Depositary, subject to the terms and conditions of this Deposit Agreement, shall upon surrender of a Receipt or Receipts for the purpose of effecting a split-up or combination of such Receipt or Receipts, execute and deliver a new Receipt or Receipts for any authorized number of American Depositary Shares requested, evidencing the same aggregate number of American Depositary Shares as the Receipt or Receipts surrendered.

The Depositary, upon surrender of certificated American Depositary Shares for the purpose of exchanging for uncertificated American Depositary Shares, shall cancel those certificated American Depositary Shares and send the Owner a statement confirming that the Owner is the owner of the same number of uncertificated American Depositary Shares. The Depositary, upon receipt of a proper instruction (including, for the avoidance of doubt, instructions through DRS and Profile as provided in Section 2.10) from the Owner of uncertificated American Depositary Shares for the purpose of exchanging for certificated American Depositary Shares, shall cancel those uncertificated American Depositary Shares and deliver to the Owner the same number of certificated American Depositary Shares.

The Depositary may appoint one or more co-transfer agents for the purpose of effecting registration of transfers of American Depositary Shares and combinations and split-ups of Receipts at designated transfer offices on behalf of the Depositary. In carrying out its functions, a co-transfer agent may require evidence of authority and compliance with applicable laws and other requirements by Owners or persons entitled to American Depositary Shares and will be entitled to protection and indemnity to the same extent as the Depositary.

8

SECTION 2.5 Surrender of American Depositary Shares and Withdrawal of Deposited Securities.

Upon surrender at the Corporate Trust Office of the Depositary of American Depositary Shares for the purpose of withdrawal of the Deposited Securities represented thereby, and upon payment of the fee of the Depositary for the surrender of American Depositary Shares as provided in Section 5.9 and payment of all taxes and governmental charges payable in connection with such surrender and withdrawal of the Deposited Securities, and subject to the terms and conditions of this Deposit Agreement and applicable law, the Owner of those American Depositary Shares shall be entitled to delivery, to him or as instructed, of the amount of Deposited Securities at the time represented by those American Depositary Shares. Such delivery shall be made, as hereinafter provided, without unreasonable delay.

A Receipt surrendered for such purposes may be required by the Depositary to be properly endorsed in blank or accompanied by proper instruments of transfer in blank. The Depositary may require the surrendering Owner to execute and deliver to the Depositary a written order directing the Depositary to cause the Deposited Securities being withdrawn to be delivered to or upon the written order of a person or persons designated in such order. Thereupon the Depositary shall direct the Custodian to deliver at the office of such Custodian, subject to Sections 2.6, 3.1 and 3.2 and to the other terms and conditions of this Deposit Agreement, to or upon the written order of the person or persons designated in the order delivered to the Depositary as above provided, the amount of Deposited Securities represented by the surrendered American Depositary Shares, except that the Depositary may make delivery to such person or persons at the Corporate Trust Office of the Depositary of any dividends or distributions with respect to the Deposited Securities represented by those American Depositary Shares, or of any proceeds of sale of any dividends, distributions or rights, which may at the time be held by the Depositary.

At the request, risk and expense of any Owner so surrendering American Depositary Shares, and for the account of such Owner, the Depositary shall direct the Custodian to forward any cash or other property (other than rights) comprising, and forward a certificate or certificates, if applicable, and other proper documents of title for, the Deposited Securities represented by the surrendered American Depositary Shares to the Depositary for delivery at the Corporate Trust Office of the Depositary. Such direction shall be given by letter or, at the request, risk and expense of such Owner, by cable, telex or facsimile transmission.

Neither the Depositary nor the Custodian shall deliver Shares (other than to the Company or its agent as contemplated by Section 4.08), or otherwise permit Shares to be withdrawn from the facility created hereby, except upon the surrender of American Depositary Shares or in connection with a sale permitted under Section 3.2, 4.3, 4.11 or 6.2 of this Agreement.

9

SECTION 2.6 Limitations on Delivery, Transfer and Surrender of American Depositary Shares.

As a condition precedent to the delivery, registration of transfer or surrender of any American Depositary Shares or split-up or combination of any Receipt or withdrawal of any Deposited Securities, the Depositary, Custodian or Registrar may require payment from the depositor of Shares or the presenter of the Receipt or instruction for registration of transfer or surrender of American Depositary Shares not evidenced by a Receipt of a sum sufficient to reimburse it for any tax or other governmental charge and any stock transfer or registration fee with respect thereto (including any such tax or charge and fee with respect to Shares being deposited or withdrawn) and payment of any applicable fees as herein provided, may require the production of proof satisfactory to it as to the identity and genuineness of any signature and may also require compliance with any regulations the Depositary may establish consistent with the provisions of this Deposit Agreement, including, without limitation, this Section 2.6.

The delivery of American Depositary Shares against deposit of Shares generally or against deposit of particular Shares may be suspended, or the transfer of American Depositary Shares in particular instances may be refused, or the registration of transfer of outstanding American Depositary Shares generally may be suspended, during any period when the transfer books of the Depositary are closed, or if any such action is deemed necessary or advisable by the Depositary or the Company at any time or from time to time because of any requirement of law or of any government or governmental body or commission, or under any provision of this Deposit Agreement, or for any other reason, subject to the provisions of the following sentence.

Notwithstanding anything to the contrary in this Deposit Agreement, the surrender of outstanding American Depositary Shares and withdrawal of Deposited Securities may not be suspended subject only to (i) temporary delays caused by closing the transfer books of the Depositary or the Company or the Foreign

Registrar, if applicable, or the deposit of Shares in connection with voting at a shareholders' meeting, or the payment of dividends, (ii) the payment of fees, taxes and similar charges, and (iii) compliance with any U.S. or foreign laws or governmental regulations relating to the American Depositary Shares or to the withdrawal of the Deposited Securities. Without limitation of the foregoing, the Depositary shall not knowingly accept for deposit under this Deposit Agreement any Shares which would be required to be registered under the provisions of the Securities Act of 1933 for public offer and sale in the United States unless a registration statement is in effect as to such Shares for such offer and sale or an exemption from registration is available such that the American Depositary Shares may be transferred without restriction.

SECTION 2.7 Lost Receipts, etc.

cancelled.

In case any Receipt shall be mutilated, destroyed, lost or stolen, the Depositary shall deliver to the Owner the American Depositary Shares evidenced by that Receipt in uncertificated form or, if requested by the Owner, execute and deliver a new Receipt of like tenor in exchange and substitution for such mutilated Receipt, upon

10

cancellation thereof, or in lieu of and in substitution for such destroyed, lost or stolen Receipt. Before the Depositary shall deliver American Depositary Shares in uncertificated form or execute and deliver a new Receipt, in substitution for a destroyed, lost or stolen Receipt, the Owner thereof shall have (a) filed with the Depositary (i) a request for such execution and delivery before the Depositary has notice that the Receipt has been acquired by a bona fide purchaser and (ii) a sufficient indemnity bond and (b) satisfied any other reasonable requirements imposed by the Depositary.

SECTION 2.8 Cancellation and Destruction of Surrendered Receipts.

All Receipts surrendered to the Depositary shall be cancelled by the Depositary. The Depositary is authorized to destroy Receipts so

SECTION 2.9 Pre-Release of American Depositary Shares.

Unless requested in writing by the Company to cease doing so, notwithstanding Section 2.3 hereof, the Depositary may deliver American Depositary Shares prior to the receipt of Shares pursuant to Section 2.2 (a "Pre-Release"). The Depositary may, pursuant to Section 2.5, deliver Shares upon the surrender of American Depositary Shares that have been Pre-Released, whether or not such cancellation is prior to the termination of such Pre-Release. The Depositary may receive American Depositary Shares in lieu of Shares in satisfaction of a Pre-Release. Each Pre-Release will be (a) preceded or accompanied by a written representation from the person to whom American Depositary Shares or Shares are to be delivered, that such person, or its customer, (i) beneficially owns the Shares or American Depositary Shares to be remitted, as the case may be, (ii) assigns all beneficial right, title and interest in such American Depositary Shares or Shares, as the case may be, to the Depositary in its capacity as such and for the benefit of the Owners and (iii) will not take any action with respect to such American Depositary Shares or Shares, as the case may be, that is inconsistent with the transfer of beneficial ownership (including, without the consent of the Depositary, disposing of such American Depositary Shares or Shares, as the case may be), other than in satisfaction of the Pre-Release, (b) at all times fully collateralized with cash or such other collateral as the Depositary deems appropriate, (c) terminable by the Depositary on not more than five (5) business days' notice, and (d) subject to such further indemnities and credit regulations as the Depositary deems appropriate. The number of Shares represented by American Depositary Shares which are outstanding at any time as a result of Pre-Release will not normally exceed thirty percent (30%) of the Shares deposited hereunder; provided, however, that the Depositary reserves the right to disregard such limit from time to time as it reasonably deems appropriate and may, with the prior written consent of the Company, change that limit for purposes of general application. The Depositary will also set Dollar limits with respect to Pre-Release transactions with any particular Pre-Releasee on a case-by-case basis as the Depositary deems appropriate. The collateral referred to in item (b) above shall be held by the Depositary as security for the performance of the Pre-Releasee's obligations in connection the related Pre-Release

11

transaction, including the Pre-Releasee's obligation to deliver Shares or American Depositary Shares upon termination of that Pre-Release transaction (and shall not, for the avoidance of doubt, constitute Deposited Securities).

The Depositary may retain for its own account any compensation received by it in connection with the foregoing.

SECTION 2.10 DTC Direct Registration System and Profile Modification System.

- (a) Notwithstanding the provisions of Section 2.4, the parties acknowledge that the Direct Registration System ("DRS") and Profile Modification System ("Profile") shall apply to uncertificated American Depositary Shares upon acceptance thereof to DRS by DTC. DRS is the system administered by DTC pursuant to which the Depositary may register the ownership of uncertificated American Depositary Shares, which ownership shall be evidenced by periodic statements issued by the Depositary to the Owners entitled thereto. Profile is a required feature of DRS which allows a DTC participant, claiming to act on behalf of an Owner of American Depositary Shares, to direct the Depositary to register a transfer of those American Depositary Shares to DTC or its nominee and to deliver those American Depositary Shares to the DTC account of that DTC participant without receipt by the Depositary of prior authorization from the Owner to register such transfer.
- (b) In connection with and in accordance with the arrangements and procedures relating to DRS/Profile, the parties understand that the Depositary will not verify, determine or otherwise ascertain that the DTC participant which is claiming to be acting on behalf of an Owner in requesting a registration of transfer and delivery as described in subsection (a) has the actual authority to act on behalf of the Owner (notwithstanding any requirements under the Uniform Commercial Code). For the avoidance of doubt, the provisions of Sections 5.3 and 5.8 shall apply to the matters arising from the use of the DRS. The parties agree that the Depositary's reliance on and compliance with instructions received by the Depositary through the DRS/Profile System and in accordance with this Deposit Agreement shall not constitute negligence or bad faith on the part of the Depositary.

# ARTICLE 3. CERTAIN OBLIGATIONS OF OWNERS AND HOLDERS OF AMERICAN DEPOSITARY SHARES

SECTION 3.1 Filing Proofs, Certificates and Other Information.

Any person presenting Shares for deposit or any Owner or Holder may be required from time to time to file with the Depositary or the Custodian such proof of citizenship or residence, exchange control approval, or such information relating to the registration on the books of the Company or the Foreign Registrar, if applicable, to execute such certificates and to make such representations and warranties, as the

12

Depositary may deem necessary or proper. The Depositary may withhold the delivery or registration of transfer of American Depositary Shares or the distribution of any dividend or sale or distribution of rights or of the proceeds thereof or the delivery of any Deposited Securities until such proof or other information is filed or such certificates are executed or such representations and warranties made. If requested in writing by the Company, the Depositary will provide the Company, as promptly as reasonably practicable and at the Company's expense, with copies of any such proofs, certificates or other information that it receives pursuant to this Section 3.1, to the extent that disclosure is permitted under applicable law.

### SECTION 3.2 Liability of Owner for Taxes.

If any tax or other governmental charge shall become payable by the Custodian or the Depositary with respect to any American Depositary Shares or any Deposited Securities represented by any American Depositary Shares, such tax or other governmental charge shall be payable by the Owner of such American Depositary Shares to the Depositary. The Depositary may refuse to register any transfer of those American Depositary Shares or any withdrawal of Deposited Securities represented by those American Depositary Shares until such payment is made, and may withhold any dividends or other distributions, or may sell for the account of the Owner thereof any part or all of the Deposited Securities represented by those American Depositary Shares, and may apply such dividends or other distributions or the proceeds of any such sale in payment of such tax or other governmental charge and the Owner of such American Depositary Shares shall remain liable for any deficiency.

# SECTION 3.3 Warranties on Deposit of Shares.

Every person depositing Shares under this Deposit Agreement shall be deemed thereby to represent and warrant that such Shares and each certificate therefor, if applicable, are validly issued, fully paid, nonassessable and were not issued in violation of any preemptive rights of the holders of outstanding Shares and that the person making such deposit is duly authorized to make such deposit. Every such person shall also be deemed to represent that the deposit of such Shares and the sale of American Depositary Shares representing such Shares by that person are not restricted under the Securities Act of 1933. Such representations and warranties shall survive the deposit of Shares and delivery of American Depositary Shares.

### SECTION 3.4 Disclosure of Interests.

The Company may from time to time request Owners or Holders or former Owners or Holders to provide information as to the capacity in which they hold or held American Depositary Shares and regarding the identity of any other persons then or previously interested in such American Depositary Shares and the nature of such interest and various other matters. Each such Owner or Holder agrees to provide any such

13

information reasonably requested by the Company or the Depositary pursuant to this Section 3.4 whether or not still an Owner or Holder at the time of such request. The Depositary agrees to use its reasonable efforts, at the Company's expense, to comply with written instructions received from the Company requesting that the Depositary forward any such requests to such Owners or Holders and to the last known address, if any, of such former Owners or Holders and to forward to the Company any responses to such requests received by the Depositary. However, nothing in this Section 3.4 shall be interpreted as obligating the Depositary to provide or obtain any such information not provided to the Depositary by such Owners or Holders or former Owners or Holders.

# ARTICLE 4. THE DEPOSITED SECURITIES

# SECTION 4.1 Cash Distributions.

Whenever the Depositary shall receive any cash dividend or other cash distribution on any Deposited Securities, the Depositary shall, as promptly as practicable, subject to the provisions of Section 4.5, convert such dividend or distribution into Dollars and shall distribute the amount thus received (net of the fees and expenses of the Depositary as provided in Section 5.9) to the Owners entitled thereto, in proportion to the number of American Depositary Shares representing such Deposited Securities held by them respectively; provided, however, that in the event that the Custodian, the Depositary or the Company shall be required by applicable law to withhold and does withhold from such cash dividend or such other cash distribution an amount on account of taxes or other governmental charges, the amount distributed to the Owner of the American Depositary Shares representing such Deposited Securities shall be reduced accordingly. The Depositary shall distribute only such amount, however, as can be distributed without attributing to any Owner a fraction of one cent. Any such fractional amounts shall be rounded to the nearest whole cent and so distributed to Owners entitled thereto. The Company or its agent will remit to the appropriate governmental agency in Denmark all amounts withheld and owing to such agency. The Depositary will, as promptly as practicable, forward to the Company or its agent such information from its records as the Company may reasonably request to enable the Company or its agent to file necessary reports with governmental agencies.

# SECTION 4.2 Distributions Other Than Cash, Shares or Rights.

Subject to the provisions of Sections 4.11 and 5.9, whenever the Depositary shall receive any distribution other than a distribution described in Section 4.1, 4.3 or 4.4, the Depositary shall cause the securities or property received by it to be distributed to the Owners entitled thereto, after deduction or upon payment of any fees and expenses of the Depositary or any taxes or other governmental charges, imposed under applicable law, in proportion to the number of American Depositary Shares representing such Deposited Securities held by them respectively, in any manner that the Depositary may deem equitable and practicable for accomplishing such distribution;

provided, however, that if in the opinion of the Depositary such distribution cannot be made proportionately among the Owners entitled thereto, or if for any other reason (including, but not limited to, any requirement under applicable law that the Company or the Depositary withhold an amount on account of taxes or other governmental charges or that such securities must be registered under the Securities Act of 1933 in order to be distributed to Owners or Holders) the Depositary deems such distribution not to be feasible, the Depositary may adopt such method as it may deem equitable and practicable for the purpose of effecting such distribution, including, but not limited to, the public or private sale of the securities or property thus received, or any part thereof, and the net proceeds of any such sale (net of the fees and expenses of the Depositary as provided in Section 5.9) shall be distributed by the Depositary to the Owners entitled thereto, all in the manner and subject to the conditions described in Section 4.1. The Depositary may withhold any distribution of securities under this Section 4.2 if it has not received satisfactory assurances from the Company that the distribution does not require registration under the Securities Act of 1933. The Depositary may sell, by public or private sale, an amount of securities or other property it would otherwise distribute under this Section 4.2 that is sufficient to pay its fees and expenses in respect of that distribution.

### SECTION 4.3 Distributions in Shares.

If any distribution upon any Deposited Securities consists of a dividend in, or free distribution of, Shares, the Depositary may, and shall if the Company shall so request in writing, deliver to the Owners entitled thereto, in proportion to the number of American Depositary Shares representing such Deposited Securities held by them respectively, an aggregate number of American Depositary Shares representing the amount of Shares received as such dividend or free distribution, subject to the terms and conditions of the Deposit Agreement with respect to the deposit of Shares and after deduction or upon issuance of American Depositary Shares, including the withholding of any tax or other governmental charge as provided in Section 4.11 and the payment of the fees and expenses of the Depositary as provided in Section 5.9 (and the Depositary may sell, by public or private sale, an amount of the Shares received sufficient to pay its fees and expenses in respect of that distribution). The Depositary may withhold any such delivery of American Depositary Shares if it has not received satisfactory assurances from the Company that such distribution does not require registration under the Securities Act of 1933. In lieu of delivering fractional American Depositary Shares in any such case, the Depositary shall sell the amount of Shares represented by the aggregate of such fractions and distribute the net proceeds, all in the manner and subject to the conditions described in Section 4.1. If additional American Depositary Shares are not so delivered, each American Depositary Share shall thenceforth also represent the additional Shares distributed upon the Deposited Securities represented thereby.

15

### SECTION 4.4 Rights.

In the event that the Company shall offer or cause to be offered to the holders of any Deposited Securities any rights to subscribe for additional Shares or any rights of any other nature, the Depositary shall, after consultation with the Company, to the extent practicable, have discretion as to the procedure to be followed in making such rights available to any Owners or in disposing of such rights on behalf of any Owners and making the net proceeds available to such Owners or, if by the terms of such rights offering or for any other reason, the Depositary may not either make such rights available to any Owners or dispose of such rights and make the net proceeds available to such Owners, then the Depositary shall allow the rights to lapse. If at the time of the offering of any rights the Depositary determines in its discretion that it is lawful and feasible to make such rights available to all or certain Owners but not to other Owners, the Depositary may distribute to any Owner to whom it determines the distribution to be lawful and feasible, in proportion to the number of American Depositary Shares held by such Owner, warrants or other instruments therefor in such form as it deems appropriate.

In circumstances in which rights would otherwise not be distributed, if an Owner requests the distribution of warrants or other instruments in order to exercise the rights allocable to the American Depositary Shares of such Owner hereunder, the Depositary will make such rights available to such Owner upon written notice from the Company to the Depositary that (a) the Company has elected in its sole discretion to permit such rights to be exercised and (b) such Owner has executed such documents as the Company has determined in its sole discretion are reasonably required under applicable law.

If the Depositary has distributed warrants or other instruments for rights to all or certain Owners, then upon instruction from such an Owner pursuant to such warrants or other instruments to the Depositary from such Owner to exercise such rights, upon payment by such Owner to the Depositary for the account of such Owner of an amount equal to the purchase price of the Shares to be received upon the exercise of the rights, and upon payment of the fees and expenses of the Depositary and any other charges as set forth in such warrants or other instruments, the Depositary shall, on behalf of such Owner, exercise the rights and purchase the Shares, and the Company shall cause the Shares so purchased to be delivered to the Depositary on behalf of such Owner. As agent for such Owner, the Depositary will cause the Shares so purchased to be deposited pursuant to Section 2.2, and shall, pursuant to Section 2.3, deliver American Depositary Shares to such Owner. In the case of a distribution pursuant to the second paragraph of this Section, such deposit shall be made, and depositary shares shall be delivered, under depositary arrangements which provide for issuance of depositary shares subject to the appropriate restrictions on sale, deposit, cancellation, and transfer under applicable United States laws.

If the Depositary determines in its reasonable discretion that it is not lawful and feasible to make such rights available to all or certain Owners, it may sell the rights, warrants or other instruments in proportion to the number of American Depositary

16

Shares held by the Owners to whom it has determined it may not lawfully or feasibly make such rights available, and allocate the net proceeds of such sales (net of the fees and expenses of the Depositary as provided in Section 5.9 and all taxes and governmental charges payable in connection with such rights and subject to the terms and conditions of this Deposit Agreement) for the account of such Owners otherwise entitled to such rights, warrants or other instruments, upon an averaged or other practical basis without regard to any distinctions among such Owners because of exchange restrictions or the date of delivery of any American Depositary Shares or otherwise.

The Depositary will not offer rights to Owners unless both the rights and the securities to which such rights relate are either exempt from registration under the Securities Act of 1933 with respect to a distribution to all Owners or are registered under the provisions of such Act; <u>provided</u>, that nothing in this Deposit Agreement shall create any obligation on the part of the Company to file a registration statement with respect to such rights or underlying securities or to endeavor to have such a registration statement declared effective. If an Owner requests the distribution of warrants or other instruments, notwithstanding that there has been no such registration under the Securities Act of 1933, the Depositary shall not effect such distribution unless it has received an opinion from recognized counsel in the United States for the Company upon which the Depositary may rely that such distribution to such

Owner is exempt from such registration <u>provided</u>, <u>however</u>, that any opinion requested by an Owner to be delivered by the Company's counsel shall be prepared by the Company's counsel at the Owner's expense.

The Depositary shall not be responsible for any failure to determine that it may be lawful or feasible to make such rights available to Owners in general or any Owner in particular.

SECTION 4.5 Conversion of Foreign Currency.

Whenever the Depositary or the Custodian shall receive foreign currency, by way of dividends or other distributions or the net proceeds from the sale of securities, property or rights, and if at the time of the receipt thereof the foreign currency so received can in the judgment of the Depositary be converted on a reasonable basis into Dollars and the resulting Dollars transferred to the United States, the Depositary shall, as promptly as practicable, convert or cause to be converted by sale or in any other manner that it may determine such foreign currency into Dollars, and such Dollars shall be distributed to the Owners entitled thereto or, if the Depositary shall have distributed any warrants or other instruments which entitle the holders thereof to such Dollars, then to the holders of such warrants and/or instruments upon surrender thereof for cancellation. Such distribution may be made upon an averaged or other practicable basis without regard to any distinctions among Owners on account of exchange restrictions, the date of delivery of any American Depositary Shares or otherwise and shall be net of any expenses of conversion into Dollars incurred by the Depositary as provided in Section 5.9.

17

If such conversion or distribution can be effected only with the approval or license of any government or agency thereof, the Depositary shall file such application for approval or license, if any, as it may deem desirable.

If at any time the Depositary shall determine that in its judgment any foreign currency received by the Depositary or the Custodian is not convertible on a reasonable basis into Dollars transferable to the United States, or if any approval or license of any government or agency thereof which is required for such conversion is denied or in the opinion of the Depositary is not obtainable, or if any such approval or license is not obtained within a reasonable period as determined by the Depositary, the Depositary may distribute the foreign currency (or an appropriate document evidencing the right to receive such foreign currency) received by the Depositary to, or in its discretion may hold such foreign currency uninvested and without liability for interest thereon for the respective accounts of, the Owners entitled to receive the same.

If any such conversion of foreign currency, in whole or in part, cannot be effected for distribution to some of the Owners entitled thereto, the Depositary may in its discretion make such conversion and distribution in Dollars to the extent permissible to the Owners entitled thereto and may distribute the balance of the foreign currency received by the Depositary to, or hold such balance uninvested and without liability for interest thereon for the respective accounts of, the Owners entitled thereto.

SECTION 4.6 Fixing of Record Date.

Whenever any cash dividend or other cash distribution shall become payable or any distribution other than cash shall be made, or whenever rights shall be issued with respect to the Deposited Securities, or whenever the Depositary shall receive notice of any meeting of holders of Shares or other Deposited Securities, or whenever for any reason the Depositary causes a change in the number of Shares that are represented by each American Depositary Share, or whenever the Depositary shall find it necessary or convenient, the Depositary shall fix a record date (the "Record Date") (a) for the determination of the Owners who shall be (i) entitled to receive such dividend, distribution or rights or the net proceeds of the sale thereof, (ii) entitled to give instructions for the exercise of voting rights at any such meeting or (iii) responsible for any fee or charge assessed by the Depositary pursuant to this Deposit Agreement, or (b) on or after which each American Depositary Share will represent the changed number of Shares. Subject to the provisions of Sections 4.1 through 4.5 and to the other terms and conditions of this Deposit Agreement, the Owners on such Record Date shall be entitled, as the case may be, to receive the amount distributable by the Depositary with respect to such dividend or other distribution or such rights or the net proceeds of sale thereof in proportion to the number of American Depositary Shares held by them respectively and to give voting instructions and to act in respect of any other such matter.

18

# SECTION 4.7 Voting of Deposited Securities.

Upon receipt of notice of any meeting or solicitation of proxies or consents of holders of Shares or other Deposited Securities, if requested in writing by the Company, the Depositary shall, as soon as practicable thereafter, mail to the Owners a notice, the form of which notice shall be in the sole discretion of the Depositary, which shall contain (a) such information (including, without limitation, solicitation materials) as is contained in such notice of meeting received by the Depositary from the Company, (b) a statement that the Owners as of the close of business on a specified record date will be entitled, subject to any applicable provision of Danish law and of the articles of association or similar documents of the Company, to instruct the Depositary as to the exercise of the voting rights, if any, pertaining to the amount of Shares or other Deposited Securities represented by their respective American Depositary Shares and (c) a statement as to the manner in which such instructions may be given, including an express indication that instructions may be given or deemed given in accordance with the last sentence of this paragraph if no instruction is received, to the Depositary to give a discretionary proxy to a person designated by the Company. Upon the written request of an Owner of American Depositary Shares on such record date, received on or before the date established by the Depositary for such purpose, the Depositary shall endeavor, in so far as practicable, to vote or cause to be voted the amount of Shares or other Deposited Securities represented by those American Depositary Shares in accordance with the instructions set forth in such request. The Depositary shall not itself exercise any voting discretion over any Deposited Securities. If (i) the Company instructed the Depositary to act under this Section 4.7 and (ii) no instructions are received by the Depositary from an Owner with respect to a matter and an amount of American Depositary Shares of that Owner on or before the date established by the Depositary for such purpose, the Depositary shall deem that Owner to have instructed the Depositary to give a discretionary proxy to a person designated by the Company with respect to that matter and the amount of Deposited Securities represented by that amount of American Depositary Shares and the Depositary shall give a discretionary proxy to a person designated by the Company to vote that amount of Deposited Securities as to that matter, except that no such instruction shall be deemed given and no such discretionary proxy shall be given with respect to any matter as to which the Company informs the Depositary (and the Company agrees to provide such information as promptly as practicable in writing, if applicable) that (x) the Company does not wish such proxy given, (y) substantial opposition exists or (z) such matter materially and adversely affects the rights of holders of Shares.

There can be no assurance that Owners generally or any Owner in particular will receive the notice described in the preceding paragraph sufficiently prior to the instruction cutoff date to ensure that the Depositary will vote the Shares or Deposited Securities in accordance with the provisions set forth in the preceding paragraph.

19

In order to give Owners a reasonable opportunity to instruct the Depositary as to the exercise of voting rights relating to Deposited Securities, if the Company will request the Depositary to act under this Section 4.7, the Company shall give the Depositary notice of any such meeting and details concerning the matters to be voted upon at least 45 days in advance of the meeting date.

# SECTION 4.8 Changes Affecting Deposited Securities.

Upon any change in nominal value, change in par value, split-up, consolidation or any other reclassification of Deposited Securities, or upon any recapitalization, reorganization, merger or consolidation or sale of assets affecting the Company or to which it is a party, or upon the redemption or cancellation by the Company of the Deposited Securities, any securities, cash or property which shall be received by the Depositary or a Custodian in exchange for, in conversion of, in lieu of or in respect of Deposited Securities, shall be treated as new Deposited Securities under this Deposit Agreement, and American Depositary Shares shall thenceforth represent, in addition to the existing Deposited Securities, the right to receive the new Deposited Securities so received, unless additional American Depositary Shares are delivered pursuant to the following sentence. In any such case the Depositary may deliver additional American Depositary Shares as in the case of a dividend in Shares, or call for the surrender of outstanding Receipts to be exchanged for new Receipts specifically describing such new Deposited Securities.

### SECTION 4.9 Reports.

The Depositary shall make available for inspection by Owners at its Corporate Trust Office any reports and communications, including any proxy solicitation material, received from the Company which are both (a) received by the Depositary as the holder of the Deposited Securities and (b) made generally available to the holders of such Deposited Securities by the Company. The Depositary shall also, upon written request by the Company, send to the Owners copies of such reports when furnished by the Company pursuant to Section 5.6. Any such reports and communications, including any such proxy soliciting material, furnished to the Depositary by the Company shall be furnished in English, to the extent such materials are required to be translated into English pursuant to any regulations of the Commission.

### SECTION 4.10 Lists of Owners.

Promptly upon request by the Company, the Depositary shall, at the expense of the Company, furnish to it a list, as of a recent date, of the names, addresses and holdings of American Depositary Shares by all persons in whose names American Depositary Shares are registered on the books of the Depositary.

20

# SECTION 4.11 Withholding.

In the event that the Depositary reasonably determines that any distribution in property (including Shares and rights to subscribe therefor) is subject to any tax or other governmental charge which the Depositary is obligated to withhold under applicable law, the Depositary may by public or private sale dispose of all or a portion of such property (including Shares and rights to subscribe therefor) in such amounts and in such manner as the Depositary deems necessary and practicable to pay such taxes or charges and the Depositary shall distribute the net proceeds of any such sale after deduction of such taxes or charges to the Owners entitled thereto in proportion to the number of American Depositary Shares held by them respectively.

The Depositary will, and will instruct the Custodian to, forward to the Company or its agents such information from its records as the Company may reasonably request and at its expense, to enable the Company or its agents to file the necessary tax reports with governmental authorities or agencies.

# ARTICLE 5. THE DEPOSITARY, THE CUSTODIANS AND THE COMPANY

# SECTION 5.1 Maintenance of Office and Transfer Books by the Depositary.

Until termination of this Deposit Agreement in accordance with its terms, the Depositary shall maintain in the Borough of Manhattan, The City of New York, facilities for the execution and delivery, registration, registration of transfers and surrender of American Depositary Shares in accordance with the provisions of this Deposit Agreement.

The Depositary shall keep books, at its Corporate Trust Office, for the registration of American Depositary Shares and transfers of American Depositary Shares which at all reasonable times shall be open for inspection by the Owners and the Company, provided that such inspection shall not be for the purpose of communicating with Owners in the interest of a business or object other than the business of the Company or a matter related to this Deposit Agreement or the American Depositary Shares.

The Depositary may close the transfer books, at any time or from time to time, when deemed reasonably expedient by it in connection with the performance of its duties hereunder or at the reasonable written request of the Company.

If any American Depositary Shares are listed on one or more stock exchanges in the United States, the Depositary shall act as Registrar or appoint a Registrar or one or more co-registrars for registry of such American Depositary Shares in accordance with any requirements of such exchange or exchanges.

### SECTION 5.2

Prevention or Delay in Performance by the Depositary or the Company.

Neither the Depositary nor the Company nor any of their respective directors, officers, employees, agents or affiliates shall incur any liability to any Owner or Holder (i) if by reason of any provision of any present or future law or regulation of the United States or any other country, or of any governmental or regulatory authority or stock exchange, or by reason of any provision, present or future, of the articles of association or similar document of the Company, or by reason of any provision of any securities issued or distributed by the Company, or any offering or distribution thereof, or by reason of any act of God or war or terrorism or other circumstances beyond its control, the Depositary or the Company shall be prevented, delayed or forbidden from, or be subject to any civil or criminal penalty on account of, doing or performing any act or thing which by the terms of this Deposit Agreement or the Deposited Securities it is provided shall be done or performed, (ii) by reason of any non-performance or delay, caused as aforesaid, in the performance of any act or thing which by the terms of this Deposit Agreement it is provided shall or may be done or performed, (iii) by reason of any exercise of, or failure to exercise, any discretion provided for in this Deposit Agreement, (iv) for the inability of any Owner or Holder to benefit from any distribution, offering, right or other benefit which is made available to holders of Deposited Securities but is not, under the terms of this Deposit Agreement, made available to Owners or Holders, or (v) for any special, consequential or punitive damages for any breach of the terms of this Deposit Agreement. Where, by the terms of a distribution pursuant to Section 4.1, 4.2 or 4.3, or an offering or distribution pursuant to Section 4.4, or for any other reason, such distribution or offering may not be made available to Owners, and the Depositary may not dispose of such distribution or offering on behalf of such Owners and make the net proceeds available to such Owners

SECTION 5.3 Obligations of the Depositary, the Custodian and the Company.

Neither the Company nor any of its directors, officers, employees, agents or affiliates assume any obligation nor shall it or any of them be subject to any liability under this Deposit Agreement to any Owner or Holder, except that the Company agrees to perform its obligations specifically set forth in this Deposit Agreement without negligence or bad faith.

Neither the Depositary nor any of its directors, officers, employees, agents or affiliates assume any obligation nor shall it or any of them be subject to any liability under this Deposit Agreement to any Owner or Holder (including, without limitation, liability with respect to the validity or worth of the Deposited Securities), except that the Depositary agrees to perform its obligations specifically set forth in this Deposit Agreement without negligence or bad faith.

22

Neither the Depositary nor the Company nor any of their respective directors, officers, employees, agents or affiliates shall be under any obligation to appear in, prosecute or defend any action, suit or other proceeding in respect of any Deposited Securities or in respect of the American Depositary Shares on behalf of any Owner or Holder or any other person.

Neither the Depositary nor the Company nor any of their respective directors, officers, employees, agents or affiliates shall be liable for any action or nonaction by it in reliance upon the advice of or information from legal counsel, accountants, any person presenting Shares for deposit, any Owner or any other person believed by any of them in good faith to be competent to give such advice or information. The Depositary and the Company and their respective directors, officers, employees, agents or affiliates may rely on and shall be protected in acting upon any written notice, request, direction or other documents believed by them to be genuine and to have been signed or presented by the proper party or parties.

The Depositary shall not be liable for any acts or omissions made by a successor depositary whether in connection with a previous act or omission of the Depositary or in connection with any matter arising wholly after the removal or resignation of the Depositary, provided that in connection with the issue out of which such potential liability arises the Depositary performed its obligations without negligence or bad faith while it acted as Depositary.

The Depositary shall not be liable for the acts or omissions of any securities depository, clearing agency or settlement system in connection with or arising out of book-entry settlement of Deposited Securities or otherwise.

The Depositary shall not be responsible for any failure to carry out any instructions to vote any of the Deposited Securities, or for the manner in which any such vote is cast or the effect of any such vote, provided that any such action or nonaction is in good faith.

No disclaimer of liability under the Securities Act of 1933 is intended by any provision of this Deposit Agreement.

SECTION 5.4 Resignation and Removal of the Depositary.

The Depositary may at any time resign as Depositary hereunder by written notice of its election so to do delivered to the Company, such resignation to take effect upon the appointment of a successor depositary and its acceptance of such appointment as hereinafter provided.

The Depositary may at any time be removed by the Company by 120 days prior written notice of such removal, to become effective upon the later of (i) the 120th

23

day after delivery of the notice to the Depositary and (ii) the appointment of a successor depositary and its acceptance of such appointment as hereinafter provided.

In case at any time the Depositary acting hereunder shall resign or be removed, the Company shall use its reasonable efforts to appoint a successor depositary, which shall be a bank or trust company having an office in the Borough of Manhattan, The City of New York or in any other place permitted by applicable law and stock exchange rules. Every successor depositary shall execute and deliver to its predecessor and to the Company an instrument in writing accepting its appointment hereunder, and thereupon such successor depositary, without any further act or deed, shall become fully vested with all the rights, powers, duties and obligations of its predecessor; but such predecessor, nevertheless, upon payment of all sums due it and on the

written request of the Company shall execute and deliver an instrument transferring to such successor all rights and powers of such predecessor hereunder, shall duly assign, transfer and deliver all right, title and interest in the Deposited Securities to such successor and shall deliver to such successor a list of the Owners of all outstanding American Depositary Shares. Any such successor depositary shall promptly mail notice of its appointment to the Owners.

Any corporation into or with which the Depositary may be merged or consolidated shall be the successor of the Depositary without the execution or filing of any document or any further act.

SECTION 5.5 The Custodians.

The Custodian shall be subject at all times and in all respects to the directions of the Depositary and shall be responsible solely to it. Any Custodian may resign and be discharged from its duties hereunder by notice of such resignation delivered to the Depositary at least 30 days prior to the date on which such resignation is to become effective. If upon such resignation there shall be no Custodian acting hereunder, the Depositary shall, promptly after receiving such notice, appoint a substitute custodian or custodians, each of which shall thereafter be a Custodian hereunder. The Depositary in its discretion may appoint a substitute or additional custodian or custodians, each of which shall thereafter be one of the Custodians hereunder. Upon demand of the Depositary any Custodian shall deliver such of the Deposited Securities held by it as are requested of it to any other Custodian or such substitute or additional custodian or custodians. Each such substitute or additional custodian shall deliver to the Depositary, forthwith upon its appointment, an acceptance of such appointment satisfactory in form and substance to the Depositary. Following any resignation or removal of the Custodian and the appointment of a substitute or additional Custodian, the Depositary will give subsequent notice thereof to the Company as promptly as practicable.

Upon the appointment of any successor depositary hereunder, each Custodian then acting hereunder shall forthwith become, without any further act or writing, the agent hereunder of such successor depositary and the appointment of such

24

successor depositary shall in no way impair the authority of each Custodian hereunder; but the successor depositary so appointed shall, nevertheless, on the written request of any Custodian, execute and deliver to such Custodian all such instruments as may be proper to give to such Custodian full and complete power and authority as agent hereunder of such successor depositary.

SECTION 5.6 Notices and Reports.

On or before the first date on which the Company gives notice, by publication or otherwise, of any meeting of holders of Shares or other Deposited Securities, or of any adjourned meeting of such holders, or of the taking of any action in respect of any cash or other distributions or the offering of any rights, the Company agrees to transmit to the Depositary and the Custodian a copy of the notice thereof in the form given or to be given to holders of Shares or other Deposited Securities.

The Company will arrange for the translation into English, if not already in English, to the extent required pursuant to any regulations of the Commission, and the prompt transmittal by the Company to the Depositary and the Custodian of such notices and any other reports and communications which are made generally available by the Company to holders of its Shares. If requested in writing by the Company, the Depositary will arrange for the mailing, at the Company's expense, of copies of such notices, reports and communications to all Owners. The Company will timely provide the Depositary with the quantity of such notices, reports, and communications, as requested by the Depositary from time to time, in order for the Depositary to effect such mailings.

SECTION 5.7 Distribution of Additional Shares, Rights, etc.

If the Company or any affiliate of the Company determines to make any issuance or distribution of (1) additional Shares, (2) rights to subscribe for Shares, (3) securities convertible into Shares, or (4) rights to subscribe for such securities (each a "Distribution"), the Company shall notify the Depositary in writing in English as promptly as practicable and in any event before the Distribution starts and, if requested in writing by the Depositary, the Company shall promptly furnish to the Depositary a written opinion from U.S. counsel for the Company that is reasonably satisfactory to the Depositary, stating whether or not the Distribution requires, or, if made in the United States, would require, registration under the Securities Act of 1933. If, in the opinion of that counsel, the Distribution requires, or, if made in the United States, would require, registration under the Securities Act of 1933, that counsel shall furnish to the Depositary a written opinion as to whether or not there is a registration statement under the Securities Act of 1933 in effect that will cover that Distribution.

The Company agrees with the Depositary that neither the Company nor any company controlled by, controlling or under common control with the Company will at any time deposit any Shares, either originally issued or previously issued and

25

reacquired by the Company or any such affiliate, unless a Registration Statement is in effect as to such Shares under the Securities Act of 1933 or the Company delivers to the Depositary an opinion of United States counsel, satisfactory to the Depositary, to the effect that, upon deposit, those Shares will be eligible for public resale without restriction in the United States without further registration under the Securities Act of 1933. Notwithstanding anything to the contrary herein, nothing in this Deposit Agreement shall be deemed to obligate the Company to file any registration statement in respect of any proposed transactions.

SECTION 5.8 Indemnification.

The Company agrees to indemnify the Depositary, its directors, employees, agents and affiliates and any Custodian against, and hold each of them harmless from, any liability or expense (including, but not limited to any fees and expenses incurred in seeking, enforcing or collecting such indemnity and the fees and expenses of counsel) which may arise out of or in connection with (a) any registration with the Commission of American Depositary Shares or Deposited Securities or the offer or sale thereof in the United States, except to the extent the liability or expense arises out of information relating to the Depositary or the Custodian furnished in writing to the Company by the Depositary expressly for use in any registration statement, proxy statement, prospectus or offering memorandum (or private placement memorandum) relating to the Shares (it being understood that, as of the date of

this Deposit Agreement, the Depositary has not furnished any information of that kind), or (b) acts performed or omitted, pursuant to the provisions of or in connection with this Deposit Agreement and of the American Depositary Shares, as the same may be amended, modified or supplemented from time to time, (i) by either the Depositary or a Custodian or their respective directors, employees, agents and affiliates, except for any liability or expense arising out of the negligence or bad faith of either of them, or (ii) by the Company or any of its directors, employees, agents and affiliates.

The indemnities contained in the preceding paragraph shall not extend to any liability or expense which arises solely and exclusively out of a Pre-Release (as defined in Section 2.9) of American Depositary Shares in accordance with Section 2.9 and which would not otherwise have arisen had such American Depositary Shares not been the subject of a Pre-Release pursuant to Section 2.9; provided, however, that the indemnities provided in the preceding paragraph shall apply to any such liability or expense (i) to the extent such liability or expense would have arisen had such American Depositary Shares not been the subject of a Pre-Release, or (ii) which may arise out of any misstatement or alleged misstatement or omission or alleged omission in any registration statement, proxy statement, prospectus (or private placement memorandum) relating to the offer of sale of American Depositary Shares, except to the extent any such liability or expense arises out of (i) information relating to the Depositary or the Custodian (other than the Company), as applicable, furnished in writing by the Depositary expressly for

26

use in any of the foregoing documents and not materially changed or altered by the Company or, (ii) if such information is provided, the failure to state a material fact necessary to make the information provided not misleading.

The Depositary agrees to indemnify the Company, its directors, officers, employees, agents and affiliates and hold them harmless from any liability or expense (including, but not limited to any fees and expenses incurred in seeking, enforcing or collecting such indemnity and the reasonable fees and expenses of counsel) which may arise out of acts performed or omitted by the Depositary or its Custodian or their respective directors, officers, employees, agents and affiliates due to their negligence or bad faith.

If an action, proceeding (including, but not limited to, any governmental investigation), claim or dispute (collectively, a "Proceeding") in respect of which indemnity may be sought by either party is brought or asserted against the other party, the party seeking indemnification (the "Indemnitee") shall promptly (and in no event more than ten (10) days after receipt of notice of such Proceeding) notify the party obligated to provide such indemnification (the "Indemnitor") of such Proceeding. The failure of the Indemnitee to so notify the Indemnitor shall not impair the Indemnitee's ability to seek indemnification from the Indemnitor (but only for costs, expenses and liabilities incurred after such notice) unless such failure adversely affects the Indemnitor's ability to adequately oppose or defend such Proceeding. Upon receipt of such notice from the Indemnitee, the Indemnitor shall be entitled to participate in such Proceeding and, to the extent that it shall so desire and provided no conflict of interest exists as specified in subparagraph (b) below or there are no other defenses available to Indemnitee as specified in subparagraph (d) below, to assume the defense thereof with counsel reasonably satisfactory to the Indemnitee (in which case all attorney's fees and expenses shall be borne by the Indemnitor and the Indemnitor shall in good faith defend the Indemnitee). The Indemnitee shall have the right to employ separate counsel in any such Proceeding and to participate in the defense thereof, but the fees and expenses of such counsel shall be borne by the Indemnitee unless (a) the Indemnitor agrees in writing to pay such fees and expenses, (b) the Indemnitee shall have reasonably and in good faith concluded that there is a conflict of interest between the Indemnitor and the Indemnitee in the conduct of the defense of such action, (c) the Indemnitor fails, within ten (10) days prior to the date the first response or appearance is required to be made in such Proceeding, to assume the defense of such Proceeding with counsel reasonably satisfactory to the Indemnitee or (d) there are legal defenses available to Indemnitee that are different from or are in addition to those available to the Indemnitor. No compromise or settlement of such Proceeding may be effected by either party without the other party's consent unless (i) there is no finding or admission of any violation of law and no effect on any other claims that may be made against such other party and (ii) the sole relief provided is monetary damages that are paid in full by the party seeking the settlement and for which the Indemnitee will not seek reimbursement of such amount from the Indemnitor. Neither party shall have any liability with respect to any compromise or

27

settlement effected without its consent, which shall not be unreasonably withheld. The Indemnitor shall have no obligation to indemnify and hold harmless the Indemnitee from any loss, expense or liability incurred by the Indemnitee as a result of a default judgment entered against the Indemnitee unless such judgment was entered after the Indemnitor agreed, in writing, to assume the defense of such proceeding.

SECTION 5.9 Charges of Depositary.

The Company agrees to pay the fees and out-of-pocket expenses of the Depositary and those of any Registrar only in accordance with agreements in writing entered into between the Depositary and the Company from time to time.

The following charges shall be incurred by any party depositing or withdrawing Shares or by any party surrendering American Depositary Shares or to whom American Depositary Shares are issued (including, without limitation, issuance pursuant to a stock dividend or stock split declared by the Company or an exchange of stock regarding the American Depositary Shares or Deposited Securities or a delivery of American Depositary Shares pursuant to Section 4.3), or by Owners, as applicable: (1) taxes and other governmental charges, (2) such registration fees as may from time to time be in effect for the registration of transfers of Shares generally on the Share register of the Company or Foreign Registrar and applicable to transfers of Shares to or from the name of the Depositary or its nominee or the Custodian or its nominee on the making of deposits or withdrawals hereunder, (3) such cable, telex and facsimile transmission expenses as are expressly provided in this Deposit Agreement, (4) such expenses as are incurred by the Depositary in the conversion of foreign currency pursuant to Section 4.5, (5) a fee of \$5.00 or less per 100 American Depositary Shares (or portion thereof) for the delivery of American Depositary Shares pursuant to Section 2.3, 4.3 or 4.4 and the surrender of American Depositary Shares pursuant to Section 2.5 or 6.2, (6) a fee payable by Owners of \$.05 or less per American Depositary Share (or portion thereof) for any cash distribution made pursuant to this Deposit Agreement, including, but not limited to Sections 4.1 through 4.4 hereof, (7) a fee payable by Owners for the distribution of securities pursuant to Section 4.2, such fee being in an amount equal to the fee for the execution and delivery of American Depositary Shares referred to above which would have been charged as a result of the deposit of such securities (for purposes of this clause 7 treating all such securities as if they were Shares) but which securities are instead distributed by the Depositary to Owners, (8) in addition to any fee charged under clause 6, a fee of \$.05 or less per American Depositary Share (or portion thereof) per annum for depositary services, which will be payable as provided in clause 9 below, and (9) any other charges payable by the Depositary, any of the Depositary's agents, including the Custodian, or the agents of the Depositary's agents in connection with the servicing of Shares or other Deposited Securities (which charges shall be assessed against Owners as of the date or dates set by the Depositary in accordance with Section 4.6 and shall be payable at the sole discretion of the Depositary by billing such Owners for such charges

or by deducting such charges from one or more cash dividends or other cash distributions).

The Depositary may collect any of its fees by deduction from any cash distribution payable to Owners that are obligated to pay those fees.

The Depositary, subject to Section 2.9 hereof, may own and deal in any class of securities of the Company and its affiliates and in American Depositary Shares.

# SECTION 5.10 Retention of Depositary Documents.

The Depositary is authorized to destroy those documents, records, bills and other data compiled during the term of this Deposit Agreement at the times permitted by the laws or regulations governing the Depositary unless the Company requests that such papers be retained for a longer period or turned over to the Company or to a successor depositary.

### SECTION 5.11 Exclusivity.

The Company agrees not to appoint any other depositary for issuance of American or global depositary shares or receipts so long as The Bank of New York Mellon is acting as Depositary hereunder.

### SECTION 5.12 List of Restricted Securities Owners.

From time to time, the Company shall provide to the Depositary a list setting forth, to the actual knowledge of the Company, those persons or entities who beneficially own Restricted Securities and the Company shall update that list on a regular basis. The Company agrees to advise in writing each of the persons or entities so listed that such Restricted Securities are ineligible for deposit hereunder. The Depositary may rely on such a list or update but shall not be liable for any action or omission made in reliance thereon. Notwithstanding any provision herein to the contrary, the Depositary may, in its discretion, at the request and expense of the Company, and subject to such terms, conditions and limitations as the Depositary may require, agree to establish procedures to permit the deposit hereunder of Shares that are Restricted Securities in order to enable the holder of such Shares to hold its ownership interests in such Restricted Securities in the form of American Depositary Shares issued under the terms of this Deposit Agreement.

### SECTION 5.13 Registration of Shares; Share Register.

The Company agrees to maintain itself or engage, subject to shareholder approval, a third party (a "Transfer Agent") reasonably acceptable to the Depositary to maintain a Share Register for the Shares for so long as any American Depositary Shares or Receipts remain outstanding hereunder or this Agreement remains in force. The

29

Company agrees that it shall, or if the Share Register is maintained by a Transfer Agent, cooperate with the Depositary to ensure that such Transfer Agent shall, at any time and from time to time: (a) take any and all action as may be necessary to assure the accuracy and completeness of all information set forth in the Share Register in respect of the Shares; (b) provide to the Depositary, the Custodian or their respective agents unrestricted access to such part of the Share Register, which relates to the Shares, during regular business hours in accordance with Danish law, in such manner and upon such terms and conditions as the Depositary may, in its sole reasonable discretion, deem appropriate, to permit the Depositary, the Custodian or their respective agents to confirm the number of Shares registered in the name of the Depositary, the Custodian or their respective nominees, as applicable, pursuant to the terms of this Deposit Agreement and, in connection therewith, to provide the Depositary, the Custodian or their respective agents, upon request, with a duplicative extract from the relevant part of the Share Register duly certified by the Company or the Transfer Agent, as applicable, (or other independent third party reasonably acceptable to the Depositary); (c) promptly effect the re-registration of ownership of Shares deposited pursuant to Section 2.2 in the Share Register in connection with any deposit or withdrawal of Shares under this Deposit Agreement; (d) permit the Depositary or the Custodian to register any Shares held hereunder in the name of the Depositary, the Custodian or their respective nominees; and (e) to the extent permissible under applicable law promptly notify the Depositary in writing at any time that (A) the Company or the Transfer Agent, as applicable, eliminates the name of a shareholder of the Company from the Share Register or otherwise alters a shareholder's interest in the Shares and such shareholder alleges to the Company or Transfer Agent, as applicable, or publicly that such elimination or alteration is unlawful; (B) the Company no longer will be able materially to comply with, or has engaged in conduct that indicates it will not materially comply with, the provisions of this Section 5.13 relating to it (C) the Company or the Transfer Agent, as applicable, refuses to re-register Shares in the name of a particular purchaser and such purchaser (or its respective seller) alleges that such refusal is unlawful; (D) the Company or the Transfer Agent, as applicable, holds Shares for the account of the Company; or (E) the Company has materially breached the provisions of this Section 5.13 relating to it and has failed to cure such breach within a reasonable time.

The Depositary agrees that it will instruct the Custodian to maintain custody of all duplicative Share Register extracts (or other evidence of verification) provided to the Depositary, the Custodian or their respective agents pursuant to Section 5.13. In the event of any material discrepancy between the records of the Depositary or the Custodian and the Share Register, then, if an officer of the ADR Department of the Depositary has actual knowledge of such discrepancy, the Depositary shall promptly notify the Company. In the event of any discrepancy between the records of the Depositary or the Custodian and the Share Register, the Company agrees that (whether or not it has received any notification from the Depositary) it will (i) use, or if the Share Register is maintained by a Transfer Agent, cooperate with the Depositary to ensure that the Transfer Agent will use its reasonable efforts to cause the Company to reconcile its

records to the records of the Depositary or the Custodian and to make such corrections or revisions in the Share Register as may be necessary in connection therewith, and (ii) to the extent the Company, or the Transfer Agent, as applicable, is unable to so reconcile such records, promptly instruct the Depositary to notify the Owners of the existence of such discrepancy. Upon receipt of such instruction, the Depositary shall promptly give such notification to the Owners pursuant to Section 4.9 (it being understood that the Depositary may at any time give such notification to the Owners, whether or not it has received instructions from the Company), and the Depositary shall promptly cease issuing American Depositary Shares pursuant to Section 2.2 until such time as, in the opinion of the Depositary, such records have been appropriately reconciled.

### ARTICLE 6. AMENDMENT AND TERMINATION

SECTION 6.1 Amendment.

The form of the Receipts and any provisions of this Deposit Agreement may at any time and from time to time be amended by agreement between the Company and the Depositary without the consent of Owners or Holders in any respect which they may deem necessary or desirable. Any amendment which shall impose or increase any fees or charges (other than taxes and other governmental charges, registration fees, cable, telex or facsimile transmission costs, delivery costs or other such expenses), or which shall otherwise prejudice any substantial existing right of Owners, shall, however, not become effective as to outstanding American Depositary Shares until the expiration of thirty days after notice of such amendment shall have been given to the Owners of outstanding American Depositary Shares. Every Owner and Holder, at the time any amendment so becomes effective, shall be deemed, by continuing to hold such American Depositary Shares or any interest therein, to consent and agree to such amendment and to be bound by the Deposit Agreement as amended thereby. In no event shall any amendment impair the right of the Owner to surrender American Depositary Shares and receive therefor the Deposited Securities represented thereby, except in order to comply with mandatory provisions of applicable law.

SECTION 6.2 Termination.

The Company may at any time terminate this Deposit Agreement by instructing the Depositary to mail a notice of termination to the Owners of all American Depositary Shares then outstanding at least 30 days prior to the termination date included in such notice. The Depositary may likewise terminate this Deposit Agreement if at any time 60 days shall have expired after the Depositary delivered to the Company a written resignation notice and if a successor depositary shall not have been appointed and accepted its appointment as provided in Section 5.4; in such case the Depositary shall mail a notice of termination to the Owners of all American Depositary Shares then outstanding at least 30 days prior to the termination date. On and after the date of termination, the Owner of American Depositary Shares will, upon (a) surrender of such American Depositary Shares, (b) payment of the fee of the Depositary for the surrender

31

of American Depositary Shares referred to in Section 2.5, and (c) payment of any applicable taxes or governmental charges, be entitled to delivery, to him or upon his order, of the amount of Deposited Securities represented by those American Depositary Shares. If any American Depositary Shares shall remain outstanding after the date of termination, the Depositary thereafter shall discontinue the registration of transfers of American Depositary Shares, shall suspend the distribution of dividends to the Owners thereof, and shall not give any further notices or perform any further acts under this Deposit Agreement, except that the Depositary shall continue to collect dividends and other distributions pertaining to Deposited Securities, shall sell rights and other property as provided in this Deposit Agreement, and shall continue to deliver Deposited Securities, together with any dividends or other distributions received with respect thereto and the net proceeds of the sale of any rights or other property, upon surrender of American Depositary Shares (after deducting, in each case, the fee of the Depositary for the surrender of American Depositary Shares, any expenses for the account of the Owner of such American Depositary Shares in accordance with the terms and conditions of this Deposit Agreement, and any applicable taxes or governmental charges).

At any time after the expiration of four months from the date of termination, the Depositary may sell the Deposited Securities then held under this Deposit Agreement and may thereafter hold uninvested the net proceeds of any such sale, together with any other cash then held by it hereunder, unsegregated and without liability for interest, for the pro rata benefit of the Owners of American Depositary Shares that have not theretofore been surrendered, such Owners thereupon becoming general creditors of the Depositary with respect to such net proceeds. After making such sale, the Depositary shall be discharged from all obligations under this Deposit Agreement, except to account for such net proceeds and other cash (after deducting, in each case, the fee of the Depositary for the surrender of American Depositary Shares, any expenses for the account of the Owner of such American Depositary Shares in accordance with the terms and conditions of this Deposit Agreement, and any applicable taxes or governmental charges. Upon the termination of this Deposit Agreement, the Company shall be discharged from all obligations under this Deposit Agreement except for its obligations to the Depositary under Sections 5.8 and 5.9.

### ARTICLE 7. MISCELLANEOUS

SECTION 7.1 Counterparts; Signatures

This Deposit Agreement may be executed in any number of counterparts, each of which shall be deemed an original and all of such counterparts shall constitute one and the same instrument. Copies of this Deposit Agreement shall be filed with the Depositary and the Custodians and shall be open to inspection by any Owner or Holder during business hours.

Any manual signature on this Deposit Agreement that is faxed, scanned or photocopied, and any electronic signature valid under the Electronic Signatures in Global

32

and National Commerce Act, 15 U.S.C. § 7001, *et. seq.*, shall for all purposes have the same validity, legal effect and admissibility in evidence as an original manual signature, and the parties hereby waive any objection to the contrary.

SECTION 7.2 No Third Party Beneficiaries.

This Deposit Agreement is for the exclusive benefit of the parties hereto and shall not be deemed to give any legal or equitable right, remedy or claim whatsoever to any other person.

### SECTION 7.3 Severability.

In case any one or more of the provisions contained in this Deposit Agreement or in the Receipts should be or become invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions contained herein or therein shall in no way be affected, prejudiced or disturbed thereby.

SECTION 7.4 Owners and Holders as Parties; Binding Effect.

The Owners and Holders from time to time shall be parties to this Deposit Agreement and shall be bound by all of the terms and conditions hereof and of the Receipts by acceptance of American Depositary Shares or any interest therein.

SECTION 7.5 Notices.

Any and all notices to be given to the Company shall be deemed to have been duly given if personally delivered or sent by mail or cable, telex, facsimile transmission or email confirmed by letter, addressed to Forward Pharma A/S, Østergade 24A, 1, Copenhagen K, DK-1100, Attention: Anders R. Therkelsen with a copy to Peder Møller Andersen or any other place to which the Company may have transferred its principal office with notice to the Depositary.

Any and all notices to be given to the Depositary shall be deemed to have been duly given if in English and personally delivered or sent by mail or cable, telex, facsimile transmission confirmed by letter, addressed to The Bank of New York Mellon, 101 Barclay Street, New York, New York 10286, Attention: American Depositary Receipt Administration, or any other place to which the Depositary may have transferred its Corporate Trust Office with notice to the Company.

Any and all notices to be given to any Owner shall be deemed to have been duly given if personally delivered or sent by mail or cable, telex, facsimile transmission or email confirmed by letter, addressed to such Owner at the address of such Owner as it appears on the transfer books for American Depositary Shares of the Depositary, or, if such Owner shall have filed with the Depositary a written request that

33

notices intended for such Owner be mailed to some other address, at the address designated in such request.

Delivery of a notice sent by mail or cable, telex, facsimile transmission or email shall be deemed to be effected at the time when a duly addressed letter containing the same (or a confirmation thereof in the case of a cable, telex, facsimile transmission or email) is deposited, postage prepaid, in a post-office letter box. The Depositary or the Company may, however, act upon any cable, telex, facsimile transmission or if applicable, email received by it, notwithstanding that such cable, telex, facsimile transmission or email shall not subsequently be confirmed by letter as aforesaid.

SECTION 7.6 Arbitration; Settlement of Disputes.

- (a) Any controversy, claim or cause of action brought by any party hereto against the Company arising out of or relating to the Shares or other Deposited Securities, the American Depositary Shares, the Receipts or this Deposit Agreement, or the breach hereof or thereof, shall be settled by arbitration in accordance with the International Arbitration Rules of the American Arbitration Association, and judgment upon the award rendered by the arbitrators may be entered in any court having jurisdiction thereof; provided, however, that in the event of any third-party litigation to which the Depositary is a party and to which the Company may properly be joined, the Company may be so joined in any court in which such litigation is proceeding; and provided, further, that any such controversy, claim or cause of action brought by a party hereto against the Company relating to or based upon the provisions of the Federal securities laws of the United States or the rules and regulations promulgated thereunder shall be submitted to arbitration as provided in this Section 7.06 if, but only if, so elected by the claimant.
- (b) The place of the arbitration shall be The City of New York, State of New York, United States of America, and the language of the arbitration shall be English.
- (c) The number of arbitrators shall be three, each of whom shall be disinterested in the dispute or controversy, shall have no connection with any party thereto, and shall be an attorney experienced in international securities transactions. Each party shall appoint one arbitrator and the two arbitrators shall select a third arbitrator who shall serve as chairperson of the tribunal. If a dispute, controversy or cause of action shall involve more than two parties, the parties shall attempt to align themselves in two sides (i.e., claimant(s) and respondent(s)), each of which shall appoint one arbitrator as if there were only two parties to such dispute, controversy or cause of action. If such alignment and appointment shall not have occurred within thirty (30) calendar days after the initiating party serves the arbitration demand, the American Arbitration Association shall appoint the three arbitrators, each of whom shall have the qualifications described above. The parties and the American Arbitration Association may appoint from among the nationals of any country, whether or not a party is a national of that country.

34

- (d) The arbitral tribunal shall have no authority to award any consequential, special or punitive damages or other damages not measured by the prevailing party's actual damages and may not, in any event, make any ruling, finding or award that does not conform to the terms and conditions of this Deposit Agreement.
- (e) Any controversy, claim or cause of action arising out of or relating to the Shares or other Deposited Securities, the American Depositary Shares, the Receipts or this Deposit Agreement not subject to arbitration under this Section 7.6 shall be litigated in the Federal and state courts in the Borough of Manhattan, The City of New York and the Company hereby submits to the personal jurisdiction of the court in which such action or proceeding is brought.

The Company hereby (i) irrevocably designates and appoints CT Corporation System, 1015 15th Street, NW, Suite 1000, Washington, DC 20005, as the Company's authorized agent upon which process may be served in any suit or proceeding (including any arbitration proceeding) arising out of or relating to the Shares or Deposited Securities, the American Depositary Shares, the Receipts or this Deposit Agreement, (ii) consents and submits to the jurisdiction of any state or federal court in the State of New York in which any such suit or proceeding may be instituted, and (iii) agrees that service of process upon said authorized agent shall be deemed in every respect effective service of process upon the Company in any such suit or proceeding. The Company agrees to deliver, upon the execution and delivery of this Deposit Agreement, a written acceptance by such agent of its appointment as such agent. The Company further agrees to take any and all action, including the filing of any and all such documents and instruments, as may be necessary to continue such designation and appointment in full force and effect for so long as any American Depositary Shares or Receipts remain outstanding or this Deposit Agreement remains in force. In the event the Company fails to continue such designation and appointment in full force and effect, the Company hereby waives personal service of process upon it and consents that any such service of process may be made by certified or registered mail, return receipt requested, directed to the Company at its address last specified for notices hereunder, and service so made shall be deemed completed five (5) business days after the same shall have been so mailed.

EACH PARTY TO THIS DEPOSIT AGREEMENT (INCLUDING, FOR AVOIDANCE OF DOUBT, EACH OWNER AND HOLDER) HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY SUIT, ACTION OR PROCEEDING AGAINST THE COMPANY AND/OR THE DEPOSITARY DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THE SHARES OR OTHER DEPOSITED SECURITIES, THE AMERICAN DEPOSITARY SHARES OR THE RECEIPTS, THIS DEPOSIT AGREEMENT OR ANY TRANSACTION CONTEMPLATED HEREIN OR THEREIN, OR THE

35

BREACH HEREOF OR THEREOF, INCLUDING WITHOUT LIMITATION ANY QUESTION REGARDING EXISTENCE, VALIDITY OR TERMINATION (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY).

SECTION 7.8 Waiver of Immunities.

To the extent that the Company or any of its properties, assets or revenues may have or may hereafter become entitled to, or have attributed to it, any right of immunity, on the grounds of sovereignty or otherwise, from any legal action, suit or proceeding, from the giving of any relief in any respect thereof, from setoff or counterclaim, from the jurisdiction of any court, from service of process, from attachment upon or prior to judgment, from attachment in aid of execution or judgment, or from execution of judgment, or other legal process or proceeding for the giving of any relief or for the enforcement of any judgment, in any jurisdiction in which proceedings may at any time be commenced, with respect to its obligations, liabilities or any other matter under or arising out of or in connection with the Shares or Deposited Securities, the American Depositary Shares, the Receipts or this Deposit Agreement, the Company, to the fullest extent permitted by law, hereby irrevocably and unconditionally waives, and agrees not to plead or claim, any such immunity and consents to such relief and enforcement.

SECTION 7.9 Governing Law.

This Deposit Agreement and the Receipts shall be interpreted and all rights hereunder and thereunder and provisions hereof and thereof shall be governed by the laws of the State of New York, except with respect to its authorization and execution by the Company, which shall be governed by the laws of Denmark.

36

IN WITNESS WHEREOF, FORWARD PHARMA A/S and THE BANK OF NEW YORK MELLON have duly executed this Deposit Agreement as of the day and year first set forth above and all Owners and Holders shall become parties hereto upon acceptance by them of American Depositary Shares or any interest therein.

# FORWARD PHARMA A/S

By:
Name:
Title:

THE BANK OF NEW YORK MELLON,
as Depositary

By:
Name:
Title:

(Each American Depositary Share represents One (1) deposited Share)

# THE BANK OF NEW YORK MELLON AMERICAN DEPOSITARY RECEIPT FOR [ORDINARY] SHARES, OF FORWARD PHARMA A/S (INCORPORATED UNDER THE LAWS OF DENMARK)

The Bank of New York Mellon, as depositary (hereinafter called the "Depositary"), hereby certifies that , or registered assigns IS THE OWNER OF

# AMERICAN DEPOSITARY SHARES

representing deposited [ordinary] shares (herein called "Shares") of Forward Pharma A/S, incorporated under the laws of Denmark (herein called the "Company"). At the date hereof, each American Depositary Share represents one Share deposited or subject to deposit under the Deposit Agreement (as such term is hereinafter defined) at the office of (herein called the "Custodian"). The Depositary's Corporate Trust Office is located at a different address than its principal executive office. Its Corporate Trust Office is located at 101 Barclay Street, New York, N.Y. 10286, and its principal executive office is located at One Wall Street, New York, N.Y. 10286.

THE DEPOSITARY'S CORPORATE TRUST OFFICE ADDRESS IS 101 BARCLAY STREET, NEW YORK, N.Y. 10286

### THE DEPOSIT AGREEMENT.

This American Depositary Receipt is one of an issue (herein called "Receipts"), all issued and to be issued upon the terms and conditions set forth in the deposit agreement, dated as of , 2014 (herein called the "Deposit Agreement"), by and among the Company, the Depositary, and all Owners and Holders from time to time of American Depositary Shares issued thereunder, each of whom by accepting American Depositary Shares agrees to become a party thereto and become bound by all the terms and conditions thereof. The Deposit Agreement sets forth the rights of Owners and Holders and the rights and duties of the Depositary in respect of the Shares deposited thereunder and any and all other securities, property and cash from time to time received in respect of such Shares and held thereunder (such Shares, securities, property, and cash are herein called "Deposited Securities"). Copies of the Deposit Agreement are on file at the Depositary's Corporate Trust Office in New York City and at the office of the Custodian.

The statements made on the face and reverse of this Receipt are summaries of certain provisions of the Deposit Agreement and are qualified by and subject to the detailed provisions of the Deposit Agreement, to which reference is hereby made. Capitalized terms defined in the Deposit Agreement and not defined herein shall have the meanings set forth in the Deposit Agreement.

# 2. SURRENDER OF RECEIPTS AND WITHDRAWAL OF SHARES.

Upon surrender at the Corporate Trust Office of the Depositary of American Depositary Shares, and upon payment of the fee of the Depositary provided in this Receipt, and subject to the terms and conditions of the Deposit Agreement, the Owner of those American Depositary Shares is entitled to delivery, to him or as instructed, of the amount of Deposited Securities at the time represented by those American Depositary Shares. Delivery of such Deposited Securities may be made by the delivery of (a) certificates or account transfer in the name of the Owner hereof or as ordered by him, with proper endorsement or accompanied by proper instruments or instructions of transfer and (b) any other securities, property and cash to which such Owner is then entitled in respect of this Receipt. Such delivery will be made at the option of the Owner hereof, either at the office of the Custodian or at the Corporate Trust Office of the Depositary, provided that the forwarding of certificates for Shares or other Deposited Securities for such delivery at the Corporate Trust Office of the Depositary shall be at the risk and expense of the Owner hereof.

# 3. TRANSFERS, SPLIT-UPS, AND COMBINATIONS OF RECEIPTS.

Transfers of American Depositary Shares may be registered on the books of the Depositary by the Owner in person or by a duly authorized attorney, upon surrender of those American Depositary Shares properly endorsed for transfer or accompanied by proper instruments of transfer, in the case of a Receipt, or pursuant to a proper instruction

(including, for the avoidance of doubt, instructions through DRS and Profile as provided in Section 2.10 of the Deposit Agreement), in the case of uncertificated American Depositary Shares, and funds sufficient to pay any applicable transfer taxes and the expenses of the Depositary and upon compliance with such regulations, if any, as the Depositary may establish for such purpose. This Receipt may be split into other such Receipts, or may be combined with other such Receipts into one Receipt, evidencing the same aggregate number of American Depositary Shares as the Receipt or Receipts surrendered. The Depositary, upon surrender of certificated American Depositary Shares for the purpose of exchanging for uncertificated American Depositary Shares, shall cancel those certificated American Depositary Shares and send the Owner a statement confirming that the Owner is the Owner of uncertificated American Depositary Shares. The Depositary, upon receipt of a proper instruction (including, for the avoidance of doubt, instructions through DRS and Profile as provided in Section 2.10 of the Deposit Agreement) from the Owner of uncertificated American Depositary Shares for the purpose of exchanging for certificated American Depositary Shares, shall cancel those uncertificated American Depositary Shares and deliver to the Owner the same number of certificated American Depositary Shares. As a condition precedent to the delivery, registration of transfer, or surrender of any American Depositary Shares or split-up or combination of any Receipt or withdrawal of any Deposited Securities, the Depositary, the Custodian, or Registrar may require payment from the depositor of the Shares or the presenter of the Receipt or instruction for registration of transfer or surrender of American Depositary Shares not evidenced by a Receipt of a sum sufficient to reimburse it for any tax or other governmental charge and any stock transfer or registration fee with respect thereto (including any such tax or charge and fee with respect to Shares being deposited or withdrawn) and payment of any applicable fees as provided in the Deposit Agreement, may require the production of proof satisfactory to it as to the identity and genuineness of any signature and may also require compliance with any regulations the Depositary may establish consistent with the provisions of the Deposit Agreement.

The delivery of American Depositary Shares against deposit of Shares generally or against deposit of particular Shares may be suspended, or the transfer of American Depositary Shares in particular instances may be refused, or the registration of transfer of outstanding American Depositary Shares generally may be suspended, during any period when the transfer books of the Depositary are closed, or if any such action is deemed necessary or advisable by the Depositary or the Company at any time or from time to time because of any requirement of law or of any government or governmental body or commission, or under any provision of the Deposit Agreement, or for any other reason, subject to the provisions of the following sentence. Notwithstanding anything to the contrary in the Deposit Agreement or this Receipt, the surrender of outstanding American Depositary Shares and withdrawal of Deposited Securities may not be suspended subject only to (i) temporary delays caused by closing the transfer books of the Depositary or the Company or the Foreign Registrar, if applicable, or the deposit of Shares in connection with voting at a shareholders' meeting, or the payment of dividends, (ii) the payment of fees, taxes and similar charges, and (iii) compliance with any U.S. or foreign laws or

A-2

governmental regulations relating to the American Depositary Shares or to the withdrawal of the Deposited Securities. Without limitation of the foregoing, the Depositary shall not knowingly accept for deposit under the Deposit Agreement any Shares which would be required to be registered under the provisions of the Securities Act of 1933, unless a registration statement is in effect as to such Shares for such offer and sale or such Shares are exempt from registration thereunder.

# 4. <u>LIABILITY OF OWNER FOR TAXES.</u>

If any tax or other governmental charge imposed by applicable law shall become payable with respect to any American Depositary Shares or any Deposited Securities represented by any American Depositary Shares, such tax or other governmental charge shall be payable by the Owner to the Depositary. The Depositary may refuse to register any transfer of those American Depositary Shares or any withdrawal of Deposited Securities represented by those American Depositary Shares until such payment is made, and may withhold any dividends or other distributions, or may sell for the account of the Owner any part or all of the Deposited Securities represented by those American Depositary Shares, and may apply such dividends or other distributions or the proceeds of any such sale in payment of such tax or other governmental charge and the Owner shall remain liable for any deficiency.

### 5. WARRANTIES ON DEPOSIT OF SHARES.

Every person depositing Shares under the Deposit Agreement shall be deemed thereby to represent and warrant, that such Shares and each certificate therefor, if applicable, are validly issued, fully paid, nonassessable and were not issued in violation of of any preemptive rights of the holders of outstanding Shares and that the person making such deposit is duly authorized so to do. Every such person shall also be deemed to represent that the deposit of such Shares and the sale of American Depositary Shares representing such Shares by that person are not restricted under the Securities Act of 1933. Such representations and warranties shall survive the deposit of Shares and delivery of American Depositary Shares.

### 6. <u>FILING PROOFS, CERTIFICATES, AND OTHER INFORMATION.</u>

Any person presenting Shares for deposit or any Owner or Holder may be required from time to file with the Depositary or the Custodian such proof of citizenship or residence, exchange control approval, or such information relating to the registration on the books of the Company or the Foreign Registrar, if applicable, to execute such certificates and to make such representations and warranties, as the Depositary may deem necessary or proper. The Depositary may withhold the delivery or registration of transfer of any American Depositary Shares or the distribution of any dividend or sale or distribution of rights or of the proceeds thereof or the delivery of any Deposited Securities until such proof or other information is filed or such certificates are executed or such representations and warranties made. If requested in writing, the

A-3

Depositary shall, as promptly as practicable, provide the Company, at the expense of the Company, with copies of any such proofs, certificates or other information it receives pursuant to Section 3.1 of the Deposit Agreement, to the extent that disclosure is permitted under applicable law. No Share shall be accepted for deposit unless accompanied by evidence satisfactory to the Depositary that any necessary approval has been granted by any governmental body in Denmark that is then performing the function of the regulation of currency exchange.

# 7. <u>CHARGES OF DEPOSITARY.</u>

The following charges shall be incurred by any party depositing or withdrawing Shares or by any party surrendering American Depositary Shares or to whom American Depositary Shares are issued (including, without limitation, issuance pursuant to a stock dividend or stock split declared by the Company or an exchange of stock regarding the American Depositary Shares or Deposited Securities or a delivery of American Depositary Shares pursuant to Section 4.3 of the Deposit Agreement), or by Owners, as applicable: (1) taxes and other governmental charges, (2) such registration fees as may from time to time be in effect for the registration of transfers of Shares generally on the Share register of the Company or Foreign Registrar and applicable to transfers of Shares to or from the name of the Depositary or its nominee or the Custodian or its nominee on the making of deposits or withdrawals under the terms of the Deposit Agreement, (3) such cable, telex and facsimile transmission expenses as are expressly provided in the Deposit Agreement, (4) such expenses as are incurred by the Depositary in the conversion of foreign currency pursuant to Section 4.5 of the Deposit Agreement, (5) a fee of \$5.00 or less per 100 American Depositary Shares (or portion thereof) for the delivery of American Depositary Shares pursuant to Section 2.3, 4.3 or 4.4 of the Deposit Agreement and the surrender of American Depositary Shares pursuant to Section 2.5 or 6.2 of the Deposit Agreement, (6) a fee payable by Owners of \$.05 or less per American Depositary Share (or portion thereof) for any cash distribution made pursuant to the Deposit Agreement, including, but not limited to Sections 4.1 through 4.4 of the Deposit Agreement, (7) a fee payable by Owners for the distribution of securities pursuant to Section 4.2 of the Deposit Agreement, such fee being in an amount equal to the fee for the execution and delivery of American Depositary Shares referred to above which would have been charged as a result of the deposit of such securities (for purposes of this clause 7 treating all such securities as if they were Shares) but which securities are instead distributed by the Depositary to Owners, (8) in addition to any fee charged under clause 6, a fee of \$.05 or less per American Depositary Share (or portion thereof) per annum for depositary services, which will be payable as provided in clause 9 below, and (9) any other charges payable by the Depositary, any of the Depositary's agents, including the Custodian, or the agents of the Depositary's agents in connection with the servicing of Shares or other Deposited Securities (which charges shall be assessed against Owners as of the date or dates set by the Depositary in accordance with Section 4.6 of the Deposit Agreement and shall be payable at the sole discretion of the Depositary by billing such

Owners for such charges or by deducting such charges from one or more cash dividends or other cash distributions).

The Depositary may collect any of its fees by deduction from any cash distribution payable to Owners that are obligated to pay those fees.

The Depositary, subject to Article 8 hereof, may own and deal in any class of securities of the Company and its affiliates and in American Depositary Shares.

From time to time, the Depositary may make payments to the Company to reimburse and / or share revenue from the fees collected from Holders, or waive fees and expenses for services provided, generally relating to costs and expenses arising out of establishment and maintenance of the American Depositary Shares program. In performing its duties under the Deposit Agreement, the Depositary may use brokers, dealers or other service providers that are affiliates of the Depositary and that may earn or share fees and commissions.

### 8. PRE-RELEASE OF RECEIPTS.

Unless requested in writing by the Company to cease doing so, notwithstanding Section 2.3 of the Deposit Agreement, the Depositary may deliver American Depositary Shares prior to the receipt of Shares pursuant to Section 2.2 of the Deposit Agreement (a "Pre-Release"). The Depositary may, pursuant to Section 2.5 of the Deposit Agreement, deliver Shares upon the surrender of American Depositary Shares that have been Pre-Released, whether or not such cancellation is prior to the termination of such Pre-Release. The Depositary may receive American Depositary Shares in lieu of Shares in satisfaction of a Pre-Release. Each Pre-Release will be (a) preceded or accompanied by a written representation from the person to whom American Depositary Shares or Shares are to be delivered, that such person, or its customer, (i) beneficially owns the Shares or American Depositary Shares to be remitted, as the case may be, (ii) assigns all beneficial right, title and interest in such American Depositary Shares or Shares, as the case may be, to the Depositary in its capacity as such and for the benefit of the Owners and (iii) will not take any action with respect to such American Depositary Shares or Shares, as the case may be, that is inconsistent with the transfer of beneficial ownership (including, without the consent of the Depositary, disposing of such American Depositary Shares or Shares, as the case may be), other than in satisfaction of the Pre-Release, (b) at all times fully collateralized with cash or such other collateral as the Depositary deems appropriate, (c) terminable by the Depositary on not more than five (5) business days' notice, and (d) subject to such further indemnities and credit regulations as the Depositary deems appropriate. The number of Shares represented by American Depositary Shares which are outstanding at any time as a result of Pre-Release will not normally exceed thirty percent (30%) of the Shares deposited hereunder; provided, however, that the Depositary reserves the right to disregard such limi

A-5

respect to Pre-Release transactions with any particular Pre-Releasee on a case-by-case basis as the Depositary deems appropriate. The collateral referred to in item (b) above shall be held by the Depositary as security for the performance of the Pre-Releasee's obligations in connection the related Pre-Release transaction, including the Pre-Releasee's obligation to deliver Shares or American Depositary Shares upon termination of that Pre-Release transaction (and shall not, for the avoidance of doubt, constitute Deposited Securities).

The Depositary may retain for its own account any compensation received by it in connection with the foregoing.

### 9. <u>TITLE TO RECEIPTS.</u>

It is a condition of this Receipt and every successive Owner and Holder of this Receipt by accepting or holding the same consents and agrees that when properly endorsed or accompanied by proper instruments of transfer, shall be transferable as certificated registered securities under the laws of the State of New York. American Depositary Shares not evidenced by Receipts shall be transferable as uncertificated registered securities under the laws of the State of New York. The Depositary, notwithstanding any notice to the contrary, may treat the Owner of American Depositary Shares as the absolute owner thereof for the purpose of determining the person entitled to distribution of dividends or other distributions or to any notice provided for in the Deposit Agreement and for all other purposes, and neither the Depositary nor the Company shall have any obligation or be subject to any liability under the Deposit Agreement to any Holder of American Depositary Shares unless that Holder is the Owner of those American Depositary Shares.

# 10. VALIDITY OF RECEIPT.

This Receipt shall not be entitled to any benefits under the Deposit Agreement or be valid or obligatory for any purpose, unless this Receipt shall have been (i) executed by the Depositary by the manual signature of a duly authorized officer of the Depositary or (ii) executed by the facsimile signature of a duly authorized officer of the Depositary or a Registrar.

# 11. REPORTS; INSPECTION OF TRANSFER BOOKS.

The Company is subject to the periodic reporting requirements of the Securities Exchange Act of 1934 and, accordingly, files certain reports with the Securities and Exchange Commission. Those reports will be available for inspection and copying through the Commission's EDGAR system on the Internet at www.sec.gov or at public reference facilities maintained by the Commission located at 100 F Street N.E. in Washington, D.C 20549.

A-6

The Depositary will make available for inspection by Owners at its Corporate Trust Office any reports, notices and other communications, including any proxy soliciting material, received from the Company which are both (a) received by the Depositary as the holder of the Deposited Securities and (b) made generally available to the holders of such Deposited Securities by the Company. The Depositary will also, upon written request by the Company, send to Owners copies of such reports when furnished by the Company pursuant to the Deposit Agreement. Any such reports and communications, including

any such proxy soliciting material, furnished to the Depositary by the Company shall be furnished in English to the extent such materials are required to be translated into English pursuant to any regulations of the Commission.

The Depositary will keep books, at its Corporate Trust Office, for the registration of American Depositary Shares and transfers of American Depositary Shares which at all reasonable times shall be open for inspection by the Owners, provided that such inspection shall not be for the purpose of communicating with Owners in the interest of a business or object other than the business of the Company or a matter related to the Deposit Agreement or the American Depositary Shares.

# 12. DIVIDENDS AND DISTRIBUTIONS.

Whenever the Depositary receives any cash dividend or other cash distribution on any Deposited Securities, the Depositary will, if at the time of receipt thereof any amounts received in a foreign currency can in the judgment of the Depositary be converted on a reasonable basis into United States dollars transferable to the United States, and subject to the Deposit Agreement, as promptly as possible, convert such dividend or distribution into dollars and will distribute the amount thus received (net of the fees and expenses of the Depositary as provided in Article 7 hereof and Section 5.9 of the Deposit Agreement) to the Owners entitled thereto; <u>provided</u>, <u>however</u>, that in the event that the Company or the Depositary is required by applicable law to withhold and does withhold from any cash dividend or other cash distribution in respect of any Deposited Securities an amount on account of taxes or other governmental charges, the amount distributed to the Owners of the American Depositary Shares representing such Deposited Securities shall be reduced accordingly.

Subject to the provisions of Sections 4.11 and 5.9 of the Deposit Agreement, whenever the Depositary receives any distribution other than a distribution described in Section 4.1, 4.3 or 4.4 of the Deposit Agreement, the Depositary will cause the securities or property received by it to be distributed to the Owners entitled thereto, in any manner that the Depositary may deem equitable and practicable for accomplishing such distribution; provided, however, that if in the opinion of the Depositary such distribution cannot be made proportionately among the Owners of Receipts entitled thereto, or if for any other reason the Depositary deems such distribution not to be feasible, the Depositary may adopt such method as it may deem equitable and practicable for the purpose of effecting such distribution, including, but not limited to, the public or private sale of the

A-7

securities or property thus received, or any part thereof, and the net proceeds of any such sale (net of the fees and expenses of the Depositary as provided in Article 7 hereof and Section 5.9 of the Deposit Agreement) will be distributed by the Depositary to the Owners of Receipts entitled thereto all in the manner and subject to the conditions described in Section 4.1 of the Deposit Agreement. The Depositary may withhold any distribution of securities under Section 4.2 of the Deposit Agreement if it has not received satisfactory assurances from the Company that the distribution does not require registration under the Securities Act of 1933. The Depositary may sell, by public or private sale, an amount of securities or other property it would otherwise distribute under this Article that is sufficient to pay its fees and expenses in respect of that distribution.

If any distribution consists of a dividend in, or free distribution of, Shares, the Depositary may, and shall if the Company shall so request in writing, deliver to the Owners entitled thereto, an aggregate number of American Depositary Shares representing the amount of Shares received as such dividend or free distribution, subject to the terms and conditions of the Deposit Agreement with respect to the deposit of Shares and after deduction or upon issuance of American Depositary Shares, including the withholding of any tax or other governmental charge as provided in Section 4.11 of the Deposit Agreement and the payment of the fees and expenses of the Depositary as provided in Article 7 hereof and Section 5.9 of the Deposit Agreement (and the Depositary may sell, by public or private sale, an amount of Shares received sufficient to pay its fees and expenses in respect of that distribution. In lieu of delivering fractional American Depositary Shares in any such case, the Depositary will sell the amount of Shares represented by the aggregate of such fractions and distribute the net proceeds, all in the manner and subject to the conditions described in Section 4.10f the Deposit Agreement. If additional American Depositary Shares are not so delivered, each American Depositary Share shall thenceforth also represent the additional Shares distributed upon the Deposited Securities represented thereby.

In the event that the Depositary determines that any distribution in property (including Shares and rights to subscribe therefor) is subject to any tax or other governmental charge which the Depositary is obligated to withhold, the Depositary may by public or private sale dispose of all or a portion of such property (including Shares and rights to subscribe therefor) in such amounts and in such manner as the Depositary deems necessary and practicable to pay any such taxes or charges, and the Depositary shall distribute the net proceeds of any such sale after deduction of such taxes or charges to the Owners of Receipts entitled thereto.

# 13. RIGHTS.

In the event that the Company shall offer or cause to be offered to the holders of any Deposited Securities any rights to subscribe for additional Shares or any rights of any other nature, the Depositary shall, after consultation with the Company, to the extent practicable, have discretion as to the procedure to be followed in making such rights

A-8

available to any Owners or in disposing of such rights on behalf of any Owners and making the net proceeds available to such Owners or, if by the terms of such rights offering or for any other reason, the Depositary may not either make such rights available to any Owners or dispose of such rights and make the net proceeds available to such Owners, then the Depositary shall allow the rights to lapse. If at the time of the offering of any rights the Depositary determines in its discretion that it is lawful and feasible to make such rights available to all or certain Owners but not to other Owners, the Depositary may distribute to any Owner to whom it determines the distribution to be lawful and feasible, in proportion to the number of American Depositary Shares held by such Owner, warrants or other instruments therefor in such form as it deems appropriate.

In circumstances in which rights would otherwise not be distributed, if an Owner requests the distribution of warrants or other instruments in order to exercise the rights allocable to the American Depositary Shares of such Owner under the Deposit Agreement, the Depositary will make such rights available to such Owner upon written notice from the Company to the Depositary that (a) the Company has elected in its sole discretion to permit such rights to be exercised and (b) such Owner has executed such documents as the Company has determined in its sole discretion are reasonably required under applicable law

If the Depositary has distributed warrants or other instruments for rights to all or certain Owners, then upon instruction from such an Owner pursuant to such warrants or other instruments to the Depositary from such Owner to exercise such rights, upon payment by such Owner to the Depositary for the account of such Owner of an amount equal to the purchase price of the Shares to be received upon the exercise of the rights, and upon payment of the fees and expenses of the Depositary and any other charges as set forth in such warrants or other instruments, the Depositary shall, on behalf of such Owner, exercise the rights and purchase the Shares, and the Company shall cause the Shares so purchased to be delivered to the Depositary on behalf of such Owner. As agent for such Owner, the Depositary will cause the Shares so purchased to be deposited pursuant to Section 2.2 of the Deposit Agreement, and shall, pursuant to Section 2.3 of the Deposit Agreement, deliver American Depositary Shares to such Owner. In the case of a distribution pursuant to the second paragraph of this Article 13, such deposit shall be made, and depositary shares shall be delivered, under depositary arrangements which provide for issuance of depositary shares subject to the appropriate restrictions on sale, deposit, cancellation, and transfer under applicable United States laws.

If the Depositary determines in its reasonable discretion that it is not lawful and feasible to make such rights available to all or certain Owners, it may sell the rights, warrants or other instruments in proportion to the number of American Depositary Shares held by the Owners to whom it has determined it may not lawfully or feasibly make such rights available, and allocate the net proceeds of such sales (net of the fees and expenses of the Depositary as provided in Section 5.9 of the Deposit Agreement and all taxes and governmental charges payable in connection with such rights and subject to the terms and

A-9

conditions of the Deposit Agreement) for the account of such Owners otherwise entitled to such rights, warrants or other instruments, upon an averaged or other practical basis without regard to any distinctions among such Owners because of exchange restrictions or the date of delivery of any American Depositary Shares or otherwise.

The Depositary will not offer rights to Owners unless both the rights and the securities to which such rights relate are either exempt from registration under the Securities Act of 1933 with respect to a distribution to all Owners or are registered under the provisions of such Act; provided, that nothing in the Deposit Agreement shall create any obligation on the part of the Company to file a registration statement with respect to such rights or underlying securities or to endeavor to have such a registration statement declared effective. If an Owner requests the distribution of warrants or other instruments, notwithstanding that there has been no such registration under the Securities Act of 1933, the Depositary shall not effect such distribution unless it has received an opinion from recognized counsel in the United States for the Company upon which the Depositary may rely that such distribution to such Owner is exempt from such registration, provided, however, that any opinion requested by an Owner to be delivered by the Company's counsel shall be prepared by the Company's counsel at the Owner's expense.

The Depositary shall not be responsible for any failure to determine that it may be lawful or feasible to make such rights available to Owners in general or any Owner in particular.

### 14. CONVERSION OF FOREIGN CURRENCY.

Whenever the Depositary or the Custodian shall receive foreign currency, by way of dividends or other distributions or the net proceeds from the sale of securities, property or rights, and if at the time of the receipt thereof the foreign currency so received can in the judgment of the Depositary be converted on a reasonable basis into Dollars and the resulting Dollars transferred to the United States, the Depositary shall, as promptly as practicable, convert or cause to be converted by sale or in any other manner that it may determine, such foreign currency into Dollars, and such Dollars shall be distributed to the Owners entitled thereto or, if the Depositary shall have distributed any warrants or other instruments which entitle the holders thereof to such Dollars, then to the holders of such warrants and/or instruments upon surrender thereof for cancellation. Such distribution may be made upon an averaged or other practicable basis without regard to any distinctions among Owners on account of exchange restrictions, the date of delivery of any American Depositary Shares or otherwise and shall be net of any expenses of conversion into Dollars incurred by the Depositary as provided in Section 5.9 of the Deposit Agreement.

If such conversion or distribution can be effected only with the approval or license of any government or agency thereof, the Depositary shall file such application for approval or license, if any, as it may deem desirable.

A-10

If at any time the Depositary shall determine that in its judgment any foreign currency received by the Depositary or the Custodian is not convertible on a reasonable basis into Dollars transferable to the United States, or if any approval or license of any government or agency thereof which is required for such conversion is denied or in the opinion of the Depositary is not obtainable, or if any such approval or license is not obtained within a reasonable period as determined by the Depositary, the Depositary may distribute the foreign currency (or an appropriate document evidencing the right to receive such foreign currency) received by the Depositary to, or in its discretion may hold such foreign currency uninvested and without liability for interest thereon for the respective accounts of, the Owners entitled to receive the same.

If any such conversion of foreign currency, in whole or in part, cannot be effected for distribution to some of the Owners entitled thereto, the Depositary may in its discretion make such conversion and distribution in Dollars to the extent permissible to the Owners entitled thereto and may distribute the balance of the foreign currency received by the Depositary to, or hold such balance uninvested and without liability for interest thereon for the respective accounts of, the Owners entitled thereto.

### 15. RECORD DATES.

Whenever any cash dividend or other cash distribution shall become payable or any distribution other than cash shall be made, or whenever rights shall be issued with respect to the Deposited Securities, or whenever the Depositary shall receive notice of any meeting of holders of Shares or other Deposited Securities, or whenever for any reason the Depositary causes a change in the number of Shares that are represented by each American Depositary Share, or whenever the Depositary shall find it necessary or convenient, the Depositary shall fix a record date (a) for the determination of the Owners who shall be (i) entitled to receive such dividend, distribution or rights or the net proceeds of the sale thereof, (ii) entitled to give instructions for the exercise of voting rights at any such meeting or (iii) responsible for any fee assessed by the Depositary pursuant to the Deposit Agreement, or (b) on or after which each American Depositary Share will represent the changed number of Shares, subject to the provisions of the Deposit Agreement.

# 16. VOTING OF DEPOSITED SECURITIES.

Upon receipt of notice of any meeting or solicitation of proxies or consents of holders of Shares or other Deposited Securities, if requested in writing by the Company, the Depositary shall, as soon as practicable thereafter, mail to the Owners a notice, the form of which notice shall be in the sole discretion of the Depositary, which shall contain (a) such information (including, without limitation, solicitation materials) as is contained in such notice of meeting received by the Depositary from the Company, (b) a statement that the Owners as of the close of business on a specified record date will be entitled, subject to any applicable provision of Danish law and of the articles of association or similar documents of the Company, to instruct the Depositary as to the

A-11

exercise of the voting rights, if any, pertaining to the amount of Shares or other Deposited Securities represented by their respective American Depositary Shares and (c) a statement as to the manner in which such instructions may be given, including an express indication that instructions may be given or deemed given in accordance with the last sentence of this paragraph if no instruction is received, to the Depositary to give a discretionary proxy to a person designated by the Company. Upon the written request of an Owner of American Depositary Shares on such record date, received on or before the date established by the Depositary for such purpose, the Depositary shall endeavor, in so far as practicable, to vote or cause to be voted the amount of Shares or other Deposited Securities represented by those American Depositary Shares in accordance with the instructions set forth in such request. The Depositary shall not itself exercise any voting discretion over any Deposited Securities. If (i) the Company instructed the Depositary to act under Section 4.7 of the Deposit Agreement and (ii) no instructions are received by the Depositary from an Owner with respect to a matter and an amount of American Depositary Shares of that Owner on or before the date established by the Depositary for such purpose, the Depositary shall deem that Owner to have instructed the Depositary to give a discretionary proxy to a person designated by the Company with respect to that matter and the amount of Deposited Securities represented by that amount of American Depositary Shares and the Depositary shall give a discretionary proxy to a person designated by the Company to vote that amount of Deposited Securities as to that matter, except that no such instruction shall be deemed given and no such discretionary proxy shall be given with respect to any matter as to which the Company informs the Depositary (and the Company agrees to provide such information as promptly as practicable in writing, if applicable) that (x) the Company does not

There can be no assurance that Owners generally or any Owner in particular will receive the notice described in the preceding paragraph sufficiently prior to the instruction cutoff date to ensure that the Depositary will vote the Shares or Deposited Securities in accordance with the provisions set forth in the preceding paragraph.

In order to give Owners a reasonable opportunity to instruct the Depositary as to the exercise of voting rights relating to Deposited Securities, if the Company will request the Depositary to act under this Article 16, the Company shall give the Depositary notice of any such meeting and details concerning the matters to be voted upon at least 45 days in advance of the meeting date.

# 17. CHANGES AFFECTING DEPOSITED SECURITIES.

Upon any change in nominal value, change in par value, split-up, consolidation, or any other reclassification of Deposited Securities, or upon any recapitalization, reorganization, merger or consolidation, or sale of assets affecting the Company or to which it is a party, or upon the redemption or cancellation by the Company of the

A-12

Deposited Securities, any securities, cash or property which shall be received by the Depositary or a Custodian in exchange for, in conversion of, in lieu of or in respect of Deposited Securities shall be treated as new Deposited Securities under the Deposit Agreement, and American Depositary Shares shall thenceforth represent, in addition to the existing Deposited Securities, the right to receive the new Deposited Securities so received, unless additional Receipts are delivered pursuant to the following sentence. In any such case the Depositary may deliver additional American Depositary Shares as in the case of a dividend in Shares, or call for the surrender of outstanding Receipts to be exchanged for new Receipts specifically describing such new Deposited Securities.

### 18. LIABILITY OF THE COMPANY AND DEPOSITARY.

Neither the Depositary nor the Company nor any of their respective directors, officers, employees, agents or affiliates shall incur any liability to any Owner or Holder, (i) if by reason of any provision of any present or future law or regulation of the United States, Denmark or any other country, or of any governmental or regulatory authority or stock exchange, or by reason of any provision, present or future, of the articles of association or any similar document of the Company, or by reason of any provision of any securities issued or distributed by the Company, or any offering or distribution thereof, or by reason of any act of God or war or terrorism or other circumstances beyond its control, the Depositary or the Company shall be prevented, delayed or forbidden from or be subject to any civil or criminal penalty on account of doing or performing any act or thing which by the terms of the Deposit Agreement or Deposited Securities it is provided shall be done or performed, (ii) by reason of any non-performance or delay, caused as aforesaid, in the performance of any act or thing which by the terms of the Deposit Agreement it is provided shall or may be done or performed, (iii) by reason of any exercise of, or failure to exercise, any discretion provided for in the Deposit Agreement, (iv) for the inability of any Owner or Holder to benefit from any distribution, offering, right or other benefit which is made available to holders of Deposited Securities but is not, under the terms of the Deposit Agreement, made available to Owners or Holders, or (v) for any special, consequential or punitive damages for any breach of the terms of the Deposit Agreement. Where, by the terms of a distribution pursuant to Section 4.1, 4.2 or 4.3 of the Deposit Agreement, or an offering or distribution pursuant to Section 4.4 of the Deposit Agreement, or for any other reason, such distribution or offering may not be made available to Owners of Receipts, and the Depositary may not dispose of such distribution or offering on behalf of such Owners and make the net proceeds available to such Owners, then the Depositary shall not make such distribution or offering, and shall allow any rights, if applicable, to lapse. Neither the Company, the Depositary, nor any of their respective directors, officers, employees, agents or affiliates, assumes any obligation or shall be subject to any liability under the Deposit Agreement to Owners or Holders, except that they agree to perform their obligations specifically set forth in the Deposit Agreement without negligence or bad faith. The Depositary shall not be subject to any liability with respect to the validity or worth of the Deposited Securities. Neither the Depositary, the Company nor any of their

respective directors, officers, employees, agents or affiliates shall be under any obligation to appear in, prosecute or defend any action, suit, or other proceeding in respect of any Deposited Securities or in respect of the American Depositary Shares, on behalf of any Owner or Holder or other person. Neither the Depositary nor the Company shall be liable for any action or nonaction by it in reliance upon the advice of or information from legal counsel, accountants, any person presenting Shares for deposit, any Owner or Holder, or any other person believed by it in good faith to be competent to give such advice or information. The Depositary and the Company and their respective directors, officers, employees, agents or affiliates may rely and shall be protected in acting upon any written notice, request, direction or other documents believed by them to be genuine and to have been signed or presented by the proper party or parties. The Depositary shall not be liable for any acts or omissions made by a successor depositary whether in connection with any previous act or omission of the Depositary or in connection with a matter arising wholly after the removal or resignation of the Depositary, provided that in connection with the issue out of which such potential liability arises, the Depositary performed its obligations without negligence or bad faith while it acted as Depositary. The Depositary shall not be liable for the acts or omissions of any securities depository, clearing agency or settlement system in connection with or arising out of book-entry settlement of Deposited Securities or otherwise. The Depositary shall not be responsible for any failure to carry out any instructions to vote any of the Deposited Securities or for the manner in which any such vote is cast or the effect of any such vote, provided that any such action or nonaction is in good faith.

No disclaimer of liability under the Securities Act of 1933 is intended by any provision of the Deposit Agreement.

# 19. RESIGNATION AND REMOVAL OF THE DEPOSITARY; APPOINTMENT OF SUCCESSOR CUSTODIAN.

The Depositary may at any time resign as Depositary under the Deposit Agreement by written notice of its election so to do delivered to the Company, such resignation to take effect upon the earlier of (i) the appointment of a successor depositary and its acceptance of such appointment as provided in the Deposit Agreement or (ii) termination by the Depositary pursuant to Section 6.2 of the Deposit Agreement. The Depositary may at any time be removed by the Company by 120 days prior written notice of such removal, to become effective upon the later of (i) the 120th day after delivery of the notice to the Depositary and (ii) the appointment of a successor depositary and its acceptance of such appointment as provided in the Deposit Agreement. The Depositary in its discretion may appoint a substitute or additional custodian or custodians.

# 20. <u>AMENDMENT.</u>

The form of the Receipts and any provisions of the Deposit Agreement may at any time and from time to time be amended by agreement between the Company and the Depositary without the consent of Owners or Holders in any respect which they may

Δ\_1/

deem necessary or desirable. Any amendment which shall impose or increase any fees or charges (other than taxes and other governmental charges, registration fees, cable, telex or facsimile transmission costs, delivery costs or other such expenses), or which shall otherwise prejudice any substantial existing right of Owners, shall, however, not become effective as to outstanding American Depositary Shares until the expiration of thirty days after notice of such amendment shall have been given to the Owners of outstanding American Depositary Shares. Every Owner and Holder of American Depositary Shares, at the time any amendment so becomes effective, shall be deemed, by continuing to hold such American Depositary Shares or any interest therein, to consent and agree to such amendment and to be bound by the Deposit Agreement as amended thereby. In no event shall any amendment impair the right of the Owner to surrender American Depositary Shares and receive therefor the Deposited Securities represented thereby, except in order to comply with mandatory provisions of applicable law.

# 21. TERMINATION OF DEPOSIT AGREEMENT.

The Company may terminate the Deposit Agreement by instructing the Depositary to mail notice of termination to the Owners of all American Depositary Shares then outstanding at least 30 days prior to the termination date included in such notice. The Depositary may likewise terminate the Deposit Agreement, if at any time 60 days shall have expired after the Depositary delivered to the Company a written resignation notice and if a successor depositary shall not have been appointed and accepted its appointment as provided in the Deposit Agreement; in such case the Depositary shall mail a notice of termination to the Owners of all American Depositary Shares then outstanding at least 30 days prior to the termination date. On and after the date of termination, the Owner of American Depositary Shares will, upon (a) surrender of such American Depositary Shares, (b) payment of the fee of the Depositary for the surrender of American Depositary Shares referred to in Section 2.5, and (c) payment of any applicable taxes or governmental charges, be entitled to delivery, to him or upon his order, of the amount of Deposited Securities represented by those American Depositary Shares. If any American Depositary Shares shall remain outstanding after the date of termination, the Depositary thereafter shall discontinue the registration of transfers of American Depositary Shares, shall suspend the distribution of dividends to the Owners thereof, and shall not give any further notices or perform any further acts under the Deposit Agreement, except that the Depositary shall continue to collect dividends and other distributions pertaining to Deposited Securities, shall sell rights and other property as provided in the Deposit Agreement, and shall continue to deliver Deposited Securities, together with any dividends or other distributions received with respect thereto and the net proceeds of the sale of any rights or other property, upon surrender of American Depositary Shares (after deducting, in each case, the fee of the Depositary for the surrender of American Depositary Shares, any expenses for the account of the Owner of such American Depositary Shares in accordance with the terms and conditions of the Deposit Agreement, and any applicable taxes or governmental charges). At any time after the expiration of four months from the date of termination, the Depositary may sell the

A-15

Deposited Securities then held under the Deposit Agreement and may thereafter hold uninvested the net proceeds of any such sale, together with any other cash then held by it thereunder, unsegregated and without liability for interest, for the pro rata benefit of the Owners of American Depositary Shares that have not theretofore been surrendered, such Owners thereupon becoming general creditors of the Depositary with respect to such net proceeds. After making such sale, the Depositary shall be discharged from all obligations under the Deposit Agreement, except to account for such net proceeds and other cash (after deducting, in each case, the fee of the Depositary for the surrender of American Depositary Shares, any expenses for the account of the Owner of such American Depositary Shares in accordance with the terms and conditions of the Deposit Agreement, and any applicable taxes or governmental charges). Upon the termination of the Deposit Agreement, the Company shall be discharged from all obligations under the Deposit Agreement except for its obligations to the Depositary with respect to indemnification, charges, and expenses.

# 22. DTC DIRECT REGISTRATION SYSTEM AND PROFILE MODIFICATION SYSTEM.

- (a) Notwithstanding the provisions of Section 2.4 of the Deposit Agreement, the parties acknowledge that the Direct Registration System ("DRS") and Profile Modification System ("Profile") shall apply to uncertificated American Depositary Shares upon acceptance thereof to DRS by DTC. DRS is the system administered by DTC pursuant to which the Depositary may register the ownership of uncertificated American Depositary Shares, which ownership shall be evidenced by periodic statements issued by the Depositary to the Owners entitled thereto. Profile is a required feature of DRS which allows a DTC participant, claiming to act on behalf of an Owner, to direct the Depositary to register a transfer of those American Depositary Shares to DTC or its nominee and to deliver those American Depositary Shares to the DTC account of that DTC participant without receipt by the Depositary of prior authorization from the Owner to register such transfer.
- (b) In connection with and in accordance with the arrangements and procedures relating to DRS/Profile, the parties understand that the Depositary will not verify, determine or otherwise ascertain that the DTC participant which is claiming to be acting on behalf of an Owner in requesting registration of transfer and delivery described in subsection (a) has the actual authority to act on behalf of the Owner (notwithstanding any requirements under the Uniform Commercial Code). For the avoidance of doubt, the provisions of Sections 5.3 and 5.8 of the Deposit Agreement shall apply to the matters arising from the use of the DRS. The parties agree that the Depositary's reliance on and compliance with instructions received by the Depositary through the DRS/Profile System and in accordance with the Deposit Agreement, shall not constitute negligence or bad faith on the part of the Depositary.

A-16

# 23. REGISTRATION OF SHARES; SHARE REGISTER.

The Company agrees to either maintain itself or engage, subject to shareholder approval, a third party (a "Transfer Agent") reasonably acceptable to the Depositary to maintain a Share Register for the Shares for so long as any American Depositary Shares or Receipts remain outstanding hereunder or this Agreement remains in force.

The Company agrees that it shall, or if the Share Register is maintained by a Transfer Agent, cooperate with the Depositary to ensure that such Transfer Agent shall at any time and from time to time: (a) take any and all action as may be necessary to assure the accuracy and completeness of all information set forth in the Share Register in respect of the Shares; (b) provide to the Depositary, the Custodian or their respective agents unrestricted access to such part of the Share Register, which relates to the Shares during regular business hours in accordance with Danish law, in such manner and upon such terms and conditions as the Depositary may, in its sole reasonable discretion, deem appropriate, to permit the Depositary, the Custodian or their respective agents to confirm the number of Shares registered in the name of the Depositary, the Custodian or their respective nominees, as applicable, pursuant to the terms of the Deposit Agreement and, in connection therewith, to provide the Depositary, the Custodian or their respective agents, upon request, with a duplicative extract from the relevant part of the Share Register duly certified by the Company or the Transfer Agent, as applicable, (or other independent third party reasonably acceptable to the Depositary); (c) promptly effect the re-registration of ownership of Shares deposited pursuant to Section 2.2 of the Deposit Agreement in the Share Register in connection with any deposit or withdrawal of Shares under the Deposit Agreement; (d) permit the Depositary or the Custodian to register any Shares held hereunder in the name of the Depositary, the Custodian or their respective nominees; and (e) to the extent permissible under applicable law, promptly notify the Depositary in writing at any time that (A) the Company or the Transfer Agent, as applicable, eliminates the name of a shareholder of the Company from the Share Register or otherwise alters a shareholder's interest in the Shares and such shareholder alleges to the Company or the Transfer Agent, as applicable, or publicly that such elimination or alteration is unlawful; (B) the Company no longer will be able materially to comply with, or has engaged in conduct that indicates it will not materially comply with, the provisions of Section 5.13 of the Deposit Agreement relating to it; (C) the Company or the Transfer Agent, as applicable, refuses to re-register Shares in the name of a particular purchaser and such purchaser (or its respective seller) alleges that such refusal is unlawful; (D) the Company or the Transfer Agent, as applicable, holds Shares for the account of the Company; or (E) the Company has materially breached the provisions of Section 5.13 of the Deposit Agreement relating to it and has failed to cure such breach within a reasonable time.

The Depositary agrees that it will instruct the Custodian to maintain custody of all duplicative Share Register extracts (or other evidence of verification) provided to the Depositary, the Custodian or their respective agents pursuant to Section 5.13(b) of the Deposit Agreement. In the event of any material discrepancy between the records of the Depositary or the Custodian and the Share Register, then, if an officer of the ADR

A-17

Department of the Depositary has actual knowledge of such discrepancy, the Depositary shall promptly notify the Company. In the event of any discrepancy between the records of the Depositary or the Custodian and the Share Register, the Company agrees that (whether or not it has received any notification from the Depositary) it will (i) use, or if the Share Register is maintained by a Transfer Agent, corporate with the Depositary to ensure that the Transfer Agent will use its reasonable efforts to cause the Company to reconcile its records to the records of the Depositary or the Custodian and to make such corrections or revisions in the Share Register as may be necessary in connection therewith, and (ii) to the extent the Company or the Transfer Agent, as applicable, is unable to so reconcile such records, promptly instruct the Depositary to notify the Owners of the existence of such discrepancy. Upon receipt of such instruction, the Depositary shall promptly give such notification to the Owners pursuant to Section 4.9 of the Deposit Agreement (it being understood that the Depositary may at any time give such notification to the Owners, whether or not it has received instructions from the Company), and the Depositary shall promptly cease issuing American Depositary Shares pursuant to Section 2.2 of the Deposit Agreement until such time as, in the opinion of the Depositary, such records have been appropriately reconciled.

### 24. ARBITRATION; SETTLEMENT OF DISPUTES.

(a) Any controversy, claim or cause of action brought by any party hereto against the Company arising out of or relating to the Shares or other Deposited Securities, the American Depositary Shares, the Receipts or the Deposit Agreement, or the breach hereof or thereof, shall be settled by arbitration in accordance with the International Arbitration Rules of the American Arbitration Association, and judgment upon the award rendered by the arbitrators may be entered in any court having jurisdiction thereof; <u>provided</u>, <u>however</u>, that in the event of any third-party litigation to which the Depositary is a party and to which the Company may properly be joined, the Company may be so joined in any court in which such litigation is proceeding; and <u>provided</u>, <u>further</u>, that any such controversy, claim or cause of action brought by a party hereto against the Company relating to or based upon the provisions of the Federal securities laws of the United States or the rules and regulations promulgated thereunder shall be submitted to arbitration as provided in this Article 24 if, but only if, so elected by the claimant.

- (b) The place of the arbitration shall be The City of New York, State of New York, United States of America, and the language of the arbitration shall be English.
- (c) The number of arbitrators shall be three, each of whom shall be disinterested in the dispute or controversy, shall have no connection with any party thereto, and shall be an attorney experienced in international securities transactions. Each party shall appoint one arbitrator and the two arbitrators shall select a third arbitrator who shall serve as chairperson of the tribunal. If a dispute, controversy or cause of action

### A-18

shall involve more than two parties, the parties shall attempt to align themselves in two sides (i.e., claimant(s) and respondent(s)), each of which shall appoint one arbitrator as if there were only two parties to such dispute, controversy or cause of action. If such alignment and appointment shall not have occurred within thirty (30) calendar days after the initiating party serves the arbitration demand, the American Arbitration Association shall appoint the three arbitrators, each of whom shall have the qualifications described above. The parties and the American Arbitration Association may appoint from among the nationals of any country, whether or not a party is a national of that country.

- (d) The arbitral tribunal shall have no authority to award any consequential, special or punitive damages or other damages not measured by the prevailing party's actual damages and may not, in any event, make any ruling, finding or award that does not conform to the terms and conditions of the Deposit Agreement.
- (e) Any controversy, claim or cause of action arising out of or relating to the Shares or other Deposited Securities, the American Depositary Shares, the Receipts or the Deposit Agreement not subject to arbitration hereunder shall be litigated in the Federal and state courts in the Borough of Manhattan, The City of New York and the Company hereby submits to the personal jurisdiction of the court in which such action or proceeding is brought.

# 25. <u>SUBMISSION TO JURISDICTION; JURY TRIAL WAIVER; WAIVER OF IMMUNITIES.</u>

In the Deposit Agreement, the Company has (i) appointed CT Corporation System, 1015 15<sup>th</sup> Street, NW, Suite 1000, Washington, DC 20005, as the Company's authorized agent upon which process may be served in any suit or proceeding (including any arbitration proceeding) arising out of or relating to the Shares or Deposited Securities, the American Depositary Shares, the Receipts or the Deposit Agreement, (ii) consented and submitted to the jurisdiction of any state or federal court in the State of New York in which any such suit or proceeding may be instituted, and (iii) agreed that service of process upon said authorized agent shall be deemed in every respect effective service of process upon the Company in any such suit or proceeding.

EACH PARTY TO THE DEPOSIT AGREEMENT (INCLUDING, FOR AVOIDANCE OF DOUBT, EACH OWNER AND HOLDER) HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY SUIT, ACTION OR PROCEEDING AGAINST THE COMPANY AND/OR THE DEPOSITARY DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THE SHARES OR OTHER DEPOSITED SECURITIES, THE AMERICAN DEPOSITARY SHARES OR THE RECEIPTS, THE DEPOSIT AGREEMENT OR ANY TRANSACTION CONTEMPLATED HEREIN OR THEREIN, OR THE BREACH HEREOF OR THEREOF, INCLUDING WITHOUT LIMITATION ANY

### A-19

# QUESTION REGARDING EXISTENCE, VALIDITY OR TERMINATION (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY).

To the extent that the Company or any of its properties, assets or revenues may have or hereafter become entitled to, or have attributed to it, any right of immunity, on the grounds of sovereignty or otherwise, from any legal action, suit or proceeding, from the giving of any relief in any respect thereof, from setoff or counterclaim, from the jurisdiction of any court, from service of process, from attachment upon or prior to judgment, from attachment in aid of execution or judgment, or other legal process or proceeding for the giving of any relief or for the enforcement of any judgment, in any jurisdiction in which proceedings may at any time be commenced, with respect to its obligations, liabilities or any other matter under or arising out of or in connection with the Shares or Deposited Securities, the American Depositary Shares, the Receipts or the Deposit Agreement, the Company, to the fullest extent permitted by law, hereby irrevocably and unconditionally waives, and agrees not to plead or claim, any such immunity and consents to such relief and enforcement.

# 26. <u>DISCLOSURE OF INTERESTS</u>

The Company may from time to time request Owners or Holders or former Owners or Holders to provide information as to the capacity in which they hold or held American Depositary Shares and regarding the identity of any other persons then or previously interested in such American Depositary Shares and the nature of such interest and various other matters. Each such Owner or Holder agrees to provide any such information reasonably requested by the Company or the Depositary pursuant to this Article 26 whether or not still an Owner or Holder at the time of such request. The Depositary agrees to use its reasonable efforts, at the Company's expense, to comply with written instructions received from the Company requesting that the Depositary forward any such requests to such Owners or Holders and to the last known address, if any, of such former Owners or Holders and to forward to the Company any responses to such requests received by the Depositary. However, nothing in this Article 26 shall be interpreted as obligating the Depositary to provide or obtain any such information not provided to the Depositary by such Owners or Holders or former Owners or Holders.

# STOCK LENDING AGREEMENT

Between Nordic Biotech Opportunity Fund K/S

c/o NB Capital ApS Østergade 24 A, 1 DK-1100 Copenhagen

Denmark

and Forward Pharma A/S

Østergade 24 A, 1 DK-1100 Copenhagen K

Denmark

and Leerink Partners LLC

1 Federal Street, 37th Floor Boston, MA 02110 United States of America

# TABLE OF CONTENTS

1	DELIVERY AND USE OF THE FIRM LENDING SHARES	9
2	DELIVERY AND USE OF THE OPTION LENDING SHARES	10
3	EXCHANGE OF SHARES	12
4	REDELIVERY	13
5	NO CLOSING	13
6	PROLONGATION OF SHARE LOAN	14
7	RIGHTS AND TITLE	14
8	REPRESENTATIONS AND WARRANTIES	15
9	INDEMNIFICATION	15
10	EXPENSES, TAXES AND OTHER DUTIES	16
11	TERMINATION	16
12	GOVERNING LAW; SUBMISSION TO JURISDICTION	16
13	SEVERABILITY	18
14	NOTICES	18
15	COUNTERPARTS	19
	2	

# INDEX OF EXHIBITS AND ANNEXES

Exhibit 1:	Notice from the Lender and the Company to the Transfer Agent of the transfer of Firm Lending Shares
Exhibit 2:	Notice from the Company and the Depository to the Transfer Agent of the transfer of Firm Lending Shares
Exhibit 3:	Form Borrowing Notice from the Company to the Lender
Exhibit 4:	Form Notice from the Company and the Depository to the Transfer Agent of the transfer of Option Lending Shares
Exhibit 5:	Notice from Leerink and the Depository to the Transfer Agent of the transfer of Firm Shares

Exhibit 6: Notice from the Company and the Depository to the Transfer Agent of the transfer of Firm Lending Shares

Exhibit 7: Form notice from Leerink and the Depository to the Transfer Agent of the transfer of Option Shares

Exhibit 11:	Form notice from the Company and the Depository to the Tra Shares in case of no Closing	ansfer Agent of the transfer of the Lending Shares and/or the Option Lending
	3	
Exhibit 12:	Form notice from the Company and the Lender to the Transfer Agent of the transfer of the Lending Shares and/or the Option Lending Shares in case of no Closing	
Exhibit 13:	Form notice of prolongation from Leerink to the Lender	
Annex 1:	Agreement between the Company, Leerink and the Depositary	
	4	
	INDEX OF DEFIN	NED TERMS
		ven to them in the Underwriting Agreement (as defined below). In addition,
ADSs	h	nas the meaning set out in recital (A).
Agreement	п	neans this Agreement.
Borrowing Notice	h	nas the meaning set out in Clause 2.1.
Business Day		means any day which is not a Saturday, a Sunday or a bank or public holiday in New York (USA) or Copenhagen (Denmark).
Closing Date		means the [·] 2014, regardless of whether settlement of the Firm ADSs actually occurs.
Company	h	nas the meaning given to in the preamble of the Agreement.
Depositary	W	means the Bank of New York Mellon, a New York banking corporation whose principal office is located at One Wall Street, New York, NY 10286 (USA).
Firm ADSs	h	nas the meaning set out in recital (A).
	5	
Firm Lending Shar	res h	nas the meaning set out in recital (B).
Firm Shares	h	nas the meaning set out in recital (A).
Leerink	h	nas the meaning given to in the preamble of the Agreement.
Lender	h	nas the meaning given to in the preamble of the Agreement.
Lending Shares	h	nas the meaning set out in recital (B).
Offering	h	nas the meaning set out in recital (A)
Option ADSs	h	nas the meaning set out in recital (A).
Option Closing Da	n A	means the expected settlement date for the Option ADSs as set out in the notice of exercise of overallotment option (as described in the Underwriting Agreement), regardless of whether settlement of the Option ADSs actually occurs.
Option Lending Sl	hares h	nas the meaning set out in recital (C).
Option Shares	h	nas the meaning set out in recital (A).

Form notice from the Company and the Depository to the Transfer Agent of the transfer of Option Lending Shares

Form notice from the Company and the Lender to the Transfer Agent of the redelivery of the Option Lending Shares

Notice from the Company and the Lender to the Transfer Agent of the redelivery of the Firm Lending Shares

Exhibit 8:

Exhibit 9:

Exhibit 10:

Ordinary Shares

means the Company's ordinary shares, nominal value DKK [0.05] per share.

Overallotment Option

has the meaning set out in recital (A).

Purchase Price

has the meaning set out in the Underwriting Agreement.

Subscription List

has the meaning set out in recital (A).

Transaction Date

means one Business Day prior to the Closing Date.

Transfer Agent

means Computershare A/S.

Underwriters

has the meaning set out in the Underwriting Agreement.

**Underwriting Agreement** 

has the meaning set out in recital (A).

7

This Stock Lending Agreement is entered into on [Date] between:

- (1) Nordic Biotech Opportunity Fund K/S, a limited partnership incorporated under the laws of the Kingdom of Denmark registered with the Danish Business Authority under CVR number 31 26 24 88, and whose registered office is c/o NB Capital ApS, Østergade 24 A, 1, DK-1100 Copenhagen (the "Lender");
- (2) Forward Pharma A/S, a public limited liability company incorporated under the laws of the Kingdom of Denmark registered with the Danish Business Authority under CVR number 28 86 58 80, and whose registered office is at Østergade 24 A, 1, DK-1100 Copenhagen (the "Company");
- (3) Leerink Partners LLC whose registered office is at 1 Federal Street, 37th Floor Boston, MA 02110 ("Leerink");
  - the Lender, Leerink and the Company hereinafter collectively referred to as the "Parties" and separately as a "Party"

# WHEREAS

Leerink is acting as Representative of the Underwriters (as defined in the Underwriting Agreement dated the date hereof among the Company and the Underwriters (the "Underwriting Agreement")) with respect to (i) the sale by the Company and the purchase by the Underwriters from the Company, acting severally and not jointly, of the respective numbers of Ordinary Shares, to be represented by and delivered in the form of American Depositary Shares (the "Firm ADSs"), each ADS representing one Ordinary Share, set forth in Schedule A to the Underwriting Agreement (the "Offering") and (ii) the grant by the Company to the Underwriters, acting severally and not jointly, of the option described in Section 2(b) hereof to purchase from the Company all or any part of [•] additional Ordinary Shares, to be represented by and delivered in the form of ADSs (the "Option ADSs" and together with the Firm ADS the "ADSs"), each Option ADS representing one Ordinary Shares solely for the purpose of covering over-allotments (the "Overallotment Option"). In connection with the Offering, Leerink (as Representative of the Underwriters) will subscribe for such number of Ordinary Shares issued by the Company as will be set out in Schedule A to the Underwriting Agreement (the "Firm Shares") pursuant to a Subscription list to be entered into to facilitate the transactions contemplated by the Underwriting Agreement (the "Subscription List"). Pursuant to

8

Clause [26] of the Underwriting Agreement, the Company has further granted the Underwriters an option, exercisable by Leerink (as Representative of the Underwriters), to subscribe for such number of Ordinary Shares issued by the Company as is set forth in the Underwriting Agreement (the "Option Shares"). Such exercise of the Underwriters' option to purchase the Option Shares shall be evidenced by a separate Subscription List. The Company desires to provide, as set forth in the Deposit Agreement among the Company and the Depositary, for the deposit of Ordinary Shares with the Depositary for the creation of ADSs representing Ordinary Shares.

- (B) Subject to the Company and the Underwriters entering into the Underwriting Agreement, the Lender desires to lend to the Company [·] existing Ordinary Shares (the "Firm Lending Shares") in an amount that equals the number of Ordinary Shares underlying the Firm ADSs in connection with the Offering, and the Company desires to borrow such Firm Lending Shares from the Lender, for the purpose of surrendering such Firm Lending Shares to the Depositary and thereby enabling the Depositary to create the Firm ADSs before the Firm Shares are paid in, issued and registered with the Danish Business Authority.
- Subject to the Underwriters exercise of the Overallotment Option, the Lender desires to lend to the Company up to [•] existing Ordinary Shares in the Company (the "Option Lending Shares" and together with the Firm Lending Shares the "Lending Shares") in an amount that equals the number of Ordinary Shares underlying the Option ADSs in connection with the Offering and the Company desires to borrow such Option Lending Shares from the Lender, for the purpose of surrendering such Option Lending Shares to the Depositary and thereby enabling the Depositary to create the Option ADSs before the Option Shares are paid in, issued and registered with the Danish Business Authority.

### **Delivery and use of the Firm Lending Shares**

1.1 The Lender hereby agrees to lend the Firm Lending Shares to the Company and the Lender and the Company shall give notice to the Transfer Agent of the transfer, for value and available for delivery not later than [Time] on the Transaction Date. No consideration is payable by the Company under this Agreement. A signed

q

notice is attached to this Agreement as <u>Exhibit 1</u> and held in escrow with automatic release on [Time] on the Transaction Date. The transfer of the Firm Lending Shares from the Lender to the Company shall be evidenced by an updated shareholders' register signed and delivered by the Transfer Agent.

- As soon as practicable following receipt by the Company of the updated shareholders' register signed and delivered by the Transfer Agent pursuant to Clause 1.1, the Firm Lending Shares shall be transferred and surrendered (which for the purpose of this Agreement is characterized as a transfer of all rights and title to the Firm Lending Shares) by the Company to the Depositary in accordance with Annex 1 to this Agreement, for value and available for delivery not later than [Time] on the Transaction Date. A signed notice is attached to this Agreement as <a href="Exhibit 2">Exhibit 2</a> (and attached as Exhibit 2.1 to Annex 1) and held in escrow with automatic release following delivery of the updated shareholders' register signed and delivered by the Transfer Agent pursuant to Clause 1.1. The transfer of the Firm Lending Shares from the Company to the Depositary shall be evidenced by an updated shareholders' register signed and delivered by the Transfer Agent.
- 1.3 The Firm Lending Shares are lent subject to the Company's separate obligation to redeliver the Firm Lending Shares or Firm Shares (as defined below) as provided for respectively in Clauses 4 or 5 hereof. The Firm Lending Shares shall be used by the Company solely for the purpose of surrendering the Firm Lending Shares to the Depositary, and thereby enabling the Depositary to create the Firm ADSs before the Firm Shares are paid in, issued and registered with the Danish Business Authority.
- 1.4 If during the period while it holds the Firm Lending Shares, the Company receives any interest, dividends or other distributions in respect of the Firm Lending Shares, it will deliver such interest, dividends or distributions to the Lender.

# 2 Delivery and Use of the Option Lending Shares

2.1 The Lender hereby agrees to lend the Option Lending Shares to the Company upon delivery to the Lender by the Company of a signed and completed notice (a form "Borrowing Notice", attached as <u>Exhibit 3</u>), specifying the number of Option Lending Shares to be borrowed. The Company may deliver multiple Borrowing Notices and as a consequence hereof, Clauses 2 (Delivery and Use of the Option Lending

10

Shares), 3 (Exchange of Shares), 4 (Redelivery), and 5 (No Closing) shall apply to each loan of Option Lending Shares.

- 2.2 The Borrowing Notice shall also be deemed as a notice to the Transfer Agent of the transfer pursuant to Clause 2.1. The transfer of the Option Lending Shares from the Lender to the Company shall be for value and available for delivery not later than [Time] on the Business Day following receipt of the Borrowing Notice and shall be evidenced by an updated shareholders' register signed and delivered by the Transfer Agent.
- As soon as practicable following receipt by the Company of the updated shareholders' register signed and delivered by the Transfer Agent pursuant to Clause 2.2, the Option Lending Shares shall be transferred and surrendered (which for the purpose of this Agreement is characterized as a transfer of all rights and title to the Option Lending Shares) by the Company to the Depositary in accordance with and pursuant to Annex 1 to this Agreement. In order to effect the aforementioned transfer, notice shall be given to the Transfer Agent. A notice form is attached to this Agreement as <a href="Exhibit 4">Exhibit 4</a> (and attached as <a href="Exhibit 3.1">Exhibit 4</a> (and attached as <a href="Exhibit 3.1">Exhibit 3.1</a> to Annex 1). The Option Lending Shares shall be transferred for value and available for delivery not later than [Time] on the Business Day following receipt of the notice referred to in the preceding paragraph and shall be evidenced by an updated shareholders' register signed and delivered by the Transfer Agent.
- 2.4 The Option Lending Shares are lent subject to the Company's separate obligation to redeliver the Option Lending Shares or Firm Option Shares as provided for in Clauses 4 or 5 hereof. The Option Lending Shares shall be used by the Company solely for the purpose of surrendering such Option Lending Shares to the Depositary and thereby enabling the Depositary to create the Option ADSs before the Option Shares are paid in, issued and registered with the Danish Business Authority.
- 2.5 If during the period while it holds the Option Lending Shares, the Company receives any interest, dividends or other distributions in respect of the Option Lending Shares, it will deliver such interest, dividends or distributions to the Lender.

11

# 3 Exchange of Shares

- Subject to, and following the issuance, creation and delivery of the Firm ADS's and/or the Option ADS's (as the case may be) by the Depositary to an account designated by Leerink, Leerink will, (as Representative of the Underwriters) and pursuant to the Subscription List, subscribe for the Firm Shares and/or the Option Shares (as the case may be).
- As soon as practicable following registration of a capital increase with the Danish Business Authority effecting the issuance of the Firm Shares and/or the Option Shares (as the case may be) and delivery of an updated shareholders' register signed and delivered by the Transfer Agent, the

Firm Shares and/or the Option Shares (as the case may be), shall, in accordance with and pursuant to Annex 1, be substituted for the Firm Lending Shares and/or the Option Lending Shares (as the case may be) to the effect that (i) the Firm Shares and/or the Option Shares (as the case may be) are transferred from Leerink to the Depositary, and (ii) as soon as practicable thereafter the Firm Lending Shares and/or the Option Lending Shares (as the case may be) are transferred from the Depositary to the Company. In order to effect the aforementioned substitution, notice shall be given to the Transfer Agent. With respect to the transfer of Firm Shares from Leerink to the Depositary, a signed notice is attached as Exhibit 5 (and attached as Exhibit 4.1 to Annex 1) and held in escrow with automatic release upon registration with the Danish Business Authority and delivery of an updated shareholders' register signed and delivered by the Transfer Agent pursuant to Clause 3.1. With respect to the transfer of Firm Lending Shares from the Depositary to the Company, a signed notice is attached as Exhibit 6 (and attached as Exhibit 4.2 to Annex 1) and held in escrow with automatic release upon release of Exhibit 5 and delivery of an updated shareholders' register signed and delivered by the Transfer Agent. With respect to the transfer of Option Shares from Leerink to the Depositary, a form notice is attached as Exhibit 5.1 to Annex 1) and with respect to the transfer of Option Lending Shares from the Depositary to the Company, a form notice is attached as Exhibit 8 (and attached as Exhibit 5.2 to Annex 1). In each case, the exchange of shares pursuant to this Clause 3.2 shall be evidenced by an updated shareholders' register signed and delivered by the Transfer Agent.

12

# 4 Redelivery

As soon as practicable following the exchange of shares pursuant to Clause 3.2, the Company shall redeliver the Firm Lending Shares and/or the Option Lending Shares (as the case may be) to the Lender. In order to effect the aforementioned redelivery, notice shall be given to the Transfer Agent by the Company and the Lender as soon as practicable. With respect to the Firm Lending Shares, a signed notice is attached as <a href="Exhibit 9">Exhibit 9</a> and held in escrow with automatic release upon delivery of an updated shareholders' register signed and delivered by the Transfer Agent pursuant to Clause 3.2. With respect to the Option Lending Shares, a form notice is attached as <a href="Exhibit 10">Exhibit 10</a>. In each case, the redelivery shall be evidenced by an updated shareholders' register signed and delivered by the Transfer Agent.

### 5 No Closing

- In the event that (i) registration of the Firm Shares or the Option Shares (as the case may be) with the Danish Business Authority has not occurred by [Time] on the Closing Date or the Option Closing Date (as the case may be), or (ii) delivery of the Firm ADS's and/or the Option ADS's (as the case may be) by the Depositary to an account designated by Leerink has not occurred by [Time] on the Closing Date or the Option Closing Date (as the case may be), the Depository shall, upon certification by Leerink that the applicable transaction has failed to close (which Leerink may decide to postpone for such period, if any, which the loan period is extended pursuant to clause 6.1), in accordance with and pursuant to Annex 1 to this Agreement, as soon as practicable redeliver to the Company the Lending Shares and/or the Option Lending Shares (as the case may be) and the Company and the Depository shall, in accordance with and pursuant to Annex 1 to this Agreement, as soon as practicable give notice to the Transfer Agent of such redelivery. A form notice is attached as Exhibit 11 (and attached as Exhibit 6.1B to Annex 1). The redelivery shall be evidenced by an updated shareholders' register signed and delivered by the Transfer Agent.
- As soon as practicable after the redelivery of the Lending Shares and/or the Option Lending Shares (as the case may be) by the Depositary to the Company pursuant to Annex 1 to this Agreement, the Company shall redeliver the Lending Shares and/or the Option Lending Shares (as the case may be) to the Lender, and the Company and the Lender shall as soon as practicable give notice to the Transfer

13

Agent of the redelivery. A form notice is attached as <u>Exhibit 12</u>. The redelivery shall be evidenced by an updated shareholders' register signed and delivered by the Transfer Agent.

# 6 Prolongation of share loan

- In the event that (i) registration of the Firm Shares or the Option Shares (as the case may be) with the Danish Business Authority has not occurred by [Time] on the Closing Date or the Option Closing Date (as the case may be), or (ii) delivery of the Firm ADS's and/or the Option ADS's (as the case may be) by the Depositary to an account designated by Leerink has not occurred by [Time] on the Closing Date or the Option Closing Date (as the case may be), Leerink may, upon written notice to the Company and the Lender (a notice form is attached as Exhibit 13 and attached as Exhibit 8 to Annex 1), elect to continue the loan of the Lending Shares until the earlier of (x) registration of the Firm Shares or the Option Shares (as the case may be) has occurred and (y) five (5) business after the Closing Date or the Option Closing Date (as the case may be).
- In case this Agreement is prolonged pursuant to Clause 6.1, Leerink (as Representative of the Underwriters) shall not be obligated to release the net amount of the Purchase Price which would otherwise be payable to the Company pursuant to the Underwriting Agreement before registration of the Firm Shares or the Option Shares (as the case may be) has been effected and this has been sufficiently documented by the Company by way of presenting a compiled summary from the Danish Business Authority with a translation into the English language. The Purchase Price shall carry no interest.

### 7 Rights and Title

7.1 Each of the Lender, the Company and Leerink shall execute and deliver all necessary documents and give all necessary instructions to procure that all right, title and interest in the Lending Shares delivered pursuant to this Agreement shall pass between the Parties (as the case may be), free from all liens, charges, encumbrances or transfer taxes.

- 8.1 Each Party hereby represents, warrants and undertakes that (a) it is not restricted under the terms of its articles of association, charter, bylaws or other governing documents, the law of its place of incorporation or jurisdiction in which it is domiciled, statute, order, rule or regulation or in any other manner from lending or borrowing (as the case may be) the Lending Shares in accordance with this Agreement, or from otherwise performing its obligations hereunder, and (b) it has all necessary licences and approvals and is duly authorised and empowered to perform its duties and obligations under this Agreement and will do nothing to prejudice the continuation of such authorisations, licences or approvals.
- In addition, each Party hereby represents, warrants and undertakes to the other Parties that it is entitled to pass full legal and beneficial ownership of the Firm Shares and/or the Option Shares (as the case may be), free from all liens, charges, encumbrances and transfer taxes.
- 8.3 Finally, the Lender hereby represents, warrants and undertakes to the Company and to Leerink (As Representative of the Underwriters) that it is acting for its own account, and it has made its own independent decisions to enter into the loan of the Lending Shares on the terms and conditions set out in this Agreement and as to whether said loan is appropriate or proper for it based upon its own judgment and upon advice from such advisers as it has deemed necessary. It is not relying on any communication (written or oral) of Leerink or any other Underwriter as investment advice or as a recommendation to enter into the loan of the Lending Shares; it being understood that information and explanations related to the terms and conditions of such loan shall not be considered investment advice or a recommendation to enter into such loan. No communication (written or oral) received from Leerink or any other Underwriter shall be deemed to be an assurance or guarantee as to the expected results of the loan of the Lending Shares.

# 9 Indemnification

9.1 The Company shall, indemnify and hold harmless each of the Lender, Leerink, and each of the other Underwriters their affiliates, their selling agents, directors, officers, members and representatives and each person, if any against any and all losses, claims, expenses, damages, liabilities and/or actions suffered or incurred

15

by any of the said persons directly or indirectly as a result of (i) breach by the Company of any of the representations and warranties made by the Company or the Lender set out in clause 8 above or (ii) failure by the Company to perform any corporate action set out or implied in this Agreement.

- 9.2 As this indemnification is a prerequisite for the transactions described in Recitals (A)-(C), the Board of Directors of the Company have in connection with signing this Agreement assessed that the indemnification in this Clause 9 is in the overall best interest of the Company.
- 9.3 Section 6(c)-6(d) and Section 7 of the Underwriting Agreement shall apply *mutadis mutandis* to the indemnification in this Clause 9.

# 10 Expenses, Taxes and other Duties

- Expenses, taxes and other duties triggered by the delivery and/or redelivery of the Firm Shares, Option Shares or Lending Shares under this Agreement, shall be borne by the Company.
- The Company shall not pay any fees, interest or other consideration or provide any collateral to the Lender or any other party as payment for the lending of the Lending Shares.

# 11 Termination

11.1 This Agreement shall terminate automatically when all Lending Shares are redelivered to the Lender.

# 12 Governing Law; Submission to Jurisdiction

- 12.1 This Agreement, the relationship among the parties and any non-contractual obligations arising out of or in connection with this Agreement shall be governed by and construed in accordance with Danish law.
- 12.2 Each of the Parties irrevocably submits to the jurisdiction of the courts of the Kingdom of Denmark, in any legal suit, action or proceeding based on or arising under this Agreement whether on a contractual or non-contractual basis and

16

agrees that all claims in respect of such suit or proceeding may be determined in any such court. Further, each of the Parties irrevocably waives the defence of an inconvenient forum or objections to personal jurisdiction with respect to the maintenance of such legal suit, action or proceeding. Lastly, and to the extent permitted by law, each of the Parties hereby irrevocably waives any objections to the enforcement by any competent court of any judgment validly obtained in any such court in the Kingdom of Denmark on the basis of any such legal suit, action or proceeding.

12.3 Notwithstanding the agreement in Clause 12.1, the Underwriters shall retain the right to bring proceedings against the Company and the Lender in any other court of competent jurisdiction or concurrently in more than one jurisdiction. The Underwriters shall retain the right to join or counterclaim under or for any breach of this Agreement against the Company and the Lender in any proceeding to which the Underwriters are or will be enjoined or a party in any other court in any jurisdiction outside of the Kingdom of Denmark relating to this Agreement, including, without prejudice to the generality of the foregoing, in any court of competent jurisdiction in the Kingdom of Denmark. The Company and the Lender irrevocably waives any objection to the jurisdiction of any court referred to in this Clause 12.3.

12.4	The Company and the Lender agrees to appoint an agent for service of process in any other jurisdiction than Denmark in which Underwriters are subject to legal suit, action or proceedings based on or arising under this Agreement within 14 days of receiving written notice of such legal suit, action or proceedings and the request to appoint such agent for service. In the event that the Company or the Lender does not appoint such an agent within 14 days of the notice requesting it to do so, the Underwriters may appoint a commercial agent for service for the Company or the Lender (as the case may be) and the Company and the Lender agrees that subject to being notified of such appointment in writing, service upon such commercial agent will constitute service upon the Company and the Lender.		
12.5	Each Party hereby waives to the fullest extent permitted by applicable law, all immunity (whether on the basis of sovereignty or otherwise) from jurisdiction, attachment (both before and after judgment) and execution to which it might otherwise be entitled in any action or proceeding in the courts of any country or jurisdiction, relating in any way to this Agreement and agrees that (i) it will not raise, claim or cause to be pleaded any such immunity at or in respect of any such action		
		17	
	or proceeding and (ii) it can be sued in its own name and is stagement.	ubject to civil and commercial laws with respect to its obligations under this	
12.6	Each party irrevocably agrees that a judgment and/or order of any court referred to in this Clause 12 based on any matter arising out of this Agreement shall be conclusive and binding on it and may be enforced against it in any other jurisdiction, whether or not (subject to due process having been served on it) it participates in the relevant proceedings.		
13	Severability		
13.1	If any provision in this Agreement should be illegal, invalid or unenforceable, the legality, validity and enforceability of the remaining provisions shall not be affected thereby. No variation of this Agreement shall be effective unless in writing and signed by or on behalf of each of the parties to this Agreement. No third party shall be required to agree to any such variation. No party may assign any of its rights under this Agreement without the consent of the party against whom the right operates.		
13.2	The Company, Lender and Leerink recognize that the Depositary is not a party to this Agreement, is not privy to its terms and is not bound by it. The Company, Lender and Leerink understand and agree that to the extent that the performance by the Depositary pursuant to the terms of the Deposit Agreement and Annex I is not consistent with the descriptions set forth in this Agreement, the terms of the Deposit Agreement and Annex I, respectively, shall prevail.		
14	Notices		
14.1	All notices and other communications hereunder shall be in writing and shall be deemed to have been duly given if mailed or transmitted and confirmed by any standard form of telecommunication. Notices to Leerink shall be given to Leerink Partners LLC, One Federal Street, Boston MA 02210, Attn: the General Counsel. Notices to the Lender shall be given to Florian Schönharting, email: fs@nordicbiotech.com. Notices to the Company shall be given to Peder Møller Andersen, email: pma@forward-pharma.com.		
		18	
15	Counterparts		
15.1	This Agreement may be executed by any one or more of the parties hereto and thereto in any number of counterparts, each of which shall be deemed to be an original, but all of such respective counterparts shall together constitute one and the same instrument.		
In witne	ess whereof, this Agreement has been duly executed on behalf of t	the parties hereto the day and year first before written.	
For and	l on behalf of Nordic Biotech Opportunity Fund K/S:		
Name:		Name:	
Title:		Title:	
For and	on behalf of Leerink Partners LLC (as Representative of the Und	erwriters):	
Name:		Name:	
Title:		Title:	
For and	l on behalf of Forward Pharma A/S:		
Name:		Name:	
Title:		Title:	

# NIELSEN NØRAGER

Forward Pharma A/S Østergade 24 A, 1. floor DK-1100 Copenhagen K Denmark LAW FIRM LLP WWW.NNLAW.DK

FREDERIKSBERGGADE 16 1459 COPENHAGEN K DENMARK

VAT NO./CVR. NO. 32 30 33 74

TEL +45 33 11 45 45 FAX +45 33 11 80 81

9 OCTOBER 2014

Ladies and Gentlemen,

# FORWARD PHARMA A/S, COMPANY REG. NO. (CVR) 28865880 - INITIAL PUBLIC OFFERING AND OFFICIAL LISTING ON NASDAQ STOCK EXCHANGE

We have acted as Danish legal counsel to Forward Pharma A/S (the "Company") in relation to an initial public offering (the "Offering") of up to 9,523,810 American Depositary Shares evidenced by American Depositary Receipts (the "Subscription ADSs"), each representing 1 newly issued ordinary share of the Company of DKK 0.10 nominal value.

The Subscription ADSs are created by The Bank of New York Mellon on the basis of up to 9,523,810 new ordinary shares (the "Subscription Shares") to be subscribed for by Leerink Partners LLC ("Leerink").

Further, up to 1,428,571 additional ADSs may be created by The Bank of New York Mellon on the basis of up to 1,428,571 additional new ordinary shares of the Company (the "**Option Shares**"), which may be subscribed for by Leerink pursuant to an over-allotment option provided by the Company.

As used herein, the term "Offer Shares" shall include any Subscription Shares and Option Shares.

The Offering is expected to take place in connection with a listing of all Subscription ADSs (and Option ADSs to the extent that these are issued) on the NASDAQ Stock Exchange.

In our capacity as such counsel, we are familiar with (i) the proceedings relating to the creation of the Company as a Danish public limited liability company organized under the laws of Denmark, and (ii) the proceedings taken and proposed to be taken by the Company in connection with the issuance of the Offer Shares.

This opinion is being furnished in connection with the registration statement, as the same may be amended from time to time, (the "**Registration Statement**") on Form F-1 filed by the Company with the Security and Exchange Commission on 9 October 2014 pursuant to the Securities Act of 1933, as amended (the "**Securities Act**").

# 1 Basis of the opinion

For the purpose of this opinion we have examined the following documents:

- a) A copy of the Registration Statement;
- b) The articles of association of the Company as registered with the Danish Business Authority on the date hereof;
- c) An online transcript from the Danish Business Authority on the date hereof, with respect to the Company;
- d) A signed copy of the minutes from the extraordinary general meeting of the Company held on 11 August 2014 authorising the board of directors to issue the Offer Shares;
- e) A signed copy of the resolution dated 2 October 2014 passed by the board of directors of the Company resolving on the increase of the share capital represented by the Subscription Shares without pre-emption rights for existing shareholders and to be subscribed for at the Offering;
- f) The form of the stock lending agreement (the "Stock Lending Agreement") to be concluded among Nordic Biotech Opportunity Fund K/S, Leerink, and the Company, dated 8 October 2014;

g) Such other documents, agreements and records as we have deemed necessary for the purposes of rendering this opinion.

The documents mentioned in Sections 1a) — 1g) above are referred to as the "**Documentation**" or individually as a "**Document**".

### 2 Assumptions

In rendering this opinion, we have relied, without independent verification, upon the following assumptions:

- a) That each Document is true, correct and fully updated and has not been amended, waived or revoked after the date of each such Document and that all material supplied to us has been supplied in full and has not subsequently been altered or amended;
- b) That the meetings which form the basis of or are referred to in Sections 1d and 1e were duly and properly convened and conducted and that all participants therein has acted *bona fide* throughout;
- c) That copies submitted to us of minutes of meetings and/or resolutions correctly record the proceedings at such meetings and/or subject matter which they purport to record, and that all resolutions set out in such copies were duly passed;
- d) The information contained in the online transcript from the Danish Business Authority (cf. Section 1c)) concerning the Company being accurate, complete and updated;
- e) The resolution of the board of directors of the Company to increase the share capital represented by the Offer Shares referred to in 1e above, upon final allotment and pricing by the board, will be duly registered with the Danish Business Authority;

3

- f) That the Stock Lending Agreement will be duly executed and delivered by all parties to said agreement;
- g) The conformity to original and final documentation to the extent we have been presented with copies or draft Documentation, and that originals were or will be executed in the manner appearing on the copy; and
- h) The genuineness of all signatures and dates on all Documentation (other than on behalf of the Company), examined by us, and that the identities of the signatories are as stated or written.

# 3 Qualifications

In addition to the assumptions set forth in Section 2 above, this opinion is subject to the following qualifications:

- a) This opinion is limited to the matters of the laws of Denmark as in effect today and as such laws are currently applied by Danish courts and we express no opinion with respect to the laws of any other jurisdiction nor have we made any investigations as to any law other than the laws of Denmark; and
- b) In rendering this opinion we have relied on certain matters of information obtained from the Company and other sources reasonably believed by us to be credible;

We assume no obligation to notify you of any changes to this opinion as a result of any facts or circumstances that may come to our attention in the future or as a result of any change in the laws of Denmark which may hereafter occur.

# 4 Opinion

Based on the assumptions set forth in Section 2 and the qualifications set forth in Section 3, we are of the opinion that:

4

- a) The Company is a Danish public limited liability company (in Danish: "aktieselskab") duly incorporated and validly existing under the laws of Denmark and registered with the Danish Business Authority;
- b) The Offer Shares will when duly subscribed for, paid and registered with the Danish Business Authority as contemplated by the Registration Statement be validly issued, fully paid and non-assessable.
- c) The Lending Shares and the Option Lending Shares (as defined in the Stock Lending Agreement) initially substituting for the Offer Shares have been validly issued, fully paid and registered with the Danish Business Authority and are non-assessable.

# 5 Reliance and limitation of liability

This opinion is limited to matters of the laws of Denmark as in effect and applied on the date of this opinion. We express no opinion with respect to the laws of any other jurisdiction, nor have we made any investigation as to any laws other than the laws of Denmark.

This opinion is strictly limited to the matters stated herein and is not to be read as extending by implication to any other matter.

We hereby consent to the filing of this opinion as an exhibit to the Registration Statement and to the reference to our law firm under the caption "Legal Matters" in the Registration Statement. In giving this consent, we do not admit that we are in the category of persons whose consent is required under Section 7 of the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

Yours sincerely, Nielsen Nørager Law Firm LLP

/s/ Jakob Mosegaard Larsen Jakob Mosegaard Larsen Attorney-at-law



October 9, 2014.

Forward Pharma A/S Østergade 24A, 1 1000 Copenhagen K Denmark

Ladies and Gentleman,

Deloitte Statsautoriseret Revisionspartnerselskab CVR-no. 33 96 35 56 Weidekampsgade 6 P.O. Box 1600 0900 Copenhagen C Denmark

Phone +45 36 10 20 30 Fax +45 36 10 20 40

www.deloitte.dk

# Assessment of Section "Taxation – Danish Tax Considerations" of the Prospectus for Forward Pharma A/S

As agreed, we are acting as special Danish tax counsel to Forward Pharma A/S, a public limited liability company registered with the Danish Business Authority under company registration number 28865880 (the "Company"), with respect to certain legal matters in connection with the registration statement on Form F-1 (Registration No. 333-198013) (as filed and amended, the "Registration Statement") filed by the Company on October 9, 2014 with the Securities and Exchange Commission (the "Commission") registering American Depositary Shares representing 9,523,810 ordinary shares of the Company, DKK 0.10 nominal value per share, under the Securities Act of 1933, as amended (the "Act"). As requested, we have examined the section "Taxation – Danish Tax Considerations" within the Registration Statement in order to assess the accuracy of the information provided in the section.

### Conclusion

Based on our review of the section "Taxation – Danish Tax Considerations" of the Registration Statement, it is our opinion that the information in the section is both accurate and complete.

In arriving at our opinion, we have examined and relied upon the Registration Statement and any other documents that in our judgment are necessary or appropriate to enable us to render our opinion.

We hereby consent to the filing of this opinion as an exhibit to the Registration Statement and to the use of our name under the caption "Legal Matters" in the Registration Statement. In giving such consent, we do not admit that we are within the category of persons whose consent is required under Section 7 of the Act, or the rules or regulations of the Commission promulgated thereunder.

Member of Deloitte Touche Tohmatsu Limited

# Deloitte

Statsautoriseret Revisionspartnerselskab

/s/ Erik Banner-Voigt Erik Banner-Voigt

Tax Partner

/s/ Jakob Schilder-Knudsen

Jakob Schilder-Knudsen

Tax Director

2



1095 Avenue of the Americas New York, NY 10036-6797 +1 212 698 3500 Main +1 212 698 3599 Fax www.dechert.com

October 9, 2014

Forward Pharma A/S Østergade 24A, 1 1100 Copenhagen K, Denmark

Re: Forward Pharma A/S Registration Statement on Form F-1

### Ladies and Gentlemen:

We are acting as U.S. counsel for Forward Pharma A/S, a public limited liability company registered with the Danish Business Authority under company registration number 28865880 (the "Company"), with respect to certain legal matters in connection with the registration statement on Form F-1 (Registration No. 333-198013) (as filed and amended, the "Registration Statement") filed by the Company on October 9, 2014 with the Securities and Exchange Commission (the "Commission") registering American Depositary Shares representing 9,523,810 ordinary shares (or up to an additional 1,428,571 ordinary shares to cover over-allotments) of the Company, DKK 0.10 nominal value per share, under the Securities Act of 1933, as amended (the "Act"). You have requested our opinion concerning the statements in the Registration Statement under the caption "U.S. Federal Income Tax Considerations for U.S. Holders."

This opinion is based on various facts and assumptions, and is conditioned upon certain representations made by the Company as to factual matters. In addition, this opinion is based upon the factual representations of the Company concerning its business, properties and governing documents as set forth in the Registration Statement.

In providing this opinion, we have made such legal and factual examinations and inquiries, including an examination of originals or copies certified or otherwise identified to our satisfaction of such documents, corporate records and other instruments, as we have deemed necessary or appropriate for purposes of this opinion. In our examination, we have assumed the authenticity of all documents submitted to us as originals, the genuineness of all signatures thereon, the legal capacity of natural persons executing such documents and the conformity to authentic original documents of all documents

US Austin Boston Charlotte Hartford Los Angeles New York Orange County Philadelphia Princeton San Francisco Silicon Valley Washington DC EUROPE Brussels Dublin Frankfurt London Luxembourg Moscow Munich Paris ASIA Beijing Hong Kong

submitted to us as copies. For the purpose of our opinion, we have not made an independent investigation or audit of the facts set forth in the above-referenced documents. In addition, in rendering this opinion we have assumed the truth and accuracy of all representations and statements made to us which are qualified as to knowledge or belief, without regard to such qualification.

We are opining herein as to the effect on the subject transaction only of the federal income tax laws of the United States and we express no opinion with respect to the applicability thereto, or the effect thereon, of other federal laws, the laws of any state or any other jurisdiction or as to any matters of municipal law or the laws of any other local agencies within any state.

Based on such facts, assumptions and representations and subject to the limitations set forth herein and in the Registration Statement, we hereby confirm that all statements of legal conclusions set forth under the caption "U.S. Federal Income Tax Considerations for U.S. Holders" in the Registration Statement constitute the opinion of Dechert LLP with respect to the matters contained therein as of the effective date of the Registration Statement, and are conditioned upon the assumptions, qualifications and limitations set forth therein. No opinion is expressed as to any matter not discussed herein.

This opinion is rendered to you as of the effective date of the Registration Statement, and we undertake no obligation to update this opinion subsequent to the date hereof. This opinion is based on various statutory provisions, regulations promulgated thereunder and interpretations thereof by the Internal Revenue Service and the courts having jurisdiction over such matters, all of which are subject to change either prospectively or retroactively. Also, any variation or difference in the facts from those set forth in the representations described above, including in the Registration Statement, may affect the conclusions stated herein.

This opinion is furnished to you, and is for your use in connection with the transactions set forth in the Registration Statement. This opinion may not be relied upon by you for any other purpose, or furnished to, assigned to, quoted to or relied upon by any other person, firm or other entity, for any purpose, without our prior written consent, except that this opinion may be relied upon by persons entitled to rely on it pursuant to applicable provisions of federal securities law.

2

 $\label{eq:Act} Act, or the rules or regulations of the Commission promulgated the$ reunder.

Very truly yours,

/s/ Dechert LLP Dechert LLP